## UNIVERSITY OF ALBERTA

## EXPERIMENTAL AND NUMERICAL ANALYSIS OF FRP SHEETS BONDED TO CONCRETE

by

## Ayman Mohamed Samir Kamel

A thesis submitted to the faculty of Graduate Studies and Research in partial fulfillment of the requirements for the degree of Doctor of Philosophy in Structural Engineering

Department of Civil and Environmental Engineering

Edmonton, Alberta

Spring 2003

National Library of Canada

Acquisitions and Bibliographic Services

395 Wellington Street Ottawa ON K1A 0N4 Canada Bibliothèque nationale du Canada

Acquisisitons et services bibliographiques

395, rue Wellington Ottawa ON K1A 0N4 Canada

> Your file Votre référence ISBN: 0-612-82124-2 Our file Notre référence ISBN: 0-612-82124-2

The author has granted a nonexclusive licence allowing the National Library of Canada to reproduce, loan, distribute or sell copies of this thesis in microform, paper or electronic formats.

The author retains ownership of the copyright in this thesis. Neither the thesis nor substantial extracts from it may be printed or otherwise reproduced without the author's permission.

L'auteur a accordé une licence non exclusive permettant à la Bibliothèque nationale du Canada de reproduire, prêter, distribuer ou vendre des copies de cette thèse sous la forme de microfiche/film, de reproduction sur papier ou sur format électronique.

L'auteur conserve la propriété du droit d'auteur qui protège cette thèse. Ni la thèse ni des extraits substantiels de celle-ci ne doivent être imprimés ou aturement reproduits sans son autorisation.



## **University of Alberta**

## Library Release Form

## Name of Author: Ayman Mohamed Samir Kamel

# Title of Thesis: Experimental and Numerical Analysis of FRP Sheets Bonded to Concrete

**Degree:** Doctor of Philosophy

Year this degree granted: 2003

Permission is hereby granted to the University of Alberta Library to reproduce single copies of this thesis and to lend or sell such copies for private, scholarly or scientific research purposes only.

The author reserves all other publication and other rights in association with the copyright in the thesis, and except as herein before provided, neither the thesis nor any substantial portion thereof may be printed or otherwise reproduced in any material form whatever without the author's prior written permission.

Kymun Hamel

13 Omar Ebn Elkatab St. Villal El-Gamaa Zagazig, Egypt

Date: NOV. 28, 2002

## **University of Alberta**

#### **Faculty of Graduate Studies and Research**

The Undersigned certify that they have read, and recommend to the Faculty of Graduate Studies and Research for acceptance, a thesis entitled Experimental and Numerical Analysis of FRP Sheets Bonded to Concrete submitted by Ayman Mohamed Samir Kamel in partial fulfillment of the requirements for the degree of Doctor of Philosophy in Structural Engineering.

Dr. Scott Alexander, Committee Chair and Examiner Dr. Jung-June Roger Cheng, Supervisor Dr. Alaa Elwi, Co-supervisor Dr. Richard Chalatarnyk, Committee member Dr. Ben Jar Committee member

Dr. Nenkumar Banthia, External Examiner

Date: Dec. 4, 2002

#### ABSTRACT

This thesis investigates the bond behaviour of fibre reinforced polymer (FRP) sheets externally bonded to concrete structures. External bonding of FRP sheets is one of the techniques used to restore deteriorating reinforced concrete structures. The primary mode of failure related to the use of FRP is debonding of the sheets from the concrete surface. Progress in understanding the mechanisms of stress transfer and bond failure that result from the use of externally bonded FRP plates and sheets is necessary to develop sound design codes.

Two experimental test series using pull-apart test method and a modified push-apart test method proposed in this study were conducted. The effects of different parameters, such as the bond length, width, and anchor sheets, on the bond behaviour between FRP sheets and concrete were investigated.

A parametric study investigating the main parameters affecting the bond behaviour was conducted using both two and three-dimensional nonlinear finite element models. The numerical models showed satisfactory agreement with the test results. Trend relationships were produced from the parametric study.

Three conventional bond test methods in addition to the modified push-apart test method were analysed using numerical models. Two bond test methods were recommended as standard test methods. A conversion standard between the different methods was also proposed based on a comparison between the test methods using numerical analysis and experimental results obtained from this study and from the literature.

Two regression analysis based models were developed to predict the effective bond lengths and the bond capacity of FRP sheets bonded to concrete. Test results from the current study and the literature were used to derive the equations and compare their accuracy compared to existing equations. The results show that the proposed equations are more accurate and reliable than those proposed in the literature.

## ACKNOWLEDGMENTS

There are a number of people who contributed to the completion of this thesis. First and foremost, the author would like to thank Prof. Alaa Elwi for his continuous guidance and support throughout. His knowledge and pleasant demeanour were very much appreciated and helped bring this work to a successful end. The author would also like to thank Prof. Roger Cheng for his substantial input towards the completion, direction of this thesis, and support throughout.

The author would like to acknowledge Richard Hilfrich and Larry Burden of the LF. Morrison Structural Laboratory at the University of Alberta for their assistance in the experimental work, as well as all the fellow graduate students who have helped in many stages of the research.

Finally, the author wishes to thank the Natural Sciences and Engineering Research Council of Canada (NSERC) and the Canadian Network of Centres of Excellence on Intelligent Sensing for Innovative Structures (ISIS Canada) for their financial assistance. The author also acknowledges Mitsubishi Canada Ltd. for providing the CFRP material and epoxy.

## **DEDICATION**

I would like to dedicate this thesis to my beloved family, especially my wife Eman who stood beside me and helped me concentrate on my thesis, my beautiful daughters Nourhan, Salma, and Liyla for being so patient, and finally my father, Professor M. S. Kamel, who has always been my inspiration.

## TABLE OF CONTENTS

## ACKNOWLEDGMENT ABSTRACT TABLE OF CONTENTS LIST OF TABLES LIST OF FIGURES LIST OF SYMBOLS

1.	INI	RODU	JCTION
	1.1.	Backg	round1
	1.2.	Scope	2
	1.3.	Object	ives2
	1.4.	Thesis	Outline
2.	LIJ	TERAT	URE REVIEW6
	2.1	Introdu	action6
	2.2	Fibre I	Reinforced Polymers (FRP)
		2.2.1	Types of Fibres
		2.2.2	Types of Matrix
		2.2.3	Forms of FRP Materials
		2.2.4	Advantages of Using FRP9
		2.2.5	Disadvantages and Limitations of using FRP9
	2.3	Adhes	ive Used to Bond FRP Sheets to Concrete10
	2.4	Streng	thening of Reinforced Concrete Beams11
		2.4.1	Flexural Strengthening of RC Beams
		2.4.2	Shear Strengthening of RC Beams
	2.5	Possib	le Failure Mechanisms14
		2.5.1	Failure Mechanisms for Beams Strengthened in Flexure
		2.5.2	Failure Mechanisms for Beams Strengthened in Shear
	2.6	Conve	ntional Bond Test Methods17
		2.6.1	Direct Tension Bond Tests

	2.6.2	Bend	ing Bond Tests	17
	2.6.3	Shear	Bond Tests	
	2.	.6.3.1	Direct Pull Bond Tests	
	2.	.6.3.2	Indirect Pull-Apart Tests	
	2.	.6.3.3	Indirect Push-Apart Tests	
2.7	Factor	rs Affe	cting Bond Behaviour	20
	2.7.1	FRP	Sheet Bond Length	
	2.7.2	FRP	Sheet Bond Width	
	2.7.3	FRP	Sheet Stiffness	23
	2.7.4	Conc	rete Strength	
	2.7.5	Conc	rete Surface Preparation	24
	2.7.6	Туре	of Test Conducted	25
	2.7.7	Anch	orage Details	25
2.8	Bond S	Strengt	n Models	
	2.8.1	Empi	rical Models	28
	2.8.2	Fract	ure Mechanics Based Models	
	2.8.3	Desig	gn Proposed Models	
2.9	Summ	ary		
2 <b>FVD</b>	FDIM	ENTR A	IDDACDAM	43
21	Introd	uction		43
2.1	Modif	fied Du	ch. A nort Test	
5.2	2 7 1	Snec	men Description	43
	3.2.1	Test	set up and Instrumentation	лл
	3.2.2	Vorie	bles Studied	
2.2	D.2.3	v alla	est	
5.5	2 2 1	spart I	man Description	
	222	Test	ant up and instrumentation	
	3.3.4 2.2.2	Voria	bles studied	40 <i>A</i> 77
<b>?</b> A	J.J.J	v ana	10165 SIUUICU	47 ۸۵
5.4		iai Pro	perioes	40، ۸۵
	3.4.1	Conc		40

	3.4.2 CFRP Sheets	
	3.4.3 Epoxy	
	3.4.4 Steel	
3.5	5 Bonding the FRP Sheets	
4 TES	ST RESULTS, OBSERVATIONS, AND DISCUSSION	
4.1	1 Introduction and Overview	
4.2	2 Failure Mechanism	63
	4.2.1 Modified Push-Apart Test Failure Mechanism	63
	4.2.2 Pull-Apart Test Failure Mechanism	
	4.2.3 Anchor Sheets	65
4.3	3 Load Transfer Process	
	4.3.1 Load Transfer Along the CFRP Sheet Length	
	4.3.2 Load Transfer Across the CFRP Sheet Width	
4.4	4 Bond Length	
	4.4.1 General Discussion	72
	4.4.2 Effective Length Evaluation	73
4.5	5 Bond Width	74
4.6	5 Test Method	76
4.7	7 L/W Ratio	
4.8	8 Anchor Sheet Configuration	
	4.8.1 Anchor Sheet Location Relative to the Tested Sheet	
	4.8.2 Anchor Distance from Crack	
	4.8.3 Anchor Extension Beyond Tested Sheet	80
5. N	NUMERICAL MODEL	116
5.1	1. Introduction and Overview	116
5.2	2. Geometric Modeling	117
	5.2.1. 2D Models	
	5.2.2. 3D Models	118
5.3	3. Boundary Conditions and Loading	
	5.3.1 Boundary Conditions and Loading of 2D Models	

	5.3.2.	Boundary Conditions and Loading of 3D Models	
5.4	1. Mate	rial Properties	
	5.4.1.	Concrete	
	5.4.2.	CFRP	
	5.4.3.	Adhesive	
	5.4.4.	Steel	
5.5	5. Conc	rete-CFRP Sheet Interface	
5.0	5. Anal	ysis Procedures	
5.7	7. Verif	ication	
	5.7.1.	2D Model Verification	
	5.7.2.	3D Model Verification	
6. PA	RAME	TRIC STUDY	160
6.1	1 Intro	oduction	160
6.2	2 FRP	Sheet Bond Length Effect	
6.3	3 FRP	Sheet Bond Width Effect	
	6.3.1	General	
	6.3.2	The Effect of FRP Sheet Width to Concrete Width Ratio	
6.4	1 FRP	Sheet Stiffness Effect	172
6.5	5 Con	crete Strength Effect	
7 CO	MPAR	ISON BETWEEN BOND TEST METHODS	
7.1	l Inti	roduction and Overview	
7.2	2 Dir	ect Pull Bond Test	
7.3	3 Pul	I-Apart Test	
7.4	4 Pus	sh-Apart Test	
7.5	5 Mo	dified Push-Apart Test	
7.6	5 Coi	mparison Between the Bond Tests	
	7.6.1	Specimen Preparation	210
	7.6.2	Test Serviceability and Execution	210
	7.6.3	Test Stability	
	7.6.4	Relationship Between Test Methods	211
		a.	

8.	FRP SHEET BOND MODEL	242
	8.1. Introduction and Overview	242
	8.2. Design Implementations of the Analytical Model	242
	8.3. Design Model	
	8.3.1. Effective Bond Length Model	244
	8.3.2 Bond Capacity Model	246
9	SUMMARY, CONCLUSIONS, AND RECOMMENDATIONS	
	9.1 Summary	
	9.2 Conclusions	
	9.2.1 Conclusions from the Experimental Analysis	
	9.2.2 Conclusions from the Numerical Analysis	
	9.3 Recommendations	
10	REFERENCES	

## LIST OF TABLES

Table 3-1 Modified push-apart test variables.	52
Table 3-2 Modified push-apart concrete compression and split cylinder test results	.52
Table 3-3 Pull-apart test concrete compression and tensile split cylinder test results	53
Table 3-4 Pull-apart test variables	53
Table 3-5 Push-apart test concrete mix for 1.0 m <sup>3</sup>	54
Table 3-6 Pull-apart test concrete mix for 1.0 m <sup>3</sup>	54
Table 3-7 Material properties	54
Table 4-1 Push-apart bond test specimens' details and results	81
Table 4-2 Pull-apart bond test specimen's details and results	81
Table 6-1 Main material properties used in the parametric study	176
Table 6-2 Models with FRP sheet bond lengths varying with bond width	177
Table 6-3 Models with bond lengths varying with sheet stiffness, Et	.178
Table 6-4 Models with bond lengths varying with concrete strength	179
Table 6-5 Models with various concrete width and FRP bond length	180
Table 6-6 3D models with various concrete width and FRP bond widths	180
Table 6-7 2D models with various E and t values but with the same Et product	.180
Table 7-1 Material properties used in all models	.214
Table 8-1 Effective bond length comparison details	.251
Table 8-2 Test vs. predicted effective bond length summary	252
Table 8-3 Summary of comparison with all test results	253

1.	INI	RODU	JCTION
	1.1.	Backg	round1
	1.2.	Scope	
	1.3.	Object	ives2
	1.4.	Thesis	Outline
2.	LIJ	TERA 1	URE REVIEW
	2.1	Introd	uction6
	2.2	Fibre 1	Reinforced Polymers (FRP)6
		2.2.1	Types of Fibres7
		2.2.2	Types of Matrix
		2.2.3	Forms of FRP Materials8
		2.2.4	Advantages of Using FRP9
		2.2.5	Disadvantages and Limitations of using FRP9
	2.3	Adhes	ive Used to Bond FRP Sheets to Concrete10
	2.4	Streng	thening of Reinforced Concrete Beams11
		2.4.1	Flexural Strengthening of RC Beams
		2.4.2	Shear Strengthening of RC Beams
	2.5	Possib	le Failure Mechanisms14
		2.5.1	Failure Mechanisms for Beams Strengthened in Flexure14
		2.5.2	Failure Mechanisms for Beams Strengthened in Shear16
	2.6	Conve	ntional Bond Test Methods17
		2.6.1	Direct Tension Bond Tests
		2.6.2	Bending Bond Tests
		2.6.3	Shear Bond Tests
		2.	6.3.1 Direct Pull Bond Tests
		2.	6.3.2 Indirect Pull-Apart Tests
		2.	6.3.3 Indirect Push-Apart Tests
	2.7	Factor	s Affecting Bond Behaviour20
		2.7.1	FRP Sheet Bond Length
		2.7.2	FRP Sheet Bond Width

		2.7.3 FRP Sheet Stiffness	
		2.7.4 Concrete Strength	
		2.7.5 Concrete Surface Preparation	
		2.7.6 Type of Test Conducted	
		2.7.7 Anchorage Details	
	2.8	Bond Strength Models	
		2.8.1 Empirical Models	
		2.8.2 Fracture Mechanics Based Models	
		2.8.3 Design Proposed Models	
	2.9	Summary	
3	EXP	ERIMENTAL PROGRAM	
	3.1	Introduction	
	3.2	Modified Push-Apart Test	
		3.2.1 Specimen Description	
		3.2.2 Test set-up and Instrumentation	
		3.2.3 Variables Studied	
	3.3	Pull-Apart Test	
		3.3.1 Specimen Description	46
		3.3.2 Test set-up and instrumentation	
		3.3.3 Variables studied	
	3.4	Material Properties	
		3.4.1 Concrete	
		3.4.2 CFRP Sheets	
		3.4.3 Epoxy	
		3.4.4 Steel	
	3.5	Bonding the FRP Sheets	
4	TES	T RESULTS, OBSERVATIONS, AND DISCUSSION	62
	4.1	Introduction and Overview	
	4.2	Failure Mechanism	63
		4.2.1 Modified Push-Apart Test Failure Mechanism	

4.2.2 Pull-Apart Test Failure Mechanism	64
4.2.3 Anchor Sheets	65
4.3 Load Transfer Process	66
4.3.1 Load Transfer Along the CFRP Sheet Length	66
4.3.2Load Transfer Across the CFRP Sheet Width	69
4.4 Bond Length	72
4.4.1 General Discussion	72
4.4.2 Effective Length Evaluation	73
4.5 Bond Width	74
4.6 Test Method	76
4.7 L/W Ratio	77
4.8 Anchor Sheet Configuration	
4.8.1 Anchor Sheet Location Relative to the Tested Sheet	
4.8.2 Anchor Distance from Crack	
4.8.3 Anchor Extension Beyond Tested Sheet	
5. NUMERICAL MODEL	
5.1. Introduction and Overview	116
5.2. Geometric Modeling	117
5.2.1. 2D Models	
5.2.2. 3D Models	
5.3. Boundary Conditions and Loading	
5.3.1. Boundary Conditions and Loading of 2D Models	120
5.3.2. Boundary Conditions and Loading of 3D Models	120
5.4. Material Properties	121
5.4.1. Concrete	121
5.4.2. CFRP	123
5.4.3. Adhesive	124
5.4.4. Steel	125
5.5. Concrete-CFRP Sheet Interface	125
5.6. Analysis Procedures	127
5.7. Verification	

5.7.1. 2D Model Verification	129
5.7.2. 3D Model Verification	130

6	. PARAMETRIC STUDY	160
	6.1 Introduction	
	6.2 FRP Sheet Bond Length Effect	
	6.3 FRP Sheet Bond Width Effect	
	6.3.1 General	
	6.3.2 The Effect of FRP Sheet Width to Concrete Width Ratio	
	6.4 FRP Sheet Stiffness Effect	
	6.5 Concrete Strength Effect	173
7	COMPARISON BETWEEN BOND TEST METHODS	
	7.1 Introduction and Overview	200
	7.2 Direct Pull Bond Test	201
	7.3 Pull-Apart Test	203
	7.4 Push-Apart Test	205
	7.5 Modified Push-Apart Test	207
	7.6 Comparison Between the Bond Tests	209
	7.6.1 Specimen Preparation	210
	7.6.2 Test Serviceability and Execution	210
	7.6.3 Test Stability	210
	7.6.4 Relationship Between Test Methods	211
8	. FRP SHEET BOND MODEL	
	8.1. Introduction and Overview	242
	8.2. Design Implementations of the Analytical Model	
	8.3. Design Model	
	8.3.1.Effective Bond Length Model	
	8.3.2 Bond Capacity Model	
9	SUMMARY, CONCLUSIONS, AND RECOMMENDATIONS	

9.1 Summary	
9.2 Conclusions	
9.2.1 Conclusions from the Experimental Analysis	
9.2.2Conclusions from the Numerical Analysis	
9.3 Recommendations	
10 REFERENCES	270

## LIST OF TABLES

Table 3-8 Modified push-apart test variables   52	2
Table 3-9 Modified push-apart concrete compression and split cylinder test results52	2
Table 3-10 Pull-apart test concrete compression and tensile split cylinder test results5	3
Table 3-11 Pull-apart test variables.   53	3
Table 3-12 Push-apart test concrete mix for 1.0 m <sup>3</sup>	4
Table 3-13 Pull-apart test concrete mix for 1.0 m <sup>3</sup>	4
Table 3-14 Material properties.	4
Table 4-1 Push-apart bond test specimens' details and results8	31
Table 4-2 Pull-apart bond test specimen's details and results	31
Table 6-1 Main material properties used in the parametric study	6
Table 6-2 Models with FRP sheet bond lengths varying with bond width17	7
Table 6-3 Models with bond lengths varying with sheet stiffness, Et17	78
Table 6-4 Models with bond lengths varying with concrete strength	9
Table 6-5 Models with various concrete width and FRP bond length	0
Table 6-6 3D models with various concrete width and FRP bond widths	0
Table 6-7 2D models with various E and t values but with the same Et product18	30
Table 7-1 Material properties used in all models	4
Table 8-1 Effective bond length comparison details   25	1
Table 8-2 Test vs. predicted effective bond length summary   25	2
Table 8-3 Summary of comparison with all test results   252	3

## LIST OF FIGURES

Figure 2-1 Typical failure stress of unidirectional GFRP with respect to axis orientation.35
Figure 2-2 Typical stress vs. strain relationship for different types of fibres
Figure 2-3 Flexure strengthened in RC beams
Figure 2-4 Load vs. deflection, flexural strengthened beams with and without FRP36
Figure 2-5 Shear strengthened RC beams
Figure 2-6 Load vs. deflection, shear-strengthened beams with and without FRP
Figure 2-7 Possible failure mechanisms of RC beams strengthened in flexure
Figure 2-8 Direct tension bond test (a) elevation (b) plan view
Figure 2-9 Bond strength-compressive strength
Figure 2-10 Bending bond test (a) elevation (b) bottom plan view40
Figure 2-11 Direct pull-apart bond test set-up40
Figure 2-12 Direct pull-apart bond test specimen free body diagram
Figure 2-13 Pull-apart bond test (a) set-up (b) free body diagram, plan view41
Figure 2-14 Push-apart bond test (a) set-up (b) free body diagram
Figure 2-15 Anchoring methods for shear strengthening
Figure 3-1 Orthographic of the modified push-apart test
Figure 3-2 Modified push-apart test steel form and cage details
Figure 3-3 Modified push-apart test (a) plan view (b) side view
Figure 3-4 Modified push-apart test set-up and instrumentation56
Figure 3-5 S-series strain gauge locations on CFRP sheets and concrete
Figure 3-6 Pull-apart forms and reinf. (a) without spiral (b) with spiral
Figure 3-7 Pull-apart specimen (a) Steel form (b) bars alignment arrangement
Figure 3-8 Pull-apart specimen dimensions, details, and anchor configurations
Figure 3-9 Pull-apart test set-up and instrumentation
Figure 3-10 Concrete girder strengthened in flexure and shear with FRP sheets60
Figure 3-11 Strain gauge locations and sheet dimensions for P9 to P1361
Figure 4-1 Crack propagation and failure zone formation sketch
Figure 4-2 Schematic sketch of the failure zone plan view in the push-apart tests
Figure 4-3 Push-apart failure surface (a) concrete (b) CFRP sheet
Figure 4-4 Side view of pull-apart specimen showing crack opening

Figure 4-5 Typical post failure mode for specimens P1 through P13, accept for specimen P3 (a) side view (b) CFRP sheet surface (c) concrete surface
Figure 4-6 Specimen P3 failure mode
Figure 4-7 Specimen P15 failure mode
Figure 4-8 Typical failure, anchor (a) above sheet (b) below sheet
Figure 4-9 Failure mode (a) Specimen P17 (b) Specimen P24
Figure 4-10 Average strain distributions for specimen S185
Figure 4-11 Average strain distributions for Specimen S2
Figure 4-12 Average strain distributions for Specimen S3
Figure 4-13 Average strain distributions for Specimen S4
Figure 4-14 Average strain distributions for Specimen S5
Figure 4-15 Strain distributions at centre of sheet for specimen P9
Figure 4-16 Strain distribution at centre of sheet for specimen P10
Figure 4-17 Strain distributions at centre of sheet for specimen P11
Figure 4-18 Strain distributions at centre of sheet for specimen P12
Figure 4-19 Strain distributions at centre of sheet for specimen P13
Figure 4-20 Load vs. transfer length for S-series specimens with $w = 150 \text{ mm} \dots 90$
Figure 4-21 Load vs. transfer length for P-series specimens with $w = 150 \text{ mm} \dots 90$
Figure 4-22 Average bond stress in sheet centre vs. relative load for S5
Figure 4-23 Schematic sketch deriving incremental average bond stress
Figure 4-24 Average bond stress in sheet centre vs. relative load for P1092
Figure 4-25 Strain distributions along the sheet width of S1 at crack location
Figure 4-26 Strain distributions along half the sheet width of S2 at 0, 15, 30, and 45 mm from the crack location
Figure 4-27 Strain distributions along the sheet width of S3 at crack location
Figure 4-28 Strain distributions along the sheet width of S4 at crack location
Figure 4-29 Strain distributions along the sheet width of S5 at crack location
Figure 4-30 Strain distributions across the width of P9 at crack location95
Figure 4-31 Strain distribution across the width of P10 at crack location96
Figure 4-32 Strain distributions across the width of P11 at crack location
Figure 4-33 Strain distributions across the width of P12 at crack location
Figure 4-34 Strain distributions across the width of P13 at crack location

Figure 4-35 CFRP sheet width vs. strain difference between the edge and centre of the sheet at the crack location
Figure 4-36 Strain distributions in the sheet centre and edge along its length at various load levels in S4
Figure 4-37 Average bond stresses along width vs. relative load level for S1
Figure 4-38 Average bond stresses across width vs. relative load level for S4
Figure 4-39 Average bond stresses across width vs. relative load level for S2 100
Figure 4-40 Average bond stresses across width vs. relative load levels for P10 100
Figure 4-41 Average bond stresses across width vs. relative load level for P11 101
Figure 4-42 Load vs. displacement curves for specimens with $w = 150 \text{ mm}$ 101
Figure 4-43 Load vs. displacement for specimens with $w = 150 \text{ mm}$
Figure 4-44 Strain distributions along the sheet centre at 20 kN, w = 150 mm 102
Figure 4-45 Strain distributions along the sheet centre at 20 kN, $w = 150 \text{ mm} \dots \dots \dots 103$
Figure 4-46 Response for specimens with same sheet width, $w = 100 \text{ mm} \dots 103$
Figure 4-47 Load vs. bond length for specimens with $w = 100 \text{ mm}$
Figure 4-48 Response for specimens with $L = 150 \text{ mm}$
Figure 4-49 Response for specimens with $L = 150 \text{ mm}$
Figure 4-50 Load vs. width and Load vs. average bond strength for P-series specimens with L = 150 mm
Figure 4-51 Strain distributions along the sheet centre, S-series, $L = 150 \text{ mm} \dots \dots 106$
Figure 4-52 Strain distributions along the sheet centre, P-series, $L = 150 \text{ mm} \dots 106$
Figure 4-53 Response comparison for specimens with $w = 150 \text{ mm}$
Figure 4-54 Strain distributions at centre of sheet for specimens P9 and S2 107
Figure 4-55 Strain distributions at centre of sheet for specimens P10 and S3 108
Figure 4-56 Strain distributions at centre of sheet for specimens P11 and S5 108
Figure 4-57 Response for specimens with the same bond area
Figure 4-58 Strain distributions along the sheet centre, same bond area 109
Figure 4-59 Response for specimens with $a = 0$ and $b = 75$ mm
Figure 4-60 Response for specimens with a =25 mm and b=75 mm 110
Figure 4-61 Response for specimens with $a = 50 \text{ mm}$ and $b=75 \text{ mm}$
Figure 4-62 Response for specimens with a =75 mm, b=75 mm
Figure 4-63 Response for specimens with $a = 25 \text{ mm}$ , $b = 50 \text{ mm}$
Figure 4-64 Response for specimens with a =25 mm, b=25 mm

Figure 4-65 Effect of distance from crack "a", anchor on top and $b = 75 \text{ mm}$	113
Figure 4-66 Effect of distance from crack "a", anchor below sheet and b=75 mm	113
Figure 4-67 Anchor distance from crack, b = 75 mm	114
Figure 4-68 Effect of anchor extension beyond sheet "b" when the anchor is above the sheet and a = 25 mm	114
Figure 4-69Effect of anchor extension beyond sheet "b" when the anchor is below the sheet and a = 25 mm	115
Figure 4-70 Anchor spacing for specimens with $a = 25 \text{ mm}$	115
Figure 5-1 Modified push-apart specimen plan view and symmetry lines	132
Figure 5-2 Pull-apart specimen elevation and symmetry lines	132
Figure 5-3 2D modified push-apart model mesh, loading, and boundary conditions	133
Figure 5-4 Half the 2D pull-apart model mesh, loading, and boundary conditions	133
Figure 5-5 Modified push-apart specimen (a) planes of symmetry (b) modeled part	134
Figure 5-6 Pull-apart specimen (a) planes of symmetry (b) modeled part	134
Figure 5-7 Modified push-apart 3D model	135
Figure 5-8 Pull-apart test mesh, loading, and boundary conditions	136
Figure 5-9 Concrete yield and failure surfaces in plain stress	136
Figure 5-10 Calculation of the displacement at which a linear loss of strength after cracking gives zero stress	137
Figure 5-11 Uniaxial behaviour of plain concrete (ABAQUS, 1998)	137
Figure 5-12 Stress-strain behaviour for CFRP and steel	138
Figure 5-13 2D modified push-apart model	138
Figure 5-14 Strain contour in load direction for modified push-apart model	139
Figure 5-15 2D pull-apart mesh, boundary conditions, and deformed shape	. 139
Figure 5-16 Load-displacement curves experimental and 2D model of S1	140
Figure 5-17 Load-displacement experimental and 2D model of S2	. 140
Figure 5-18 Load-displacement experimental and 2D model of S3	. 141
Figure 5-19 Load-displacement experimental and 2D model of S4	. 141
Figure 5-20 Load-displacement experimental and 2D model of S5	. 142
Figure 5-21 Load-displacement for experimental and 2D model of P9	. 142
Figure 5-22 Load-displacement for experimental and 2D model of P10	. 143
Figure 5-23 Load-displacement for experimental and 2D model of P11	. 143
Figure 5-24 Load-displacement for experimental and 2D model of P12	. 144

Figure 5-25 Load-displacement for experimental and 2D model of P13...... 144 Figure 5-26 Strain distributions for experimental and 2D model of S1...... 145 Figure 5-27 Strain distributions for experimental and 2D model of S2...... 145 Figure 5-28 Strain distributions for experimental and 2D model of S3...... 146 Figure 5-29 Strain distributions for experimental and 2D model of S4...... 146 Figure 5-30 Strain distributions for experimental and 2D model of S5...... 147 Figure 5-31 Strain distributions for experimental and 2D model of P9...... 147 Figure 5-32 Strain distribution for experimental and 2D model of P10 ...... 148 Figure 5-33 Strain distributions for experimental and 2D model of P11...... 148 Figure 5-34 Strain distributions for experimental and 2D model of P12...... 149 Figure 5-35 Strain distributions for experimental and 2D model of P13...... 149 Figure 5-36 Load-displacement for experimental and 3D model of S1..... 150 Figure 5-37 Load-displacement for experimental and 3D model of S3...... 150 Figure 5-38 Load-displacement for experimental and 3D model of S4..... 151 Figure 5-39 Load-displacement for experimental and 3D model of S5.....151 Figure 5-40 Load-displacement for experimental and 3D model of P9...... 152 Figure 5-41 Load-displacement for experimental and 3D model of P10...... 152 Figure 5-42 Load-displacement for experimental and 3D model of P11......153 Figure 5-43 Load-displacement for experimental and 3D model of P12...... 153 Figure 5-44 Load-displacement for experimental and 3D model of P13...... 154 Figure 5-45 Strain distributions across sheet width of S1 at the crack location ... 154 Figure 5-47 Strain distributions across sheet width of S3 at the crack location ... ........ 155 Figure 5-48 Strain distributions across sheet width of S4 at the crack location ... ........ 156 Figure 5-50 Strain distributions across the width of P9 at crack location...... 157 Figure 5-52 Strain distributions across the width of P11 at crack location...... 158 Figure 5-53 Strain distributions across the width of P12 at crack location....... 158 Figure 5-54 Strain distributions across the width of P13 at crack location...... 159 

Figure 6-4 Average bond strength-bond length at different w values
Figure 6-5 Average effective bond strength-bond length, various w values
Figure 6-6 Bond length-ultimate load at different Et values
Figure 6-7 Effective length-stiffness for $w = 100 \text{ mm}$
Figure 6-8 Load-displacement, modified push-apart models, L = 200 mm
Figure 6-9 Bond length-ultimate load, various f'c
Figure 6-10 Bond length-Average bond strength, various f' <sub>c</sub>
Figure 6-11 Bond length-Average effective bond strength, various f' <sub>c</sub>
Figure 6-12 Bond width-ultimate load at different L values
Figure 6-13 Ultimate load per width vs bond length at various bond widths
Figure 6-14 Bond width-bond strength at different L values
Figure 6-15 Bond width-bond strength at different L values, $L_e = 110 \text{ mm}$
Figure 6-16 Load-Displacement, various $w_c$ , $w = 200$ and $L = 50$ mm
Figure 6-17 Load-Displacement, various $w_c$ , $w = 200$ and $L = 150$ mm
Figure 6-18 Strain E11 contour lines (a) isometric view for $w/w_c = 0.7$ (b) elevation view for $w/w_c = 0.7$ (c) elevation view for $w/w_c = 1.0$
Figure 6-19 Strain distribution across the concrete width
Figure 6-20 Stress distribution across the concrete width at ultimate load
Figure 6-21 Strain distribution across bond width, 15 kN 192
Figure 6-22 Strain distribution across bond width, $P/P_u = 0.5$
Figure 6-23 Strain distribution across bond width, $P/P_u = 0.7$
Figure 6-24 Strain distribution across bond width, $P/P_u = 1.0$
Figure 6-25 Strain difference-bond width response, $w_c = 200 \text{ mm}$
Figure 6-26 Strain distribution across FRP sheet width at 15 kN 194
Figure 6-27 FRP stiffness vs. ultimate load at various bond lengths 195
Figure 6-28 FRP stiffness vs. average bond strength at various bond lengths 195
Figure 6-29 Effective FRP stiffness vs. bond length relationship 196
Figure 6-30 Concrete strength vs. ultimate load at various bond lengths
Figure 6-31 Strain distribution for various $f_c'$ values
Figure 6-32 Concrete strength vs. ultimate load at various bond lengths 197
Figure 6-33 Load-displacement response, pull-apart models, L = 200 mm
Figure 6-34 Concrete strength vs. ultimate load, push-apart model

Figure 6-35 Strain distribution for various $f'_c$ at ultimate load, pull-apart model	199
Figure 7-1 Direct pull bond test	215
Figure 7-2 Load-displacement for direct pull tests with various unbonded lengths	216
Figure 7-3 Development of principal stresses in direct pull specimen	217
Figure 7-4 Load-displacement for direct pull models	218
Figure 7-5 Load level-displacement for direct pull models	218
Figure 7-6 Strain distribution for direct pull models, w = 25 mm	219
Figure 7-7 Strain distribution for direct pull models, w = 150 mm	219
Figure 7-8 Pull-apart bond test	220
Figure 7-9 Load-displacement for pull-apart models	221
Figure 7-10 Normilized load-displacement response for pull-apart models	221
Figure 7-11 Strain distribution for pull-apart models, w = 25 mm	222
Figure 7-12 Strain distribution for pull-apart models, w = 150 mm	222
Figure 7-13 Push-apart bond test	223
Figure 7-14 Load-displacement push-apart tests, unbonded sheet lengths	223
Figure 7-15 Load-displacement for push-apart models	224
Figure 7-16 Normilized load-displacement response for push-apart models	224
Figure 7-17 Strain distribution for push-apart models, w = 25 mm	225
Figure 7-18 Strain distribution for push-apart models, w =150 mm	225
Figure 7-19 Modified push-apart test	226
Figure 7-20 Load-displacement response for models with $L = 150 \text{ mm}$	227
Figure 7-21 Normilized load-displacement response for models with $L = 150 \text{ mm}$	227
Figure 7-22 Strain distribution for modified push-apart models, w = 25 mm	228
Figure 7-23 Strain distribution for modified push-apart models, w =150 mm	228
Figure 7-24 Load-displacement for modified push-apart tests, various arm widths	229
Figure 7-25 Load-displacement behaviour for modified push-apart with arm widths ranging from 40 to 100 mm	229
Figure 7-26 Load-displacement behaviour for modified push-apart with arm widths ranging from 100 to 300 mm	230
Figure 7-27 Strain distribution along the concrete arm in modified push-part model wi 40 mm arm width	th 230
Figure 7-28 Stress distribution along the concrete arm for modified push-part model w 40 mm arm width	'ith 231

Figure 7-29 Strain distribution in perpendicular direction to loading for modified push- apart model with 300 mm arm width	231
Figure 7-30 Stress distribution across outer concrete for modified push-part model with 300 mm arm width	ı 232
Figure 7-31 Load-displacement for all bond tests, w = 25 mm	232
Figure 7-32 Load-displacement for all bond tests, w = 50 mm	233
Figure 7-33 Load-displacement for all bond tests, w = 100 mm	233
Figure 7-34 Load-displacement for all bond tests, w = 150 mm	234
Figure 7-35 Ultimate load vs. bond width for $L = 150 \text{ mm}$	234
Figure 7-36 Ultimate load per width vs. bond width for $L = 150 \text{ mm}$	235
Figure 7-37 Ultimate load per width vs. bond width, pull-apart tests	235
Figure 7-38 Ultimate load per width vs. bond length, direct pull tests	236
Figure 7-39 Ultimate load per width vs. bond length, pull-apart tests	236
Figure 7-40 Ultimate load per width vs. bond length, push-apart tests	237
Figure 7-41 Ultimate load per width vs. bond length, modified push-apart test	237
Figure 7-42 Strain distribution for models with FRP width of 25 mm at 0.6 $P_u$	238
Figure 7-43 Strain distribution for models with FRP width of 25 mm at 0.8 $P_u$	238
Figure 7-44 Strain distribution for models with FRP width of 25 mm at 0.9 $P_u$	239
Figure 7-45 Strain distribution for models with $w = 25$ mm at ultimate load2	239
Figure 7-46 Strain distribution for models with $w = 150$ mm at ultimate load	240
Figure 7-47 Strain distribution for models with FRP width of 25 mm at 4 kN	240
Figure 7-48 Strain distribution for models with FRP width of 150 mm at 24 kN	241
Figure 8-1 Schematic sketch and derivation of development length	256
Figure 8-2 Effective length-stiffness for $w = 100 \text{ mm}$ (Section 6.2)	256
Figure 8-3 Effective length-stiffness for w = 100 mm	257
Figure 8-4 Effective length-Bond width relationship (Section 6.2)	257
Figure 8-5 Effective length vs. square root of bond width relationship	258
Figure 8-6 Test vs. predicted effective length	258
Figure 8-7 Concrete compressive strength vs. ultimate load (Section 6-5)	259
Figure 8-8 Ultimate load vs. FRP sheet bond width (Section 6.3.1)	259
Figure 8-9 Ultimate load vs. Bond length (Section 6.2)	260
Figure 8-10 Ultimate load vs. bond area	260
Figure 8-11 Ultimate load test to predicted values, Equation 8-7	261

Figure 8-12 Ultimate load vs. square root of stiffness (Section 6-4)	261
Figure 8-13 Ultimate load test to predicted ratio vs $\sqrt{E_p t_p}$ /wL <sub>e</sub> , all tests	262
Figure 8-14 Ultimate load test to predicted values, Equation 8-9	262
Figure 8-15 Ultimate load test to predicted values, Equation 8-10	263
Figure 8-16 Ultimate load test to predicted values, Equation 8-11	263

## LIST OF ABBREVIATIONS AND SYMBOLS

Hereafter the abbreviations and symbols used in this thesis. Symbols are also defined where they first appear within the thesis.

#### Abbreviations

3D Three-dimensional

ACI American Concrete Institute

C3D8R Eight-node cube elements with reduced integration

CFRP Carbon Fibre Reinforced Plastic

CPS4R Four-node plain stress solid elements with reduced integration

FRP Fibre Reinforced Plastic

GFRP Glass Fibre Reinforced Plastic

ISIS Intelligent Sensing for Innovative Structures

kN Kilo Newton

LVDT Linear Variable Differential Transducer

mm Millimetre

MPa Mega Pascal

MS Microsoft

MTS Material Testing System

RC Reinforced Concrete

με Micro Strain

#### **Arabic Symbols**

- a Distance of anchor sheet from crack, mm
- A Coefficient
- b Anchor sheet extension beyond the tested sheet, mm
- b<sub>a</sub> Width of adhesive, mm
- b<sub>c</sub> Width of the concrete member, mm

- b<sub>p</sub> Width of the bonded plate, mm
- c Length of initial failure, mm
- c<sub>f</sub> Constant
- d<sub>1</sub> Distance of the metal sheet edge from the concrete surface, mm
- E Modulus of Elasticity, MPa
- E<sub>1</sub> Modulus of elasticity in main direction in the plane of sheet, MPa
- E<sub>2</sub> Modulus of elasticity in secondary direction in the plane of sheet, MPa
- E<sub>a</sub> Young's modulus of adhesive, MPa
- E<sub>c</sub> Young's modulus for concrete, MPa
- E<sub>c</sub> Modulus of elasticity of the concrete, MPa
- E<sub>p</sub> Modulus of elasticity of FRP, MPa
- f Critical interface stress criterion, MPa
- f'<sub>c</sub> Concrete compressive strength, MPa
- F<sub>1</sub> Forces in the composite material plate at gauge 1, kN
- f<sub>ct</sub> Tensile strength of concrete, MPa
- f<sub>cy</sub> Concrete compressive yield stress, MPa
- f<sub>su</sub> Steel ultimate strength, MPa
- G<sub>12</sub> Shear modulus in 1-2 plan, N.mm/mm<sup>2</sup>
- $G_f$  Fracture energy, N.mm/mm<sup>2</sup>
- g<sub>i</sub> Strain value at point i
- I<sub>p</sub> Moment of inertia of FRP plate, mm<sup>4</sup>
- k Length factor
- k<sub>p</sub> Width ratio or geometric factor
- L Bond length, mm

- L' Unbonded FRP sheet, mm
- Le Effective bond length, mm
- L<sub>ea</sub> Anticipated effective length, mm
- P Applied load, kN
- P<sub>u</sub> Ultimate load or bond capacity, kN
- R Average bond resistance, MPa
- t<sub>a</sub> Thickness of adhesive, mm
- t<sub>c</sub> Thickness of the concrete, mm
- t<sub>p</sub> FRP sheet thickness, mm
- u<sub>o</sub> Displacement, mm
- w FRP sheet bond width, mm
- w<sub>c</sub> Concrete specimen width, mm

#### **Greek Symbols**

- v Poisson's ratio
- $v_c$  Poisson's ratio for concrete
- $\Delta L$  Distance between gauges, mm
- $v_s$  Poisson's ratio for steel
- $\psi$  Coefficients obtained from curve fitting the test results
- $\epsilon_{cp}$  Concrete absolute plastic strain
- $\varepsilon_{sp}$  Steel plastic strain at ultimate strength
- $\phi$  Angle of failure surface
- σ Critical normal stresses, MPa

- $\sigma_f$  Limiting normal stress failure value, MPa
- $\sigma_n$  Observed normal stress carried across the interface, MPa
- $\tau$  Critical interface shear, MPa
- $\tau_b$  Average bond strength
- $\tau_f$  Limiting shear stress failure value, MPa
- $\tau_n$  Observed shear stress carried across the interface, MPa

## **1. INTRODUCTION**

#### 1.1. Background

A common rehabilitation solution used to repair concrete bridges employs steel plates bonded and/or mechanically fastened to the concrete surface. One of the main advantages of the steel plating rehabilitation method is that all work can be carried out from beneath the superstructure and it is not necessary to disturb the deck or perform substantial drilling. However, the weight of the plates makes the bonding work difficult. Generally they must be held in place under pressure until cure is complete, to ensure a good bond. The proper formation of joints was a problem, due to the limited delivery lengths of the plates. Also of prime concern is the possibility of corrosion at the steel-concrete interface due to moisture migration at this sensitive location, which in turn leads to the deterioration of the bond at the steel-concrete interface. The problem is even more severe for bridges where de-icing chemicals are commonly used in cold seasons.

An effective way of eliminating the corrosion problem is to replace the steel plates with corrosion-resistant synthetic materials. In the last 20 years a new family of materials has gained increasing exposure in civil engineering applications, namely Fibre Reinforced Polymer (FRP) materials. These materials, which were originally developed and applied in aeronautics, space, and sports and recreation industries, posses certain properties that make them superior to steel in some respects. The FRPstrengthening technique has found wide acceptance among researchers and engineers today in many parts of the world, and is no longer considered to be a new technique for certain types of strengthening jobs (Triantafillou 1998).

Although FRP sheets have been widely applied to rehabilitate and upgrade concrete columns, beams and slabs, no rational design method with FRP sheets has yet been established. This is because complete understanding of the effects bonded FRP materials have on the performance of structural systems has not yet been achieved.

1

Recent developments in composite material technologies, with numerous new fibres and resin systems emerging in the construction industry, require not only a basic knowledge of the material properties of each new product but also their interfacial behaviour when applied as external reinforcement. The local bond strength when delamination occurs and the average bond strength are likely dependent on many factors such as the FRP bonded length, width, and sheet stiffness as well as concrete strength and surface condition. To develop rehabilitation procedures for concrete structures involving externally applied FRP sheets, a good understanding of the behaviour and strength of the bond is necessary.

#### 1.2. Scope

Unidirectional FRP sheets externally bonded to concrete are investigated in this research with the stress condition being tensile stress in the direction of the fibres. The failure mode investigated is the bond failure at the crack location.

Numerous bond stress curves and equations are proposed in the literature. Each describe well the data from which they were defined. However, there is large discrepancy and scatter between these curves. The expected reason is the different bond test methods and loading arrangements used to study the bond behaviour.

Several researchers have proposed different models to predict the bond strength and the effective length. The models can be classified into three categories: empirical models based on regression analysis of test data, fracture mechanics based models, and design models, which use simplified assumptions. A model needs to be recommended or a new model proposed based on the evaluation of the proposed models in the literature, an extensive parametric study, and the test results existing in the literature. This is the thrust of the investigation reported in this thesis.

#### **1.3.** Objectives

There are five main objectives in this thesis. The first is to have a better understanding of the bond behaviour between fibre reinforced polymer (FRP) sheets and concrete. This was accomplished by investigating the role of the main parameters affecting the bond behaviour through experimental and numerical analyses. The parameters studied were the FRP sheet bond length, bond width, stiffness, and concrete strength. For practical implementation of the configuration and detailing of the bonded sheets, the anchor requirements for FRP sheets were also investigated experimentally.

The second objective is to produce charts showing the effect of the main parameters on the bond behavior and to propose guidelines for design purposes. This is to be done by producing a numerical model of the debonding type failure that accurately predicts the load-displacement behavior and the strain distribution in the bonded FRP sheets. The charts showing the effect of the different parameters on the bond behavior are then produced through an extensive parametric study.

The third objective is to develop a rational standard test method for FRP sheets bonded to concrete. This objective arose from the literature review that showed that the type of bond test method used in predicting the bond behaviour had an effect on the test results. This is to be done by analyzing the various conventional bond test methods using numerical models and evaluating the advantages and drawbacks of each method. Based on the investigation, a standard test method is recommended.

A fourth objective is to develop regression analysis based models predicting the effective bond lengths and the ultimate bond capacity of FRP sheets bonded to concrete.

#### **1.4.** Thesis Outline

Chapter 2 presents an overview of the FRP material, the advantages and disadvantages of using the material, and the forms in which they are manufactured. The strengthening schemes for flexure and shear and the possible failure mechanisms are also presented. A lack of understanding of the bond behaviour is seen through a discussion of the effects of the parameters proposed by previous research. Reviewing the different conventional bond test methods used in different bond research points out the need to develop a rational standard test method for FRP sheets bonded to concrete. Finally the

3
existing bond strength and FRP effective length models are discussed. This also shows the need to evaluate the existing models and propose or develop a model, that gives more reliable results.

Two experimental test series were conducted in this research using two different types of bond test methods. One series used a modified push-apart test and the other used a pull-apart test. A detailed description of each test specimen is presented in Chapter 3. The test set-up, instrumentation, and the property of the materials used are also presented in the chapter.

The test results and observations of both experimental series are presented in Chapter 4. The discussion and analysis of the test results and observations are also presented in the same chapter.

The numerical model geometry, material properties, loading, and boundary conditions are presented and discussed in Chapter 5. The concrete-FRP interface model of the debonding behaviour is also discussed in this chapter. The numerical model simulates the modified push-apart and pull-apart test methods and the output is compared to the experimental test results.

The parametric study is presented in Chapter 6. The main parameters affecting the bond behaviour between concrete and FRP sheets are investigated using the proposed numerical model. These parameters are the bond length, bond width, sheet stiffness, and the concrete strength. Charts showing the effect of each parameter are developed and analysis of the effect of each parameter on the bond behaviour is discussed.

Chapter 7 details an investigation of the comparison between different bond test methods using numerical models. The relationship between the results of each method is discussed and evaluated. Based on the analysis, two test methods are proposed as standard test methods for FRP sheets bonded to concrete based on the analysis.

Regression analysis based models predicting the effective bond length and the bond capacity of FRP sheets bonded to concrete are detailed in Chapter 8. Comparisons with other proposed models from the literature were performed using experimental test results from this study and from the literature. Based on the comparison, the regression analysis based model proposed in this study showed more accurate and reliable predictions than the models proposed in the literature.

A summary and the conclusions of the research conducted are presented and recommendations for changes to the FRP design codes and for future research are proposed in Chapter 9.

# 2. LITERATURE REVIEW

### 2.1 Introduction

This chapter presents an overview of FRP materials, the advantages and disadvantages of using the material, and the forms in which they are manufactured. The strengthening schemes for flexure and shear and the possible failure mechanisms are also presented. The parameters affecting the bond behaviour are discussed and the findings and conclusions of the previous studies are summarized. A review of the different conventional bond test methods used in different bond research is presented. Finally the most significant existing bond strength and FRP effective length models are presented in this chapter.

#### 2.2 Fibre Reinforced Polymers (FRP)

FRP materials consist of fibres of high strength and high modulus of elasticity embedded in a matrix of polymeric material at distinct interfaces. Both fibres and matrix retain their physical and chemical identities and they also produce a combination of properties that cannot be achieved with either acting alone. In general, fibres are the principal load-carrying elements, providing the major contribution of strength and stiffness of the composite and most of its desirable properties. The surrounding matrix keeps the fibres in the desired location and orientation, acts as a load transfer medium, and protects the fibres from environmental damages (e.g. temperature, and humidity). The properties of the composite material depend largely on the properties of the fibres and matrix used, and the relative proportion in which each component is present (Drimoussis and Cheng, 1994).

Fibres may be oriented randomly within the matrix, in certain prescribed alignments, or unidirectionally. As all the attractive properties of the composite depend on the fibres, the greatest benefit is gained when it is loaded in the direction in which the fibres are oriented. When the load is applied at a different orientation, away from the fibres, the properties of the matrix begin to dominate. This effect can

be seen in Figure 2-1. Although shown in particular for a glass FRP (GFRP) material, this is typical of FRP's in general (Holloway and Leeming, 1999).

# 2.2.1 Types of Fibres

There are many types of fibres, some of which are not commonly used in civil engineering projects (e.g. polyethylene, boron, and ceramic) and some of which are commonly used because of their properties. These are glass, carbon, and aramid fibres.

Glass fibres are commonly used because there is more experience with them and because they are relatively less expensive than the carbon and aramid fibres. Glass fibres are available in several types according to the required application (e.g. E-glass used for electrical applications, C-glass for chemical applications). Some of the important properties of glass fibres are a low tensile modulus and high strength, a softening point of 800-1000°C, high insulating properties, sensitivity to alkaline environment, and a relatively low resistance to sustained loads and moisture. The fibres are coated with an agent to minimise abrasion damage but fibres can be easily damaged in handling.

Carbon fibres are gaining interest for use in structural applications despite their higher cost compared to glass fibres because some of their properties are superior to those of glass in some areas (e.g. durability in marine environments). Carbon fibres have two main types namely PAN-based carbon and Pitch-based carbon. PAN-based carbon is derived from acrylic nitrile and has a modulus of elasticity of the order 200 GPa and tensile strength of the order of 3000 MPa. Pitch-based carbon is derived from coal tar and has a modulus of elasticity of the order 50 GPa and tensile strength of the important properties of carbon fibres are excellent fatigue resistance, high conductivity to heat and electricity, and excellent moisture and chemical resistance. Carbon fibres are also sensitive to galvanic corrosion and have an elastic modulus that is comparable to that of steel (Drimoussis and Cheng, 1994, and Tadros et al., 1998).

Aramid fibres are used in fewer applications than glass and carbon fibres. They have low tensile modulus and high strength and their electrical conductivity is low. Temperature and alkaline solutions increase creep strains and moisture has the effect of decreasing tensile properties. Figure 2-2 shows a typical comparison of responses of different fibre types compared to steel properties.

### **2.2.2** Types of Matrix

There are two main different types of resin matrix, namely polymer and metal. Polymer matrix is the one used in civil engineering applications, where it has two main types, the first is thermoset matrices (e.g. epoxy, polyester, and vinyl ester) and the other is thermoplastic matrices (e.g. polyether ether ketone, polyphenylene sulfide, and polysulfone). Thermoset matrices are most commonly used because of preferable properties such as high modulus and strength and superior characteristics under longterm loading. Thermoset matrices undergo irreversible chemical change when cured but are stable under high temperatures and aggressive environments.

On the other hand thermoplastic has undesirable properties such as its lowmodulus and strength and it is also susceptible to softening at high temperatures. Due to its disadvantages, it is not permitted for tendons and primary reinforcement according to the CSA-S6 (1988).

### **2.2.3 Forms of FRP Materials**

FRP materials used in structural engineering applications can be supplied as finished end products in a variety of pultruded structural shapes or in the form of bars, tendons, wire meshes, cables, plates, sheets or as micro fibres to be used as additives to concrete mixes. However, the properties of the composites will be different, depending on the size of the fibres and their orientation. Using longer fibres significantly increases both the strength and the modulus of elasticity. Continuous fibres are used as reinforcements when high performance and weight reduction is

needed and when thermal expansion is to be kept at very low levels (Donnet and Bansal, 1990).

External reinforcement of concrete structures with bonded FRP requires the use of either sheets or plates. Sheets are made of dry or pre-preg fibres to be fully impregnated with resin during installation. They are flexible in that they can easily follow the contour of a surface and bend at corners. Plates are fully cured material systems manufactured by pultrusion or other suitable techniques such as lay-up. The application technique is similar to steel plates. Plates, independently of their thickness, have their own rigidity. In the case of plate adhesion, the adhesive layer is perhaps thicker and more easily identifiable. Changing the thickness of the FRP plate significantly modifies the crack distribution and changes the failure mechanism (Arduini et al., 1996). In this study only FRP sheets are investigated.

## 2.2.4 Advantages of Using FRP

FRPs have high strength-to-weight and high modulus. Because of their strength and toughness they can be used in thin sections, leading to lightweight and easy handling and installation. FRPs are non-conductive, non-magnetic, and have excellent corrosion resistance, eliminating some of the disadvantages of using steel. They also have high fatigue resistance. FRPs have low relaxation creep, which leads to lower long-term deflections. They also have a longer useful life, which in turn gives a lower life cycle cost. FRPs are easy to form, where they can be bonded to curved or irregular surfaces. They can also be supplied in long lengths, thus reducing the need for lapping at joints. FRPs are light and easy to handle on a construction site so that there is no need for special equipment (Cheng, 1998, and Tadros et al., 1998).

#### 2.2.5 Disadvantages and Limitations of using FRP

FRP materials have certain advantages, but they also have limitations, which must be circumvented, if they are to be used for best results. FRPs have higher first cost. But especially in bridge works, the cost savings due to labour savings can be substantial if the time required for retrofit and traffic disruption could be minimised.

FRPs have lower modulus of elasticity than steel, which means lower stiffness and higher deflection. FRPs have unknown long-term durability in construction, because these materials are relative newcomers and their history is short.

There is relative lack of experience in using FRPs in civil engineering projects. As owners, designers, and contractors become more familiar with this technology, it is envisioned that the number of applications will rapidly increase. Design codes and standards for the use of FRPs are unavailable, where many manufacturers produce their own proprietary products. In North America work is in progress towards producing FRP design codes. Draft documents are being circulated within technical committees and should be completed in the near future (Nanni, 1997).

FRPs have lack of apparent ductility and low fire resistance. Because the addition of FRPs having linear behaviour up to fracture eliminates the possibility of flexural tension softening, there is concern that beams reinforced in this manner would have inadequate ductility (Bonacci, 1996). Fires of the intensity that occur in buildings can destroy all plastics. Some burn readily and ought not to be employed in structural engineering; others are slow burning or self-extinguishing.

Any particular FRP material is likely to posses some but not all of both advantages and disadvantages. The designer must decide which properties are most important for a particular application and then choose the appropriate material.

### 2.3 Adhesive Used to Bond FRP Sheets to Concrete

The success of this strengthening technique is critically dependent on the performance of the epoxy used to bond the FRP plates/sheets to the concrete surface. Numerous types of epoxies with a wide range of mechanical properties are commercially available.

The adhesive must be compatible with both FRP plates/sheets and concrete. Researchers showed that rubber-toughened epoxies are particularly suited for this

application. The adhesive should have sufficient stiffness and strength to transfer the shear force between the concrete beam and the composite plate. The adhesive should be tough enough to prevent brittle bond failure caused by cracking of the concrete and have good resistance to salt and moisture and be usable under the prevailing environmental conditions (Saadatmanesh and Ehsani, 1991a, Ritchie, 1991, and Chajes et al., 1996).

Chajes, et al. (1996) studied the effect of the type of adhesive on the average bond strength. They concluded after studying four types of different adhesive that an "off-the-shelf" epoxy could be effectively used to bond composite material plates to concrete. They also pointed out that the use of ductile adhesives (i.e. those having a low stiffness and a large strain to failure) leads to less effective bond.

Swamy, et al. (1986) tested various lap shear specimens with steel plates having adhesive thickness ranging from 0.5 mm to 3.0 mm and found no variation in results due to adhesive layer thickness.

## 2.4 Strengthening of Reinforced Concrete Beams

FRP sheets and plates have been used to strengthen various types of Reinforced Concrete (RC) members. The effectiveness of the strengthening reinforcement, that is, the load carried by the FRP at the ultimate limit state, depends on its failure mechanism, which, in turn, depends on various factors (Triantafillou, 1998). The common FRP strengthening methods using bonded sheets and plates used to strengthen concrete girders in flexure and shear will be presented.

#### **2.4.1** Flexural Strengthening of RC Beams

One or more layers of FRP sheets or plates are externally bonded to the tension surface of the RC beam in order to strengthen it in flexure, as shown in Figure 2-3. When FRP plates are used, special attention shall be given to the debonding and end peeling problems. Vertical wrapping is an effective way for anchoring the flexural

FRP sheet or plate. Taping at the end of the FRP sheets and the use of GFRP vertical wrapping could also be a solution (Cheng, 1998).

Tumialan et al. (1999) carried out an experimental investigation and concluded that strengthening of RC beams with externally bonded CFRP sheets was effective and lead to increases in flexural strengthening between 30% and 60%. Arduini and Nanni (1997a) and Cheng (1998) also stated that the load carrying capacity of RC beams increased and the deflection decreased at service loads due to flexural strengthening with FRP sheets as illustrated in Figure 2-4.

### 2.4.2 Shear Strengthening of RC Beams

A RC member must be designed to develop its full flexural strength to insure a ductile flexural failure mode under extreme load. Hence a beam must have safety margins against other types of failure modes that are more dangerous and less predictable than flexural failure. Shear failure or diagonal tension failure is a failure mode that has a catastrophic effect, should it occur. If a RC member deficient in shear strength is overloaded, shear failure may occur suddenly without advance warning of distress. Therefore, a RC member must have sufficient shear strength to insure ductile flexural failure.

FRP sheets are used to wrap concrete sections in the transverse direction in order to reinforce diagonal tension cracks in much the same way as steel stirrups. The most effective method of shear strengthening with FRP sheets is to wrap the entire cross section of the beam with FRP. Typically, this is not practical from a construction standpoint. The presence of monolithic slabs or other supported elements often prevents wrapping the sheet around the top of the section.

It is more difficult to add shear reinforcement to concrete beams, especially when the beams are not isolated but are instead part of a floor beam system. To best provide external shear reinforcement, it is desirable to encase the beam web with external reinforcement. This can be more easily accomplished using a flexible

material, such as a FRP sheet, than using a plate, since the sheet can conform to any arbitrary shape (Chajes et al., 1995).

Typical FRP configurations for shear strengthening of concrete beams are shown in Figure 2-5. The external reinforcement in the figure is in the form of bonded sheets extending into the compression zone (U-jacket), where bonding may be supplemented by mechanical fastening, as in Figure 2-5(b). Another possibility is that of Figure 2-5(d), where FRP sheets are wrapped around the web of the beam (Triantafillou, 1998). Horizontal anchor sheets can be used to prevent peeling at the top of the shear reinforcement. The load carrying capacity is increased due to shear strengthening as shown in Figure 2-6.

Shear strengthening reinforcement may be in the form of a continuous jacket or as spaced strips (discontinuous strapping). Discontinuous strapping, as shown in Figure 2-5, avoids trapping water inside the girder, allows for evaporation, is easier for inspection, and is effective in optimizing the amount of material used.

Because FRP is an anisotropic material with high strength in the direction of the fibres, the fibres may be oriented in such a way to best reinforce diagonal tension cracks. This is achieved by the use of inclined strips, Figure 2-5. However, vertically oriented plies are easier to install and may reduce the total length of the wrap. Cheng (1998) concluded that vertical and inclined shear strengthening yield almost similar increase in shear strength but stated that lower strain was observed in stirrups for the beams strengthened with inclined straps.

It has been found that the use of biaxial FRP reinforcement increases the overall performance of the system (Triantafillou, 1998). Biaxial FRP reinforcement is achieved by placing two unidirectional FRP plies in mutually perpendicular directions, as shown in Figure 2-5. The ply in the primary direction acts to provide most of the reinforcement. While the ply in the secondary direction limits shear crack openings and provides anchorage for the ply in the primary direction.

## 2.5 Possible Failure Mechanisms

The failure mechanisms observed and recorded in RC beams strengthened with FRP sheets could be categorised into two main groups depending on whether they were strengthened in flexure or in shear.

### **2.5.1** Failure Mechanisms for Beams Strengthened in Flexure

All failure modes show at first a linear-elastic behaviour followed by a crack in the mid-span region of the beam. Thereafter, a large non-linear phase is recorded with the development of numerous flexural cracks, in this phase, the strain in the FRP plate increased considerably, as well as the beam deflection. Changing the thickness of the FRP plate, the bonded length, or adding FRP shear reinforcement significantly modifies the crack distribution along the beam and changes the failure mechanism (Arduini et al., 1997a). In the following the failure mechanisms are discussed in detail.

**FRP rupture** in tension occurs when the FRP strain exceeds the ultimate value in the zone of maximum moment or at stress concentrations (e.g. corners or debonded areas). This occurs after large deflection of the member has taken place, which shows good structural performance (Figure 2-7a). From an economical point of view, the rupture of the FRP plate or sheet seems to be desirable because it means that all the resources of FRP are utilised. Appreciable strength gains for fracture type failures are only possible when the original beam is very lightly reinforced (Bonacci, 1996).

*Concrete crushes* when the concrete compressive strain exceeds its ultimate value in the cross-section of maximum moment as shown in Figure 2-7b. This occurs after a large deflection of the member, which is a better structural performance. Bonacci (1996) stated that specimens failing in flexural compression preserved considerably more of the deflection capacity of the conventionally reinforcement control beams than were the case for the other failure modes.

**Peeling-off failure** occurs due to failure at the concrete-adhesive interface, which is debonding between FRP and concrete. The bond between the sheet and the concrete may fracture in a sudden manner as the result of catastrophic propagation of a crack along the FRP-concrete interface. Possible reasons for the existences of such a crack are imperfections in the spreading of the adhesive, flexural cracking in the concrete, peeling-off of the plate when the concrete tensile face is not perfectly flat, or fatigue loads. This failure mechanism is facilitated by low ultimate strain values of the adhesive. As the adhesive fails in uniaxial tension in a given zone, there is no shear stress or normal stress transfer between the FRP and the concrete. At the ends of this zone, a high shear and normal stress transfer is needed to balance the force in the unbounded FRP plate (Triantafillou and Plevris, 1992). This failure mechanism is shown in Figure 2-7c.

*Concrete cover delamination* results from a combination of shear and normal tensile stress in the concrete in the plane of the longitudinal steel bars. This failure mechanism initiates at the ends of the FRP plate, results in the propagation of a horizontal crack, and causes separation of the concrete cover, as shown in Figure 2-7d. It is also brittle and occurs at load values lower than expected with conventional design equations. The occurrence of shear-tension indicates that it is no longer possible to increase the flexural capacity of the member by increasing the FRP thickness.

In the first two modes of failure, the ultimate strength of the structural member can be predicted easily by following conventional RC flexural theory. However, whenever the mode of failure is peeling-off or concrete cover delamination, the strengthened member is not able to reach its ultimate strength. From the point of view of design, it is important to recognize this premature type of failure and determine algorithms for its prediction. These failures are sudden, brittle and occur with no warning (Tumialan et al., 1999).

#### 2.5.2 Failure Mechanisms for Beams Strengthened in Shear

Once shear forces develop inclined cracks in the concrete, high tensile stresses are developed in the portions of the FRP sheet that bridge these cracks. The tensile stresses in vertically oriented FRP sheets are a result of the vertical separation of rigid bodies of concrete on either side of the crack. These tensile stresses must be transferred to the concrete on each side of the crack by interfacial bond stresses. If this interfacial bond is compromised before rupture of the FRP sheet, a delamination failure occurs. The delamination failure could take the form of debonding at the concrete-adhesive interface or shearing of the concrete below the concrete-FRP sheets interface, as discussed in Section 2.5.1. The tensile fracture of the FRP sheet at a stress, which may be lower than the tensile strength of the FRP material, occur due to stress concentrations (e.g. at corners or debonded areas).

Whether delamination or fracture of the FRP sheet will occur first depends on the bond conditions, the available anchorage length and/or the type of anchorage at the FRP ends, the thickness of the laminates and others. In many cases, the actual failure mechanism is a combination of FRP delamination at certain areas and fracture at others. In general, as the FRP become stiffer and thicker, delamination dominates over tensile fracture (Triantafillou, 1998).

Debonding and concrete shearing underneath the FRP sheet failures are more difficult to characterize and analyse than the other potential modes. They depend on factors that are foreign to analyses of conventional nature such as mechanical preparation of the concrete surface before the application of the epoxy and the sensitivity to faulting motions along member cracks propagating to the tension face (Bonacci, 1996).

## 2.6 Conventional Bond Test Methods

There are several different test arrangements conventionally used for FRP bond research. The main categories of bond tests based on the type of load on the FRP sheets are direct tension, bending, and shear type tests.

### 2.6.1 Direct Tension Bond Tests

The direct tension type bond test shown in Figure 2-8 measures the normal bond stress between the FRP sheet and concrete. The FRP sheet is bonded to the surface of a concrete block. The block is secured in place and an apparatus that applies the load is bonded to the FRP sheet. The tensile load is applied perpendicular to the direction of the fibres through the bonded apparatus. The bond strength is measured by dividing the ultimate tensile load by the bond area. The bond area is usually square  $40 \times 40$  mm, which is relatively small compared to the other test methods. The test results conducted by Horiguchi and Saeki (1997) show that the bond strength recorded using this test method is higher than both the bending and shear type specimens as shown in Figure 2-9. The corresponding type of application is rare and the stress condition is undesirable.

#### 2.6.2 Bending Bond Tests

The bending type bond test shown in Figure 2-10 uses a simply supported RC beam specimen with FRP sheets bonded to the bottom tensile face. The beam has a rectangular or inverted T-shape cross-section. The beam is saw cut in the centre up to about half the depth from the tensile lower surface to simulate a crack at the centre of the beam. The specimen has either a notch or a hinge in the centre of the compression side as shown in Figure 2-10, in order to ease the bending of the specimen and to control the distribution of the internal forces (De Lorenzis et al., 2000). The FRP sheet is bonded to the lower tensile face of the beam. A four-point loading is usually adopted in this test method. As the load increases the concrete cracks at the saw cut location and both tensile and bending stresses are transferred to the FRP sheet.

### 2.6.3 Shear Bond Tests

The shear type test specimens better represent the bond in shear strengthening of RC members. The shear type bond tests are grouped in this study into three categories that are based on the method of load application. These categories are direct pull bond tests, indirect pull-apart tests, and indirect push-apart tests.

### 2.6.3.1 Direct Pull Bond Tests

The single-lap bond test set-up, shown in Figure 2-11, is a direct pull bond test. This method is one of the simple test methods used to calculate the bond strength between concrete and FRP sheets. It consists of a concrete block secured to the fixed head of a testing machine with one end of an FRP sheet bonded to one side and the other end gripped in the moving head of the testing machine. The load is applied directly to the FRP sheet when the moving head rises. This test requires special attention during bonding the FRP sheet and then aligning it in the testing machine. The width of the sheet is restricted to 25 mm due to grip problems.

From the test results of Chajes et al. (1996) using this test method it was indicated that the maximum interfacial bond stress would be 5.5 MPa. In this test arrangement the concrete block encounters compression while the FRP is in tension as shown in the free body diagram in Figure 2-12. Bizindavyi and Neale (1999) used the same test specimen but bonded the sheet at a distance away from the loaded edge to overcome the high compression at the loaded end of the concrete block.

#### 2.6.3.2 Indirect Pull-Apart Tests

The specimen in this method consists of two concrete prisms with a steel rebar embedded in the centre of each prism. The prisms are aligned and the FRP sheets are bonded to two opposite sides of the prism. The specimen is placed in a testing machine with the machine heads gripping the rebar. As the head moves the force generated in the embedded bars is transferred to the FRP sheets through bond between the concrete and the FRP sheets, as shown in the free body diagram in Figure 2-13(b). Maeda et al. (1997) indicated that the highest average bond stress calculated using this test method was 2.94 MPa. Bonding the sheets, and handling the specimen after bonding the sheets is problematic in this test arrangement due to the gap between the concrete prisms. The load in the FRP sheets, and in return the bonded area, is limited to the bond slip strength of the steel rebar.

#### **2.6.3.3 Indirect Push-Apart Tests**

The push-apart bond test, shown in Figure 2-14(a) only requires the use of a hydraulic jack to apply the load, which makes it suitable for field verifications in retrofit situations. It is clear from the free body diagram shown in Figure 2-14(b) that the concrete is in compression while the FRP is in tension, similar to that of the direct pull test. Special care is also required with aligning the concrete blocks during bonding the sheets and during the test set-up.

Several researchers used this type of specimen. The results from Drimoussis and Cheng (1994) and Alexander and Cheng (1997) showed a maximum average bond stress of 2.6 and 2.3 MPa respectively while Dolan et al. (1998) had a maximum of 3.6 MPa. The difference is attributed to the different material properties and bond lengths used in the studies.

There are many tests conducted using the previous conventional bond test methods and from the previous discussion it is clear that there is a need to assess the performance of these methods and to develop a rational and simplified standard test method. An attempt to relate the different types of tests in order to be able to use their results in developing design equations and guidelines based on a broad database of consistent results is also required.

### 2.7 Factors Affecting Bond Behaviour

Despite the diversity of reinforcing materials used, the different concrete compressive strengths investigated, and the different specimen geometry and loading employed, there is a rather general agreement on the main factors affecting the bond strength between FRP and concrete. These factors are the FRP bond length, bond width, stiffness, concrete tensile strength, and concrete surface preparation. Certain other factors have also been reported to have a significant effect on the bond behaviour such as the test method and the anchorage type and detail.

#### 2.7.1 FRP Sheet Bond Length

The effect of the bond length of FRP sheets bonded to concrete on the bond behaviour has been the focus of study by several researchers. Drimoussis and Cheng (1994) tested push-apart specimens with bond lengths of 100 mm, 200 mm, and 300 mm. They observed that there was no significant change in the test results due to the increase in length from 200 mm to 300 mm.

Chajes et al. (1996) conducted a series of direct pull tests with FRP bond lengths ranging from 50 to 200 mm. They concluded that there exists a development length for a joint beyond which no further increase in failure load can be achieved (i.e. effective length,  $L_e$ ). They also concluded that the ultimate capacity ( $P_u$ ) is directly proportionate to the bond length (L) but with the effective length as a limit. The relationship is given by

$$P_u = RL \qquad L < L_e \tag{2-1a}$$

$$\mathbf{P}_{\mathbf{u}} = \mathbf{R}\mathbf{L}_{\mathbf{e}} \qquad \mathbf{L} \ge \mathbf{L}_{\mathbf{e}} \tag{2-1b}$$

in which R is the average bond resistance defined as:

$$R = \frac{F_1 - F_2}{\Delta L}$$
(2-1c)

Here  $F_1$  and  $F_2$  are the forces in the composite material plate at gauges 1 and 2, and  $\Delta L$  is the distance between gauges 1 and 2. Chajes et al. (1996) also stated that

additional experimental and analytical research is needed and that the plate width effect needs to be investigated.

Alexander and Cheng (1997) studied the effect of the bond length on the bond behaviour using a series of sixteen push-apart tests. The bond lengths investigated ranged from 50 mm to 175 mm. They proposed a bond stress vs. bond length relationship in which the effective bond length is 110 mm.

Brosens and van Gemert (1997) performed direct pull tests with a constant bond width of 80 mm but with bond lengths ranging between 150 mm and 250 mm. They concluded that an increase in the bond length increased the failure load, but stated that when the bond length exceeds a critical length, the fracture load remains constant. They stated that the value of the effective length could be determined experimentally and that in their case the value was larger than 275 mm. They also concluded that the influence of the bond length decreases at longer lengths.

Another study by Maeda et al. (1997) was conducted on the bond mechanism of CFRP sheets bonded to concrete using pull-apart tests. They concluded that bond lengths above 100 mm did not change the ultimate load, implying the existence of an effective bond length of 100 mm. They also stated that the effective bond length decreases as the FRP sheet stiffness increases.

Horiguchi and Saeki (1997) studied pull-apart specimens with CFRP bond length of 40, 100, and 200 mm and with a fixed bond width of 75 mm. They concluded that the effective bond length was 60 mm. Täljsten (1997) carried out direct pull tests on steel and CFRP bonded plates and came to the same conclusion of the existence of an effective bond length.

From the above it is clear that there is a general agreement on the existence of an effective bond length above which no increase in the failure load is achieved. The value of the effective bond length proposed varies significantly from one study to the other, ranging from 60 mm to over 275 mm. The variation is mainly due to the different configurations, material properties, and test methods used. A detailed study on the parameters affecting the effective length is required.

### 2.7.2 FRP Sheet Bond Width

Brosens and Van Gemert (1999) developed a bond strength model that included the effect of the FRP bond width. The width effect was introduced through a coefficient that includes the ratio between the FRP sheet width to the concrete specimen width. They also concluded through pull-apart tests with CFRP sheet widths of 80 mm and 120 mm and various numbers of plies that the anchorage length (effective length) is nearly independent of the sheet width.

Ueda et al. (1999) studied the bond width effect using a series of pull-apart bond tests. The width of the CFRP sheets ranged from 10 mm to 200 mm. FRP sheets with bond widths less than 40 mm failed by sheet breakage. They concluded that the FRP sheets with narrower width indicated greater bond strength than sheets with wider width. They also observed that the failure zone is wider than the width of the FRP sheet.

De Lorenzis et al. (2000) studied the effect of the width of CFRP sheets on the bond behaviour using bending tests. The widths of the sheets studied were 50 mm and 100 mm. The findings showed that doubling the width doubled the failure load, which suggests that the width did not influence the average bond strength.

Again the limited number of studies and the discrepancies between them are apparent, which shows the need to study the effect of the bond width on the bond behaviour.

## 2.7.3 FRP Sheet Stiffness

Maeda et al. (1997) studied the effect of the number of layers of FRP sheets on the bond mechanism using pull-apart tests. They concluded that the ultimate load increases as the FRP sheet stiffness increases. They also stated that the effective bond length decreases as the sheet stiffness increases.

Ueda et al. (1999) concluded in their research on CFRP sheets using different test methods that the greater the stiffness of the sheets the greater the bond strength and the less the strain gradient in the sheets.

De Lorenzis et al. (2000) studied the effect of the number of CFRP plies using bending tests. They concluded that the failure load increases with the increase of the number of plies but the proportion is not linear, where doubling the plies increases the failure load by only 50%.

Nakaba et al. (2001) concluded in their experimental investigation on pull-apart tests that the maximum load increases as the stiffness of the FRP sheet increases. They also stated that the FRP stiffness influences the bond strength and the shape of the stress distributions.

#### 2.7.4 Concrete Strength

Swamy et al. (1986) tested a series of double-lap shear tests (pull-apart tests) with steel adherents bonded to concrete. They tested samples with concrete compressive strengths ranging from 24 to 70 MPa and found that the average bond strength at failure varied almost linearly from 2.15 to 3.3 MPa.

Chajes et al. (1996) conducted a series of direct pull tests with compressive strengths ranging from 24 to 45 MPa. They concluded that if the failure mode is governed by shearing of the concrete below the glue line, the ultimate bond strength is

proportional to the square root of the concrete compressive strength. They also recommended that more tests be done before a final relationship can be proposed.

Horiguchi and Saeki (1997) studied the effect of the concrete quality on the bond behaviour of CFRP sheets using three different bond test methods, namely the shear test (pull-apart test), the bending test, and the tensile test. They concluded that the bond strength increases as the concrete compressive strength increases. The effect was most significant for the tensile test and least for the pull-apart test. They also concluded that the value of the concrete compressive strength (f'<sub>c</sub>) influences the failure mechanism. Failure occurred in the concrete when  $f'_c < 25.3$  MPa and delamination at the adhesive interface (debonding) occurred when  $f'_c$  was high.

Nakaba et al. (2001) and Kobayashi et al. (2001) concluded that the maximum local bond stress increases as the concrete compressive strength increases. Both tests used pull-apart bond tests in their investigations. De Lorenzis et al. (2000) on the other hand, who used beam specimens observed no evidence that  $f'_c$  had an effect on the bond strength. They referred this observation to the failure mode, which was at the concrete-epoxy interface.

### 2.7.5 Concrete Surface Preparation

Chajes et al. (1996) studied the effect of the surface preparation on the average bond strength using direct pull tests. They concluded that concrete surface preparation could influence the bond strength.

Yoshizawa et al. (1996) conducted a study using pull-apart tests on the effect of the type of concrete surface preparation on the bond of CFRP sheets. They concluded that the surface preparation by using water jet doubled the capacity of the specimen as compared to sandblasting.

De Lorenzis et al. (2000) stated that the performance of the bending specimen with roughened surface was much better than that of the specimen with sandblasted 24

surface. They concluded that the surface preparation of the concrete could significantly affect the average bond strength but also stated that further study is required to quantify the influence of the surface preparation on the bond performance.

#### **2.7.6** Type of Test Conducted

Several researchers have indicated the effect of the type of bond test method conducted on the test results. Chajes et al. (1996) indicated the need to conduct additional experimental and analytical research to study the effect of the type of test conducted (direct pull test versus pull-apart test).

Horiguchi and Saeki (1997) studied the effect of the test method conducted on the bond of CFRP sheets. Three different types of tests were conducted; namely, shear tests (pull-apart tests), flexural tests, and direct tensile tests. They concluded that the test method had a significant effect on the bond, where the tensile test produced the highest average bond strength followed by the bending test. The lowest average bond strength was found in the shear tests, as shown in Figure 2-9.

#### 2.7.7 Anchorage Details

Because of premature failure due to debonding of FRP sheets from the concrete surface, it is often very difficult to develop the full strength of the composite material. Thus, many investigations have been conducted to determine required development lengths and the effect of end anchorage on the behaviour of FRP sheets bonded to concrete.

Examination of the various premature failure modes in concrete leads to the conclusion that premature failure usually initiates at the end of the plate in flexural strengthening and in the region of diagonal shear cracks in shear strengthening. The reason for the premature failure in flexure strengthening is the stress concentration created by termination of the plate. This implies that proper detailing of the plate end

and additional plate anchorage techniques should be used. A wide variety of plate termination details and end-anchorage techniques have been studied in the literature.

By contrast the reason for premature failure in the shear strengthening case is the existence of shear cracks and insufficient FRP sheet bond development length.

A number of studies reported using special anchorage details for FRP sheet flexure strengthening, consisting of anchor bolts, U-shaped straps near the cut-off, or staggered cut-offs of multiple-layered plates. In addition to these particular details, there were studies on the variation in the distance between the end support and the edge of the FRP plate. The capacity increase of strengthened beams and the conventionally reinforced control beams showed little sensitivity to either the plate gap near the support or to the use of special anchorage details. About half of the cases with special anchorage details still failed by debonding (Bonacci, 1996).

Sharif et al. (1994) found that anchor bolts successfully eliminated plate separation for FRP but that premature failure was still observed as longitudinal cracks develop in the concrete adjacent to the FRP plate, causing separation of the concrete cover below the longitudinal steel. Roberts and Haji-Kazemi (1989) concluded that the use of anchor bolts should not be expected to reduce significantly the magnitude of the normal forces and increase the strength of the plated beams.

By using glued angle plates or transverse FRP wraps, the plating material can be effectively anchored to the tensile face of the beam. The effectiveness of this technique has been commented on by Arduini et al. (1995) who found that a great deal of ductility could be observed until plate rupture or shear failure of the wrap or angle plates. Erki and Heffernan (1995) found that the failure mode of a CFRP plated beam could be changed from premature plate debonding to tensile failure of the sheet at mid span by adding a CFRP wrap around the sides and bottom of the beam in the anchorage zones.

Ali and Oehlers (1997) studied the effect of bonding side plates on the resistance of plated beams to peeling in the presence of longitudinal cracks. Experimental results indicated that side plates provided additional shear reinforcement and increased the sheet resistance to peeling due to cracking in the order of 100 %.

The use of one-piece FRP I-shaped sheets has been investigated by Sharif et al. (1994). In a comparison of anchor bolts, side plates, and I-shaped sheets as anchoring techniques for FRP sheets, the I-shaped sheets provided superior results. Experimental data indicated that I-shaped sheets could change beam behaviour from brittle to ductile and allow the plated member to achieve the full composite strength.

Because FRP sheets are much thinner than steel plates of equivalent capacity the peeling effect at the end of the sheets is highly reduced and the likelihood of a need for end fixing is reduced (Hollaway and Leeming, 1999). Depending on the sheet/plate location and concrete cross section the choice of anchorage technique will be different. Because research in this area is limited, the only way to be certain that a particular anchorage technique will work in practice is to conduct laboratory tests. This will ensure a safe and effective solution. However, research indicates that FRP side wraps for FRP plating should be used wherever possible.

Sato et al. (1997a) studied several anchoring methods for CFRP sheets used for shear strengthening. Figure 2-15 shows the anchorage methods investigated. They recommended using longer anchoring bolts that penetrate the entire beam width. They also concluded that sheets with mechanical anchorage were more effective compared to those without them. They qualitatively demonstrated the shear strength improvement due to the mechanical anchors but recommended further study on the quantitative assessment of the anchorage efficiency.

Mechanical anchors made of steel, although effective in the laboratory, are not very practical for field application due to drawbacks such as stress concentration and, in the case of bolting, discontinuity of the FRP at drilling locations. In some cases bolting or jacketing cannot be carried out because of possible damage to the reinforcing or prestressing bars or due to insufficient space around the beam. In the case of CFRP, there is also a likelihood of galvanic corrosion because of steel-carbon fibre contact, which is also a concern.

#### **2.8 Bond Strength Models**

In section 2.6 failure modes were presented and discussed. An enormous amount of research effort has focused on determination of the types, causes, and prevention of premature failure modes. Various attempts have been made to characterize the behaviour of these failure mechanisms so that a rational design approach can be formulated.

Different researchers have proposed several bond strength models. The models can be classified into three categories: empirical models based on regression of test data, fracture mechanics based models, and design models that use simplified assumptions. A review of the significant proposed models is presented below.

#### 2.8.1 Empirical Models

Sato et al. (1996) stated that Tanaka (1996) carried out an experimental series on push-apart test specimens and presented a simple expression for the average bond stress

$$\tau_{\rm b} = 6.13 - 0.92 \ln L \ (MPa) \tag{2-2}$$

in which L is the bond length in mm. The ultimate bond capacity of the joint is given by multiplying  $\tau_b$  by the width  $b_p$  and length L of the bond area. This model does not take into consideration the effective bond length.

Alexander and Cheng (1997) studied the effect of the bond length on the bond behaviour using a series of push-apart tests. They proposed a bond stress model in which:

$$\tau_{b} = \begin{cases} 2.3 & L \le 60 \text{ mm} \\ 3.86 - 0.026L & 60 \text{ mm} < L < L_{e} \\ 1.0 & L \ge L_{e} \end{cases}$$
(2-3)

Here  $\tau_b$  is the average bond strength in MPa, L is the bond length in mm, and L<sub>e</sub> is the effective bond length taken as 110 mm.

Hiroyuki and Wu (1997) conducted a set of pull-apart tests on CFRP sheets. Based on the test results, they proposed the following empirical equation that predicts the average bond stress at failure,  $\tau_b$ , in terms of the bond length, L (cm).

$$\tau_{\rm b} = 5.88 {\rm L}^{-0.669} \,\,({\rm MPa}) \tag{2-4}$$

The ultimate load was obtained by multiplying  $\tau_b$  by the bond area  $b_pL$ . Again, this model does not take the effective bond length into consideration.

The model adopted in the CSA S806-02 (2002) and ISIS Canada Design Manual (2001) for predicting the average bond strength at failure,  $\tau_b$ , is that proposed by Maeda et al. (1997). The proposal was based on the experimental results obtained from pull-apart test specimens using CFRP sheets.

$$\tau_{\rm b} = 110.2 \times 10^{-6} E_{\rm p} t_{\rm p} \ (\text{MPa}) \tag{2-5}$$

where  $t_p$  (mm) and  $E_p$  (MPa) are the FRP sheet thickness and modulus of elasticity, respectively. The ultimate load was obtained by multiplying  $\tau_b$  by the effective bond area  $L_e b_p$ . The following equation in terms of  $E_p$  (MPa) and  $t_p$  (mm) give the effective length,  $L_e$  in mm.

$$L_{e} = \frac{25350}{\left(t_{p} \cdot E_{p}\right)^{0.58}}$$
(2-6)

Bizindavyi and Neale (1999) tested a series of direct pull tests on both CFRP and GFRP sheets with 25 mm bond width. They proposed an exponential equation based on curve fitting of the experimental results. The equation took the form:

$$\tau_{\rm b} = \mathsf{A} \cdot \exp(-\psi \cdot \mathsf{L}) \tag{2-7}$$

where  $\tau_b$  and L are given in MPa and mm respectively and A and  $\psi$  are coefficients obtained from curve fitting the test results. The coefficients were shown to be

different for CFRP sheets and GFRP sheets and also different for one ply and two plies of FRP.

Ueda et al. (1999) carried out a series of tests and proposed an equation to predict the maximum local bond stress. The equation was derived based on force equilibrium and the observed distribution of the strain in the FRP sheets. The equation took the form:

$$\tau_{\rm b} = 0.19 t_{\rm p} E_{\rm p} \le 7.3 \text{ MPa}$$
(2-8)

where  $t_p$  and  $E_p$  are the sheet thickness in mm and the modulus of elasticity in MPa respectively. Only the sheet stiffness was taken into consideration with a  $\tau_b$  limit value of 7.3 MPa, which is based on a maximum effective stiffness of 38.4 GPa.mm.

### 2.8.2 Fracture Mechanics Based Models

Chen and Teng (2001) reported that Holzenkämpfer (1994) investigated the bond strength between steel plates and concrete using non-linear fracture mechanics (NLFM). The bond capacity,  $P_u$ , is calculated as:

$$\mathsf{P}_{\mathsf{u}} = \begin{cases} 0.78\mathsf{b}_{\mathsf{p}}\sqrt{2\mathsf{G}_{\mathsf{f}}\mathsf{E}_{\mathsf{p}}\mathsf{t}_{\mathsf{p}}} & \text{if } \mathsf{L} \ge \mathsf{L}_{\mathsf{e}} \\ 0.78\mathsf{b}_{\mathsf{p}}\sqrt{2\mathsf{G}_{\mathsf{f}}\mathsf{E}_{\mathsf{p}}\mathsf{t}_{\mathsf{p}}} \frac{\mathsf{L}}{\mathsf{L}_{\mathsf{e}}} \left(2 - \frac{\mathsf{L}}{\mathsf{L}_{\mathsf{e}}}\right) & \text{if } \mathsf{L} < \mathsf{L}_{\mathsf{e}} \end{cases}$$
(kN) (2-9a)

where the effective length Le and the fracture energy Gf are given by

$$L_{e} = \sqrt{\frac{E_{p}t_{p}}{4f_{ct}}} \quad (mm)$$
(2-9b)

$$G_{f} = c_{f} k_{g}^{2} f_{ct} \quad (N.mm/mm^{2})$$
(2-9c)

$$k_{g} = \sqrt{1.125 \frac{2 - b_{p}/b_{c}}{1 + b_{p}/400}}$$
(2-9d)

Here  $f_{ct}$  is the tensile strength of concrete;  $c_f$  is a constant determined in a linear regression analysis using pull-apart tests and  $k_g$  is a geometrical factor related to the width of the bonded plate  $b_p$  and the width of the concrete member  $b_c$ .

Täljsten (1994) investigated the bond strength between concrete and both steel and CFRP plates using NLFM analysis.

$$\mathsf{P}_{\mathsf{u}} = \sqrt{\frac{2\mathsf{E}_{\mathsf{p}}\mathsf{t}_{\mathsf{p}}\mathsf{G}_{\mathsf{f}}}{1+\alpha_{\mathsf{T}}}}\mathsf{b}_{\mathsf{p}} \tag{2-10a}$$

where

$$\alpha_{\rm T} = \frac{E_{\rm p} t_{\rm p}}{E_{\rm c} t_{\rm c}} \tag{2.10b}$$

in which  $E_p$  and  $E_c$  are the modulus of elasticity in MPa of the FRP and concrete, respectively,  $t_p$  and  $t_c$  are the thickness in mm of the FRP and concrete, respectively, and  $G_f$  is the fracture energy in N·mm/mm<sup>2</sup>. The fracture energy needs to be calculated from the measured values of the load, deformation, and bond shear stress.

Neubauer and Rostásy (1999) conducted a series of pull-apart tests on CFRP sheets bonded to concrete. They concluded that the fracture energy could be calculated using:

$$G_{f} = k_{p}c_{f}f_{ct}$$
(2-11)

in which  $f_{ct}$  is the tensile strength of concrete, and the factor  $c_f$  contains all the secondary effects and was taken as 0.202 mm, which is an average value from 70 specimens. The factor  $k_p > 1.0$  considers the influence of the plate width relative to the width of the concrete member and usually does not exceed 1.3. They also proposed a modified form of Holzenkämpfer's (1994) model, so that it applies to both steel and CFRP. The model takes the form:

$$P_{u} = \begin{cases} 0.64k_{p}b_{p}\sqrt{E_{p}t_{p}f_{ct}} & \text{if } L \ge L_{e} \\ 0.64k_{p}b_{p}\sqrt{E_{p}t_{p}f_{ct}} \frac{L}{L_{e}} \left(2 - \frac{L}{L_{e}}\right) & \text{if } L < L_{e} \end{cases}$$
(2-12a)

$$L_{e} = \sqrt{\frac{E_{p}t_{p}}{2f_{ct}}}$$
(2-12b)

Because of the complexity of the fracture mechanics based models they are not used as design rules. Other researchers proposed simplified models to be used in design as shown in the following subsection.

#### 2.8.3 Design Proposed Models

By assuming a rectangular shear stress distribution across the full bond length, Van Gemert (1980) proposed an expression for the ultimate load of a bonded FRP sheet joint:

$$P_u = 0.5 w L f_{ct}$$
(2-13)

in which w and L are the FRP bond width and length, and  $f_{ct}$  is the concrete tensile strength. The equation implies that a sufficiently long bonded length can carry any given load, which contradicts the well-established fact that any additional bond length beyond the effective bond length cannot increase the ultimate load.

Chaalal et al. (1998) assumed that the bond behaves as a Mohr-Coulomb material. For shear strengthening, they assumed that the maximum shear stress is twice the average stress and the maximum shear stress does not exceed the Mohr-Coulomb strength equation given by Varastehpour and Hamelin (1996). This gives:

$$\tau_{\rm b} = \frac{2.7}{1 + k_{\rm i} \tan 33^{\circ}} \tag{2-14a}$$

where

$$k_{1} = t_{p} \sqrt[4]{\frac{k_{a}}{4E_{p}I_{p}}}$$
(2-14b)

$$k_a = E_a \frac{b_a}{t_a} \tag{2-14c}$$

where  $b_a$ ,  $t_a$ , and  $E_a$  are the width, thickness, and Young's modulus of the adhesive, respectively; and  $I_p$  is the moment of inertia of the FRP plate. This proposal also does not take the effective bond length into consideration.

Khalifa et al. (1998) proposed a modification of the model proposed by Maeda et al. 1997 and included the effect of the concrete strength. The tensile strength is assumed proportional to  $(f'_c)^{2/3}$ , where  $f'_c$  is normalized relative to the concrete strength of 42 MPa in Maeda et al.(1997) tests.

$$\tau_{\rm b} = 110.2 \times 10^{-6} \left(\frac{f_c'}{42}\right)^{2/3} E_{\rm p} t_{\rm p} \tag{2-15}$$

The effective bond length is calculated using Equation (2.5)

Chen and Teng (2001) based on existing fracture mechanics analysis and experimental observations proposed the following bond capacity model:

$$P_{u} = 0.427k_{p}k_{L}\sqrt{f_{c}'}b_{p}L_{e}$$
(2-16a)

$$L_{e} = \sqrt{\frac{E_{p}t_{p}}{\sqrt{f_{c}'}}}$$
(2-16b)

$$k = \begin{cases} 1 & \text{if } L \ge L_e \\ \sin \frac{\pi L}{2L_e} & \text{if } L < L_e \end{cases}$$
(2-16c)

$$k_{p} = \sqrt{\frac{2 - b_{p} / b_{c}}{1 + b_{p} / b_{c}}}$$
(2-16d)

where  $k_{\text{p}}$  and k are the width ratio factor and the length factor respectively.

Chen and Teng (2001) also compared the performance of the empirical and design models with experimental results. They concluded that the models that do not take the effective length into consideration significantly underestimate the bond strength and lead to very large scatter. They also stated that the models by Khalifa et al. (1998) and Neubauer and Rostásy (1997) showed reasonable performance.

### 2.9 Summary

A review of the literature indicates that a considerable research effort has been directed towards the use of FRP sheets for structural strengthening and rehabilitation. The technique of FRP plating is widely recognised within the research community and is gaining acceptance in the construction industry with hundreds of field applications worldwide. However, rational design approaches for the use of FRP materials in construction need to be formulated.

There is limited information on the anchorage of FRPs to concrete or on the most important design factors to ensure a quality bond, such that the full capacity of the composite can be developed. A lack of understanding of the true effect of the parameters controlling the bond behaviour was seen from the discussions presented and from the wide discrepancy in critical values observed.

Reviewing the different conventional bond test methods used in different bond research pointed out the need to develop a rational standard test method for FRP sheets bonded to concrete. Developing a relationship between the results of these models will also help in developing a consistent database.

Although to date a considerable amount of research has been devoted to the bond capacity of FRP sheets and plates to concrete, there is no commonly accepted model predicting the failure load and the effective bond length (Chen and Teng, 2001). The available empirical models are all based on the bond length or the stiffness of the FRP sheet. On the other hand fracture mechanics models that include various variables are not used in design due to their complexity.

The following chapter describes in detail the experimental program conducted in this research in order to investigate the effect of different parameters on the bond behaviour between FRP sheets and concrete.



Figure 2-1 Typical failure stress of unidirectional GFRP with respect to axis orientation (Holloway and Leeming, 1999)



Figure 2-2 Typical stress vs. strain relationship for different types of fibres







Figure 2-4 Load vs. deflection relationship for beams with and without FRP flexural reinforcement (Cheng, 1998)



Figure 2-5 Shear strengthened RC beams (Cheng, 1998)



Figure 2-6 Load vs. deflection relationship for beams with and without FRP shear reinforcement (Cheng, 1998)



(c) Peeling-off failure





Figure 2-8 Direct tension bond test (a) elevation (b) plan view



Figure 2-9 Bond strength-compressive strength (Horiguchi and Saeki, 1997)


Figure 2-10 Bending bond test (a) elevation (b) bottom plan view



Figure 2-11 Direct pull-apart bond test set-up (Chajes et al., 1996)



Figure 2-12 Direct pull-apart bond test specimen free body diagram







Figure 2-14 Push-apart bond test (a) set-up (b) free body diagram



Figure 2-15 Anchoring methods for shear strengthening (Sato et al., 1997a)

# **3 EXPERIMENTAL PROGRAM**

### 3.1 Introduction

The primary objective of the experimental program was to investigate in detail the parameters controlling the bond length and strain distribution for FRP sheets bonded to concrete. The experimental program consists of two test series. Each test series was conducted using a different bond test method. The first series (S-series) used a proposed modified push-apart test and the second series (P-series) used a pull-apart test. A detailed description of each test specimen, the test set-up, instrumentation, and the property of the materials used are presented. The variables studied in each series are also discussed and presented in this chapter.

### **3.2 Modified Push-Apart Test**

A series of five modified push-apart test specimens were carried out in order to assess the new test method and to study the effect of the FRP sheet dimensions on the bond behaviour. The specimen is a modification to the conventional push-apart type specimen presented in Section 2.8.1. All the specimens in the series had the same concrete shape and dimensions. The variables in this series were the FRP sheet bond length and width.

### **3.2.1** Specimen Description

Each specimen was a rectangular concrete block with a rectangular empty core in the centre, as shown in the isometric view in Figure 3-1. The empty core was obtained by securing styrofoam to the form base, as shown in Figure 3-2. The styrofoam had the required core dimension and was removed after casting the concrete. The concrete dimensions were chosen to prevent the crushing of the concrete in compression at the load location and also to minimize the deformation of the concrete arms at high loads. Metal sheets were placed along the width of the specimen arms in their centre to force the crack to form in the centre of the specimen arms. A rigid steel plate was fixed to the inner face of the specimen to form a flat surface for applying the load. The steel reinforcement was detailed in such a way to prevent the failure of the inner corner of the specimens. Figure 3-3 shows the details and dimensions of the specimen, the reinforcement, as well as the loading arrangement.

The arrangement of the CFRP sheets bonded to the arms of the specimen eliminates the compression field in the concrete within the test region, on which the sheets are bonded, thus closely simulating a crack in a beam. One side of the CFRP sheets has an anchor sheet bonded along the width to force the failure to be on the other (tested) side. This was done in order to reduce the amount of instrumentation on the sheets

## 3.2.2 Test Set-up and Instrumentation

Each specimen was laid on a flat smooth steel surface on which it was free to slide. Steel plates 20 mm thick were placed vertically and secured with plaster to the inner face of the specimen as shown in Figure 3-3. A 220 kN capacity hydraulic jack placed in the centre of the hollow core applied the load in stroke control with a rate of 0.05 mm/min. The jack was controlled by an MTS controller system. A 220 kN capacity load cell located between the hydraulic jack and the steel plate measured the applied load. An LVDT inside the hydraulic jack was used to measure the displacement at the centre of the specimen and was also used in regulating the stroke applied by the jack. LVDTs' were also used to measure the displacement at the faces of the specimen. The change in displacement between the centre and the face of the specimen allows an estimate of the deflection of the concrete arms to which the sheets are applied. Figure 3-4 shows the modified push-apart test set-up and instrumentation.

Strain gauges with 5 mm gauge length were applied to the face of the CFRP sheets on the tested side at various locations in order to measure the strain distribution along the length and width of the CFRP sheets. Figure 3-5 shows the location of the strain gauges on the CFRP sheet for each tested specimen. The bond length and width of the CFRP sheets are also shown in the Figure.

## 3.2.3 Variables Studied

Besides implementing the new test method, several variables were investigated in this test series. Specimen S2, S3, and S5 all had the same bond widths of 150 mm but different bond lengths of 50, 100, and 150 mm, respectively, in order to study the bond length effect on the bond behaviour. The lengths were chosen based on previous studies in the literature (Drimoussis and Cheng, 1994, Chajes et al., 1996, Maeda et al., 1997, and Horiguchi and Saeki, 1997), which showed the effective bond length to range between 60 and 110 mm.

Specimens S4 and S5 had the same bond length of 150 mm but different bond widths of 100 and 150 mm, respectively, in order to study the bond width effect on the bond behaviour. Another specimen with a width of 50 mm was prepared for testing but the specimen was damaged in a laboratory accident. The CFRP width values were taken based on strip widths typically used in practical beam repairs (Deniaud and Cheng, 2000). The second series of tests investigated bond widths of 25, 50, 100, and 150 mm as will be discussed later.

Specimen S1, S3, and S4 all had the same CFRP sheet bond area of 15,000 mm<sup>2</sup> but each had a different L/w ratio in order to study the FRP sheet L/w ratio effect on the bond behaviour. Table 3-1 lists the variables for the modified push-apart test specimens.

### 3.3 Pull-Apart Test

A series of 26 pull-apart test specimens were carried out in order to assess the test method, compare the results with the modified push-apart test, and to investigate the anchorage requirements for the FRP sheets. The test specimen is a slightly modified from the specimen used in the pull-apart test method discussed in Section 2.6.3.2. All specimens in the series had the same concrete dimensions.

## **3.3.1** Specimen Description

Each specimen was a concrete prism with two embedded concentric steel bars and metal sheets placed at mid height, to act as crack initiators when the load is applied. An overview of the steel forms, and reinforcement is shown in Figure 3-6 and Figure 3-7(a). Specimens studying the anchor sheet effect had anchor sheets on both sides of the tested sheet. These specimens had spiral reinforcement confining the steel bars, as shown in Figure 3-6(b), which were added in order to reduce the possibility of any bond-slip of the rebars that apply the load to the concrete. The steel bars were 25 mm diameter and each 500 mm long with half their length inside the concrete prism. The bars were secured in place by passing each bar through an opening in the side form and placing it inside a 28.5 mm diameter and 30 mm long pipe bonded to the metal sheet in the centre of the form. The steel bars alignment configuration is illustrated in Figure 3-7(b). Figure 3-8 shows the specimen dimensions and details of the FRP sheets for the different bond configurations.

### **3.3.2** Test Set-up and Instrumentation

Each specimen was placed vertically in an MTS 1000 testing machine. The specimen top and bottom bars were secured in the upper and lower head of the MTS machine, respectively, using hydraulic grips. The test was conducted in stroke control with a rate of 0.05 mm/min. A data acquisition system recorded the strains, the applied load, and the displacement. The load-displacement response was plotted on a computer screen during the test. Figure 3-9 shows the pull-apart test set-up and instrumentation.

The strain gauges were applied to study the strain distribution along the length and width of the sheets and to compare the results with previous results from the modified push-apart tests. Strain gauges with 5 mm gauge length were applied to the face of the CFRP sheets on the tested side of specimens P9 through P13 at various locations, as shown in Figure 3-11.

# 3.3.3 Variables Studied

This test series had several objectives including studying the same variables as the S-series in order to verify the results and compare the test methods. Another objective was to investigate an experimental procedure to estimate the effective bond length. Then finally an investigation into the effect of anchor FRP sheets on the behaviour of FRP sheet bonded to concrete.

Specimens P1 through P8 were used to investigate an experimental procedure to determine the effective bond length. The FRP sheet width was taken as 100 mm in all eight specimens and the sheet bond length was taken equal to  $kL_{ea}$ , where k is the length factor, ranging from 0.6 to 2.0 with increments of 0.2. The anticipated effective length,  $L_{ea}$ , was estimated using an equation proposed by Maeda et al (1997):

$$\tau_{\rm b} = 110.2 \times 10^{-6} E_{\rm p} t_{\rm p} \ (\text{MPa}) \tag{3-1}$$

in which  $t_p$  and  $E_p$  are the FRP thickness in mm and modulus of elasticity in MPa, respectively. The thickness of the laminated CFRP sheet,  $t_f$ , was assumed 0.51 mm, and the modulus of elasticity,  $E_f$ , was assumed as 45,000 MPa in the equation. The value of  $L_{ea}$  based on the previously stated values was calculated to be 75 mm. The number of specimens was chosen in order to have enough results to accurately identify the effective bond length. The bond length values,  $kL_{ea}$ , were chosen in order to capture the behaviour within a sufficient range.

Specimens P9, P10, and P11 with the same bond width of 150 mm and with bond lengths of 50, 100, and 150 mm, respectively, were used to study the bond length effect. Specimens P12, P13, P8, and P11 had the same bond length of 150 mm and bond widths 25, 50, 100, and 150 mm, respectively, and were used to study the sheet width effect on the bond behaviour. The pull-apart specimens P9 through P13 had the same FRP sheet bond length and width as the specimens in the S-series. The effect of the test method used was investigated by comparing the results of specimens P9 through P13 with the results from the modified push-apart specimens S1 through S5 as will be shown in Chapter 4.

Although the use of externally bonded FRP composites has been implemented in many strengthening projects, there has always been a concern for the end anchorage. This is particularly important when the length of the FRP system is restricted and the bonded length beyond a critical section is not sufficient to achieve the ultimate strength of the FRP reinforcement. There has been research on the use of different anchorage methods for both flexure and shear strengthening as presented in Section 2.7.7. The literature shows the importance of the external anchorage on preventing plate peeling. Externally bonded anchor sheets are preferred over mechanical anchorages that add significantly to both project cost and complexity (Spadea et al., 1998) in addition to the drawbacks discussed in Section 2.7.7 of using anchors made of steel.

The effect of bonded FRP anchorage sheets was investigated using specimens P14 through P26, which studied some of the CFRP sheet anchor requirements. The effect of the distance of the anchor from the crack, a, was studied using specimens P14 through P17, which had their anchors above the tested sheet, and P21 through P24, which had their anchors below the tested sheet. The effect of the anchor extension beyond the tested sheet, b, was studied using specimens P17 through P20, which had anchors above the test sheet, and P24 through P26, which had the anchors below the test sheet. The sheets and anchors configurations studied are illustrated in Figure 3-8. Four specimens were used to verify the results of certain specimens in the series. The variables for each specimen are also detailed in Table 3-4.

Figure 3-10 shows the practical simulation of the distances "a" and "b" illustrated on a concrete girder strengthened in flexure and shear using FRP sheets. The distance "a" represents the distance from the anchor sheet to the crack and "b" represents half the spacing between the anchor sheets.

## **3.4** Material Properties

### 3.4.1 Concrete

Gravel with a nominal maximum size of 10 mm was used for coarse aggregate. The fine aggregate used was coarse sand. Both the gravel and the sand were dry, of uniform characteristics, and free from injurious materials. Type 50 cement was used in the concrete mix. Clean tap drinking water free from impurities was used in the concrete mix and in the curing of the specimens.

Most research done in the literature used concrete with compressive strengths in the range of 45 MPa (Alexander and Cheng, 1997, Maeda et al., 1997, Brosens and Van Gemert, 1997, Udea et al., 1999, and Bizindavy and Neale, 1999). Because a comparison with the research results from the literature was planned, the concrete mix was designed to have a compressive strength of 45 MPa and is shown in Table 3-5 and Table 3-6 from the push-apart tests and the pull-apart tests, respectively. The concrete was mixed using a 0.3 m<sup>3</sup> capacity concrete mixer. Three compression cylinders and two split cylinders were taken from each batch of concrete. The specimens along with the cylinders were cured with wet cloth for two days and then left in normal room temperature for 28 days. The average concrete compressive strength was 50.2 MPa and 47.1 MPa for the modified push-apart tests and pull-apart tests, respectively. The tensile strength was taken as 0.65 of the strength resulting from the split cylinder tests (Collins and Mitchell, 1987), which was calculated using the following equation:

$$f_{sp} = \frac{2P}{\pi Ld}$$
(3-2)

in which P is the crushing load, L is the length of the cylinder, and d is the diameter of the cylinder. The average tensile strength of the concrete was 2.6 MPa and 2.3 MPa for the modified push-apart tests and pull-apart tests, respectively. The results of the concrete compression tests and the split cylinder tests conducted in accordance with CSA/CAN Standard A23.2 (1994) are shown in Table 3-2 and Table 3-3 for the modified push-apart tests and pull-apart tests, respectively.

### 3.4.2 CFRP Sheets

The CFRP sheets used in these tests were Replark type 20 single layered unidirectional tapes supplied by Mitsubishi Canada Ltd. Coupon tests were carried out in accordance with ASTM Standard D 3039/D 3039M (1995) to determine the material properties of the CFRP sheets. The average result from five 300 mm long and 25 mm

wide specimens was used. The applied load and the strains in the sheet were monitored and recorded for the duration of the tests. Strain gauges were applied in the direction of the fibres and in the direction perpendicular to the fibres in order to plot the stress-strain curve for the sheet and hence determine the modulus of Elasticity of the sheet and also to measure the lateral Poisson's ratio. Although the sheets are unidirectional, ASTM procedure advises that Poisson's ratio be measured. The value of the modulus of elasticity and Poisson's ratio will be used in the numerical model that will be discussed in detail in Chapter 5. The ultimate stress, Poison's ratio, and modulus of elasticity of the manufactures specifications and the coupon tests are shown Table 3-7. The thickness of the tested CFRP sheet was 0.38 mm for both test series.

### 3.4.3 Epoxy

The primer and resin used were both a two-component, high-modulus, and highstrength epoxies in which the two components are mixed at a 2:1 ratio by weight. The putty used was a two-component epoxy mixed at a 1:1 ratio by weight. Mitsubishi Canada Ltd supplied the primer, resin, and putty. The modulus of Elasticity (E), Poisson's ratio ( $\nu$ ), and the ultimate tensile strength (f<sub>ct</sub>) of the materials obtained from the manufactures specifications, is listed in Table 3-7.

### 3.4.4 Steel

High-grade 10M rebars (Figure 3-3) were used in the modified push-apart tests to prevent any corner failure of the concrete specimen. 4 mm diameter plain steel bar stirrups were also used to separate the top and bottom 10M reinforcement. 25M deformed steel bars were used for the rebar in the pull-apart specimens. The stress in the bars was predicted to be much lower than the typical yield strength of these bars. The expected extension of the rebars under load was also negligible.

## **3.5** Surface Preparation and Bonding the FRP Sheets

The sheets were bonded to the concrete according to the manufacturer's recommendations and procedures. After grinding the concrete surface with a stone

wheel, the surface was cleaned and an epoxy primer applied to it in order to create a smooth flat bond surface. The primer was applied to concrete with a roller in the amount of  $0.25 \text{ kg/m}^2$ . The same procedure was used for both test series.

In the P-series it was noticed that there were small voids in the surface. A layer of putty was applied after 24 hours, when the primer layer was no longer sticky, to fill the voids in the surface.

The resin was applied after 24 hours with a roller in the amount of  $0.3 \text{ kg/m}^2$  after which the CFRP was bonded. The sheets were bonded in position within 10 minutes of applying the resin coating using a roller. Rolling continued until sheets were fully impregnated with resin. A plastic roller was used to smooth out the CFRP sheet surface. A resin topcoat in the amount of  $0.2 \text{ kg/m}^2$  was then applied after 30 minutes to the entire sheet. The specimens were left for 24 hours to dry after which the specimens were left to cure in room temperature for a week until tested.

L w		L/w	f"c
(mm)	(mm)		(MPa)
75	200	0.38	51.1
50	150	0.33	50.4
100	150	0.67	51.3
150	100	1.50	50.4
150	150	1.00	51.1
	L (mm) 75 50 100 150 150	L         w           (mm)         (mm)           75         200           50         150           100         150           150         100           150         150	LwL/w(mm)(mm)752000.38501500.331001500.671501001.501501501.00

 Table 3-1 Modified push-apart test variables

Table 3-2 Modified push-apart concrete compression and split cylinder test results

Specimen	f'c (MPa)	d (mm)	L (mm)	P (kN)	( <b>f</b> <sub>sp</sub> ) <sub>av</sub> (MPa)
S1, S3	51.3	152.5	305.0	326.0	4.1
<b>S</b> 5	51.1	152.7	305.0	310.0	4.3
S2, S4	50.4	152.4	305.0	284.0	4.1

Specimen	(f'c) <sub>av</sub> (MPa)	<b>d</b> (mm)	L (mm)	P (kN)	f <sub>sp</sub> (MPa)	( <b>f</b> <sub>sp</sub> ) <sub>av</sub> (MPa)
P1- P10	46.9	151.8 152.1	305 304	247.3 225.2	3.4 3.1	3.25
P11- P20	47.3	151.7 152.4	305 305	283.4 262.8	3.9 3.6	3.75
P21- P30	46.4	151.3 151.9	305 304.5	275.4 239.8	3.8 3.3	3.55

Table 3-3 Pull-apart test concrete compression and tensile split cylinder test results

**Table 3-4 Pull-apart test variables** 

Specimen	L	W	a	b	f'c	Demontra
	(mm)	(mm)	(mm)	(mm)	(MPa)	Kemarks
P1	45	100	* * *		46.9	
P2	60	100			46.9	
P3	75	100	25	50	46.9	
P4	90	100			46.9	Estimating the
P5	105	100			46.9	effective bond length
<b>P6</b>	120	100			46.9	(No anchor sheets)
<b>P7</b>	135	100			46.9	
P8	150	100		400 - 000 - 000	46.9	
P9	50	150	25	25	46.9	Length effect
P10	100	150	25	25	46.9	(Anchor sheet at one end)
P11	150	150	25	25	47.3	
P12	150	25	25	75	47.3	Width effect, including P11
P13	150	50	25	75	47.3	(Anchor sheet at one end)
P14	100	50	0	75	47.3	- -
P15	100	50	25	75	47.3	
<b>P16</b>	100	50	50	75	47.3	Anchor sheets at both ends
<b>P17</b>	100	50	75	75	47.3	and above the tested sheet
P18	100	50	25	50	47.3	
P19	100	50	25	25	47.3	
P20	100	50	*		47.3	
P21	100	50	0	75	46.4	
P22	100	50	25	75	46.4	
P23	100	50	50	75	46.4	Anchor sheets at both ends
<b>P24</b>	100	50	75	75	46.4	and below the tested sheet
P25	100	50	25	50	46.4	
P26	100	50	25	25	46.4	

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

water/cement	Water	Cement	Sand	Gravel
	(kg)	(kg)	(kg)	(kg)
0.57	200	350	700	1055

Table 3-5 Push-apart test concrete mix for 1.0 m<sup>3</sup>

Table 3-6 Pull-apart test concrete mix for 1.0 m<sup>3</sup>

water/cement	Water	Cement	Sand	Gravel
	(kg)	(kg)	(kg)	(kg)
0.54	240	442	921	790

# **Table 3-7 Material properties**

	E (GPa)	ν	<b>f</b> <sub>c</sub> , (MPa)	Remarks
Concrete	32.1**	0.15	2.6	Push-apart test
	30.8**	0.15	2.3	Pull-apart test
CFRP	230	0.26	3400	Manufactures specifications*
	45	0.26	422	Coupon tests
Adhesive	2.8	0.35	42	Manufactures specifications*

\* Based on fibre strength\*\* Based on 45% secant modulus

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.



Figure 3-1 Orthographic of the modified push-apart test



Figure 3-2 Modified push-apart test steel form and cage details







Figure 3-4 Modified push-apart test set-up and instrumentation



Figure 3-5 S-series strain gauge locations on CFRP sheets and concrete



Figure 3-6 Pull-apart specimen forms and reinforcement (a) without spiral (b) with spiral







Figure 3-8 Pull-apart specimen dimensions, details, and sheet and anchor configurations



Figure 3-9 Pull-apart test set-up and instrumentation



Figure 3-10 Concrete girder strengthened in flexure and shear with FRP sheets





### **4** TEST RESULTS, OBSERVATIONS, AND DISCUSSION

### 4.1 Introduction and Overview

Two series of concrete specimens bonded with CFRP sheets were tested in the University of Alberta structural laboratory. The first series, S-series, employed a modified push-apart procedure and the second series, P-series, employed a pull-apart procedure. Specimen description, instrumentation, variables and the material properties for each test series were discussed in detail in Chapter 3. This chapter presents and discusses the test results and the observed behaviour of the specimens in both series.

Besides implementing the modified push-apart test method, specimens S2, S3, and S5 in the S-series investigated the CFRP bond length effect (L), specimens S4 and S5 investigated the bond width effect (w), and finally specimens S1, S3, and S4 investigated the length to width, L/w, ratio effect on the bond behaviour.

The P-series, which employed a pull-apart procedure, consisted of 26 specimens. Specimens P1 through P8 in the P-series implemented a procedure to determine the effective bond length. Specimens P9 through P11 studied the bond length effect while specimens P8, P11, P12, and P13 investigated the bond width effect on the bond behaviour. Specimens P9 through P13 had the same variables as specimens S1 through S5 in order to study the effect of the test method on the results.

The effect of anchor sheets on the bond behaviour included specimens P14 through P26. The specimens had different anchor configurations designed to study the anchor location when having the anchor sheet bonded above or below the tested sheet. The effect of the distance of the anchor sheet from the crack "a" and the effect of the anchor sheet extension beyond the tested sheet "b" on the bond behaviour were also studied.

# 4.2 Failure Mechanism

### 4.2.1 Modified Push-Apart Test Failure Mechanism

As described in detail in Chapter 3, the load was applied to all specimens in the Sseries with a 220 kN capacity hydraulic jack. The load was applied monotonically in stroke control until failure and the load together with the displacements and strains were read directly from a data acquisition system.

Specimen S2 had a unidirectional CFRP sheet bond length of 50 mm on each side of the crack. An anchor sheet was applied to one side of the tested sheet. The sheet in S2 was mistakenly cut in the wrong direction causing the fibres to be in the direction of the crack instead of perpendicular to the crack. This caused the sheet to fail as soon as the concrete cracked. The expression "joint" used here refers to the assembly of concrete, adhesive, and FRP sheet. The specimen was then clamped and CFRP sheets bonded above the initial fractured sheets. The specimen in this case became a precracked specimen and the CFRP sheet became bi-directional.

All specimens failed by debonding type failure, with a portion of the concrete shearing near the crack, and forming a wedge of concrete bonded to the FRP sheet. The metal sheet placed in the specimen arm to initiate the crack was at a distance averaging 18 mm from the bond face of the concrete. This caused a stress concentration in the concrete at this neck area, which caused the formation of the concrete wedge. Bizindavyi and Neale (1999) reported a similar three-dimensional mode of failure in the direct pull concrete block occurred at the loaded end. The crack propagation and the formation of the failure surface are shown in Figure 4-1. As the load increases, micro cracks appear, which then propagate to form shear cracks. The failure surface close to the crack was inclined at an angle  $\phi$  in all the specimens, as shown in Figure 4-2. The angle  $\phi$  ranged between 34° and 41.6°, and averaged 38° as shown in Table 4-1. This initial failure zone formed close to the failure load and this area of concrete stayed bonded to the CFRP sheet after failure occurred, as shown in Figure 4-3b. As a specimen reaches the failure load, the failure zone propagates parallel to the concrete surface rapidly with no increase in

load, and the amount of concrete bonded to the remaining portion of the sheet is an irregular thin layer as shown in Figure 4-3b.

Table 4-1 lists the initial failure zone dimensions for all specimens. The initial failure length is almost the same in all specimens ( $c \approx 38 \text{ mm}$ ) except for specimen S1, which had a CFRP sheet covering the full specimen width (c = 61 mm). This variance in behaviour points out the need to study the effect of the FRP sheet width relative to the specimen width on the failure mechanism and the behaviour of the joint in general. The average bond strength  $\tau_b$  listed in Table 4-1 was calculated by

$$\tau_{\rm b} = \frac{{\sf P}_{\rm u}}{2{\sf L}\cdot{\sf W}} \tag{4.1}$$

in which,  $P_u$  is the measured ultimate load in the load cell, and L and w are the CFRP sheet length and width, respectively, in the test zone.

### 4.2.2 Pull-Apart Test Failure Mechanism

As described in detail in Chapter 3, the specimens of the P-series were placed in a MTS universal-testing machine in a vertical position. The load was applied monotonically in stroke control until failure and the load together with the displacements and strains were read directly from a data acquisition system. Specimens P1 through P13 are a subset of specimens with variable bond length and width. Specimens P14 through P26, which had anchor sheets on both sides of the tested sheet but constant bond length and width, are considered a different subset.

The results from specimens P6 and P20 were inconsistent with the results from all other specimens in the series. The reason for the relatively high load capacity in these specimens was unknown but could be due to better concrete surface condition that limited the use of putty. These specimens were, therefore, discarded from the analysis.

Failure started in all the specimens in the first subset, P1-P13, with the concrete cracking at the metal sheet location (the crack initiator location), then shear cracks

appeared and gradually increased in length until the CFRP sheet debonded. Debonding started at the crack location and progressed towards the free end of the sheet. Ultimate load occurred when the sheet was totally debonded in a sudden zip type failure. The crack in the centre of all the specimens opened at one face only until failure, as shown in Figure 4-4. The face that had the crack opening was eventually the face that had the bond failure.

The amount of concrete attached to the sheet at the crack location after failure ranged from a very thin layer to a chunk 20 mm wide and inclined  $42^{\circ}$  at the crack location. A typical post failure mode for specimens P1 through P13, except for specimen P3, is shown in Figure 4-5. The distance of the metal sheet from the concrete surface to which the CFRP sheets were bonded (d<sub>1</sub>) averaged 18.8 mm in the S-series and was nil in the P-series. The distance d<sub>1</sub> affected the amount of concrete attached to the CFRP sheet after failure. The S-series type failure simulates a failure at which the crack is developed after bonding the CFRP sheets while the P-series simulates a failure at which the CFRP sheets are bonded to a pre-cracked concrete surface.

## 4.2.3 Anchor Sheets

Specimen P3 had an anchor sheet placed 90° to the load direction at one end of the tested sheet, partially covering the test sheet and partly bonded to the concrete beyond the test sheet end. The specimen failed at the anchored side of the sheet, which was 70 mm in length compared to an unanchored side of 80 mm. As the load increased, the anchor was torn at the location where the tested sheet ends, as shown in Figure 4-6. This caused the effective width of the anchor to be 50 mm. This failure mode showed that an extra 10 mm in length was stronger than having a 50 mm wide anchor perpendicular to the sheet. Further investigation to verify this observation is required.

The second subset specimens P14 through P26 that were intended to study the anchor sheet effect on the bond behaviour exhibited uniformly the same failure mechanism. Failure started with the concrete cracking at the metal sheet location. Shear cracks then appeared and gradually increased and widened. Debonding started in

the tested sheet at the crack location then propagated towards the end of the sheet. Anchor debonding started over the area in contact with the tested sheet and occurred simultaneously as the tested sheet debonded. As the centre crack widened, the anchor appeared to stretch towards the crack location, as shown in Figure 4-7. The debonding in the anchor then propagated towards one of its edges at about 20% of the ultimate load.

The location of the anchor sheet being above or below the tested sheet did not show any change in the failure mode. For example Figure 4-8 shows the failure modes of specimen P15 with anchor sheet above the tested sheet and P22 with anchor sheet below the tested sheet. The anchor sheet in specimens P17 and P24 was torn at the location where the tested sheet ends, as shown in Figure 4-9. In this case the tested sheet debonded from the concrete. The effective width of the anchor corresponded to the part of the anchor directly above or below the sheet for specimens P14 through P19 and P21 through P26, respectively.

It is concluded from the previous discussion that the width of the perpendicular anchor sheet that is not directly above or below the tested sheet is ineffective. It is also observed that placing the anchor sheet above or below the test sheet makes no difference to the failure mechanism.

### 4.3 Load Transfer Process

Strain gauges were applied to the CFRP sheet surface along the length and across the width of the sheet in the modified push-apart tests and five of the pull-apart tests, as detailed in section 3.2.2 and section 3.3.2, respectively. The strain gauges were applied in order to study the load transfer process along the bond length and across the bond width of the CFRP sheet, which would provide a better understanding of the bond behaviour between FRP sheets and concrete.

### **4.3.1** Load Transfer Along the CFRP Sheet Length

Figure 4-10 through Figure 4-14 show the average strain distributions at the centre of the CFRP sheets along the length at different load levels for specimens S1 through S5,

respectively. The figures show that as the load increase the strain values increase across an increasing portion of the bond length, indicating that more bond area resists the load as it increases.

The strain distributions along the length of specimens P9 through P13 are shown in Figure 4-15 through Figure 4-19, respectively. The strain distribution along the length shows a change in trend as the load increases. The strain distribution is concave at lower loads then transforms into a convex shape as the load reaches 70% of the ultimate load. All specimens of the P-series showed this trend except for specimen P9, which did not have enough strain gauges to show it. The same observation was made in the S-series. The change in the strain distribution trend with a large increase in the strain values is similar to a snap-through behaviour and indicates debonding of the CFRP sheet. This observation reflects on the comparison between the strain distributions of different specimens, where the strain distribution shape and hence values at various locations will depend on the load level at which the comparative strain distributions are at.

The term transfer length is defined as the distance from the crack location to the point where the strain is negligibly small. Because of the limited number of strain readings along the CFRP sheet length, the transfer lengths were reasonably estimated from Figure 4-10 through Figure 4-19. Figure 4-20 shows the load vs. the estimated transfer length for specimens S2, S3, and S5 with the same CFRP sheet bond width of 150 mm. The figure shows that the transfer length increases as the load increases. The increase in the transfer length at the lower load levels was almost constant but the rate then increased rapidly at the higher loads for specimens with bond lengths above the effective length, as seen by the steeper slopes for S3 and S5 in Figure 4-20. The same behaviour occurred in the P-series as shown in Figure 4-21, which shows the load vs. the transfer length for specimens P9, P10, and P11 having the same bond width of 150 mm. This behaviour, the transfer length rate increasing as the load increases, indicates an increase in the CFRP sheet debonding rate, which leads to the zip action type failure.

Bizindavyi and Neale (1999) stated that there exists an initial transfer length that is constant for all load levels lower than the initial cracking load. They stated that the initial cracking load ranged from 55 % to 68% of the ultimate tensile capacity of the FRP sheets depending on the number of plies. This statement could not be verified either from their published test results or the test results published in this research considering the limited number of strain readings along the CFRP sheet length.

In order to investigate the load transfer mechanism along the length in detail, the average bond strength in the centre of the CFRP sheet along the bond length for specimen S5 is shown in Figure 4-22 as a function of the relative load level, which is the current load divided by the ultimate load ( $F/F_{max}$ ). The average bond stress ( $\tau_b$ ) was calculated by

$$\tau_{\rm b} = \frac{\mathrm{E}_{\rm p} \mathrm{t}_{\rm p} (\mathrm{g}_{\rm i} - \mathrm{g}_{\rm j})}{\Delta \mathrm{L}_{\mathrm{i}-\mathrm{i}}} \tag{4.2}$$

in which  $g_i$  and  $g_j$  are the strain values at points i and j, respectively,  $\Delta L_{i-j}$  is the distance between i and j, and  $E_p$  and  $t_p$  are the modulus of elasticity and thickness of the CFRP sheet, respectively. A schematic sketch and derivation of equation (4.2) are shown in Figure 4-23. The distances between the strain gauges for S5, shown in Figure 3-5, were divided into segments. The segment numbering and location along the CFRP sheet length are indicated in the legend of Figure 4-22. Line (1) in the figure identifies the location where  $\tau_b$  in segment 2 increases as  $\tau_b$  in segment 1 decreases. Line (2) and (3) also show the same behaviour for segments 3 and 2, and 4 and 3, respectively. The decrease in  $\tau_b$  in a region corresponds to debonding taking place and the build-up of  $\tau_b$ in the adjacent region indicates that the load is transferring to that region. The same behaviour was observed for the P-series as shown in Figure 4-24. This figure shows the average bond strength distribution plotted against the relative load level for segments along the CFRP sheet bond length of specimen P10.

The negative strain gradient that appears within 30 mm from the crack at 65% of the ultimate load (see Figure 4-22) is attributed to the concrete active crack zone shifting relative to the strain gauge location as the load increases. Similar behaviour was

observed in all S-series specimens. P-series specimens did not show such behaviour because there was no flexural shear crack zone in these specimens.

From the previous discussion it is concluded that the load is transferred from the crack location towards the free end of the CFRP sheet and that the bonded area of the sheet resisting the load increases as the load increases. Debonding takes place as the bond strength is exceeded in the bonded area resisting the load. The bonded area then shifts away from the crack and moves towards the free end of the sheet, which marks the debonding propagation. The rate of debonding increases as the ultimate load is approached. This behaviour causes the zip action that is observed in these kinds of tests, but more importantly it indicates that an "effective bond length" that does not necessarily start at the crack is perhaps best used to estimate the bond strength.

### 4.3.2 Load Transfer Across the CFRP Sheet Width

Figure 4-25 through Figure 4-29 show the strain distribution across the CFRP sheet width at the crack location for specimens S1 through S5, respectively. The figures show that the strain distribution along the sheet width did not have a defined trend as a whole, where the strain values at the edge were always higher compared to those at the centre of the sheet while the strain values half way between the centre and the edge of the sheet fluctuated relative to that at the centre. At the crack location, the increase in the average strain value at the edge of the sheet over that at the centre of the sheet at ultimate load ranged from 9.3% in S1 to 134% in S2 and averaged 58% for all five specimens. The total average value of the difference between the edge and centre strain was 39%, excluding S2, which had a CFRP sheet with fibres parallel to the crack bonded below the tested sheet.

Figure 4-26 shows the strain distributions at different load levels along half the width of the CFRP sheet in specimen S2 at distances 0, 15, 30, and 45 mm from the crack location. The figure shows that the edge strain was consistently higher than the centre strain throughout the sheet length and at all load levels.

The strain distributions along the width of specimens P9 through P13 are shown in Figure 4-30 through Figure 4-34, respectively. As in the S-series, the strain distribution along the width of the sheet did not have a defined trend as a whole but the strain value at the edge of the sheet was consistently larger than that in the centre of the sheet for all specimens. At the crack location, the increase in the average strain value at the edge of the sheet compared to that at the centre of the sheet at the ultimate load ranged from 8% in P13 to 48.6% in P11 and averaged 31% for all five specimens.

It is not simple to back calculate the stress distribution from the strain values for a number of reasons. The shear lag effect complicates the picture and may interfere with accuracy of the strain gauges located near the edges of the sheet. Debonding may not occur evenly across the sheet width. But the trend is nonetheless true as will be shown later using numerical analysis.

Figure 4-35 shows the CFRP sheet width plotted against the strain difference between the edge and the centre of the sheet at the crack location at different load levels. The figure was developed using the results from specimens S4, S5, P11, P12, and P13 that had the same bond length of 150 mm. Least squares prediction lines were developed using the built in option in MS EXCEL (1999) program for load levels of 50%, 70%, and 100% of the ultimate load. The least squares prediction lines show that the difference in strain between the edge and centre of the CFRP sheet increases as the width of the sheet increases. This behaviour is further studied with the 3D numerical model detailed in Chapter 6.

Figure 4-36 shows the strain distributions at both the sheet centre and at the edge along the length at various load levels in specimen S4. The shaded areas in the figure show that the difference between the strain values at the edge of the sheet and its centre increases as the load increases. The difference is also greater near the crack than away from the crack location.

Figure 4-37 and Figure 4-38 show the average bond stress ( $\tau_b$ ) distributions across the CFRP sheet width at the crack location for specimens S1 and S4, respectively.

Specimens S1 and S4 had different FRP sheet lengths and widths and were chosen as examples to illustrate a behaviour. The charts legends show the distance from the CFRP sheet centre. Equation (4-2) was used to calculate the values of  $\tau_b$  in the charts. The figures show that  $\tau_b$  distribution in the CFRP sheets develops after the concrete specimen cracks. This occurred at around 20% of the ultimate load for all specimens except for S2, which was pre-cracked. The  $\tau_b$  distribution in specimen S2 started as soon as the load was applied as shown in Figure 4-39. Closer to the crack, the magnitude of  $\tau_b$  at the edge of the sheet is higher than that at the centre in all specimens. The  $\tau_b$  distributions show several trend changes as indicated by the vertical dash lines, where the sudden change in the trend indicates debonding.

Figure 4-40 and Figure 4-41 show the average bond stress ( $\tau_b$ ) distributions across the CFRP sheet width at the crack location for specimens P10 and P11, respectively. Specimens P10 and P11 had different bond widths and were chosen as an example to illustrate a behaviour. Equation (4-2) was also used to calculate the values of  $\tau_b$  in those charts. The figures show that  $\tau_b$  distribution in the CFRP sheets develops as soon as the load is applied, which is similar to the behaviour of specimen S2 that was precracked. The crack initiator metal sheet placed in the P-series specimens causes the concrete specimen to perform as pre-cracked. The figures also show that there are trend changes in the  $\tau_b$  diagram at the locations shown by vertical dashed lines. The sudden drop in the  $\tau_b$  diagram indicates debonding of the sheet, where the test was in stroke control.

From the previous discussion it is concluded that the strain values at the edge of the CFRP sheet were higher than those at the centre throughout the sheet length and at all load levels. The difference in strain increases as the load increases and as the distance from the crack decreases. The difference between the strain values at the edge and those at the centre of the sheet was 35% on average at ultimate load at the crack location. The difference also appeared to increase as the sheet width increases.

# 4.4 Bond Length

## 4.4.1 General Discussion

The modified push-apart specimens S2, S3, and S5 had the same bond width of 150 mm, but had bond lengths of 50, 100, 150 mm, respectively. Figure 4-42 shows the load response plotted against the deflection for these specimens. It was observed from the curves that the specimens had the same trend except for specimen S2, which was pre-cracked and showed a slightly softer behaviour. The ultimate capacity also increased with increasing the bond length but to a certain limit. This is apparent from specimens S3 and S5 with bond lengths of 100 mm and 150 mm, respectively, where there was no significant change in the ultimate capacity with the increase in bond length. It is concluded that the bonded length beyond a certain distance from the crack causes no increase in the ultimate capacity of the joint. This distance is referred to as the effective bond length. This conclusion is similar to the findings of other researchers in the literature (Drimoussis and Cheng, 1994, Chajes et al., 1996, Alexander and Cheng, 1997, Brosens and Van Germert, 1997, Maeda et al., 1997, Horiguchi and Saeki, 1997, Täljsten, 1997).

For a given configuration (sheet width, stiffness, surface conditions, etc.), ultimate strength develops over this effective length, and then debonding propagates through the rest of the sheet without a limit. The value of the effective length depends on many variables of which the most significant, according to the literature, are the sheet stiffness and the concrete tensile strength (Holzenkämpfer, 1994, Neubauer and Rostásy, 1997, Chen and Teng, 2001). It appears from Figure 4-42 that the value of the effective bond length in the S-series with w = 150 mm is between 50 mm and 100 mm. Taking the effective bond length of 100 mm for specimens S4 and S5 resulted in  $\tau_b$ equal to 2.13 MPa and 1.79 MPa, respectively. The difference in  $\tau_b$  between S3 and S5 having bond lengths of 100 and 150 mm, respectively was 5% and the overall average  $\tau_b$  value was 2.06 MPa. The overall average  $\tau_b$  for conventional push-apart tests studied by Drimoussis and Cheng (1994), and Alexander and Cheng (1997) were 2.6 and 2.3 MPa, respectively. The difference in the  $\tau_b$  values is due to the different concrete strengths, CFRP stiffness, and test specimens configurations used.

Figure 4-43 shows the load-displacement responses for the P-series specimens P9, P10, and P11 with bond lengths of 50, 100, and 150 mm, respectively. The specimens had the same CFRP sheet bond width of 150 mm. The curves show the same trend but with P10 and P11 having higher ultimate capacity than P9. There is no significant change in the ultimate load when increasing the bond length from 100 mm to 150 mm, which is similar to the findings from the S-series.

Figure 4-44 shows the strain distribution in the centre of the CFRP sheet along the bond length for specimens S2, S3, and S5 at 20 kN. There was no significant difference between the strain distribution of specimens S3 and S5. This indicates that the strain distribution and magnitude does not change on increasing the bond length beyond the effective bond length. However, the strain values at the crack location were higher for specimen S2 with a smaller bond length. This indicates that specimens S3 and S5 with larger bond lengths had better stress distributions.

Specimens P9, P10, and P11 had the same variables as those in S2, S3, and S5. Figure 4-45 shows the strain distribution in the centre of the CFRP sheet along the bond length at 20 kN for these specimens. The figure shows that the P-series had the same behaviour as the S-series but with different strain values.

### 4.4.2 Effective Length Evaluation

Conclusions from the literature and the current research showed the importance of evaluating the effective bond length in order to assess the average bond strength that is used in strengthening and design calculations. An experimental procedure, explained in section 3.3.3, was proposed to estimate the effective bond length for FRP sheets bonded to concrete. The procedure is also adopted in Annex J in the CSA-S806-02 (2002). Specimens P1 through P8 investigated the validity of the proposed procedure.

The load response plotted against the deflection for specimens P1 through P8, with the exception of P6 as discussed in section 4.2.2, are shown in Figure 4-46. The curves show the same trend but with different ultimate loads and displacement values. Specimen P7 showed a slightly softer behaviour and a reduced ultimate load due to the

existence of air voids under the CFRP sheets that created discontinuity in the glue joint and weakened the bond strength by reducing the area of adhesive. Figure 4-46 shows that the maximum displacement increases as the bond length increases, which maybe interpreted as an increase in ductility of the joint as the bond length increases.

The ultimate loads achieved are plotted against the bond lengths for specimens P1 through P8 in Figure 4-47. A moving average trend line was added using a built-in option in program MS EXCEL (2000) and a linear approximation was plotted manually. The linearized moving average trend line is bilinear with an inclined segment and a horizontal segment. The intersection of the segments indicates the location of the effective bond length, which is estimated to be 80 mm. This value is close to the 75 mm estimated value used in the selection of the tested bond lengths (Equation 3-1).

Based on the previous discussion, refinements to the proposed procedure to determine the value of the FRP sheet effective bond length are recommended. Due to the relatively high variability in bond related test results, taking the average of two tests for each bond length is proposed. Together with the expected ultimate load trend of previous and following bond lengths, any exceptional test result can be neglected and therefore safeguard for unexpected results. The FRP sheet bond lengths should be taken as  $kL_{ea}$ , where  $L_{ea}$  is the estimated bond length value obtained from equation (8-11), as discussed in Chapter 8. The length factor, k, is proposed to range from 0.6 to 1.6 with increments of 0.2 in order to optimize the number of specimens tested. Twelve specimens would be tested with two specimens for each variable in order to evaluate the effective bond length.

#### 4.5 Bond Width

The load vs. displacement responses for specimens S4 and S5 with bond widths of 100 and 150 mm, respectively and with equal bond lengths of 150 mm are shown in Figure 4-48. The figure shows that increasing the bond width in specimen S5 by 50% over that of specimen S4 (from 100 mm to 150 mm) increased the ultimate load by 25.31%. This resulted in the average bond strength decreasing as the sheet width increased. This conclusion is similar to that of Ueda et al. (1999), which was discussed in

section 2.7.2. Their explanation was that the failure zone was wider than the width of the FRP sheet. This was also observed in the S-series but was only evident for the initial failure zone, which is the distance from the crack to the end of the chunk of concrete attached to the CFRP sheet (Figure 4-1d).

The initial failure zone was different in each test series as discussed in section 4.2.2. Increasing the width of P11 by 50% over that of P8 (from 100 to 150 mm), which is similar to that in the S-series, increased the ultimate load by 52.6%. This resulted in no significant change in the average bond strength as the bond width increased. This conclusion is different than that from the S-series and is similar to the findings of Brosens and Van Gemert (1999) and De Lorenzis et al. (2000) also discussed in section 2.7.2. Figure 4-48 also shows that the stiffness of the joint increased as the bond width increased.

Figure 4-49 shows the load-displacement response for specimens P12, P13, P8, and P11 with bond widths of 25, 50, 100, and 150 mm, respectively and having the same bond length of 150 mm. The figure shows that the overall stiffness of the joint increased as the CFRP sheet width increased, which is similar to the findings in the S-series. Figure 4-50 shows the CFRP sheet bond width vs. ultimate load for the same specimens. The figure shows that the ultimate load increases as the bond width increases. The figure also shows the sheet bond width plotted against the average bond strength. The average bond strength in the figure was calculated from equation (4-1). The diagram shows that increasing the bond width decreases the average bond strength to a certain limit at which the bond strength stabilizes with the increase in the bond width. This indicates the existence of an effective bond width beyond which no increase in the average bond strength is obtained. This effect has been studied in the parametric study that is detailed in Chapter 6.

The strain distributions in the centre of the CFRP sheets along the bond length of specimens S4 and S5 at different load levels are shown in Figure 4-51. The chart shows that at the same load of 20 kN, the strain distribution in S4, with smaller bond width, had higher values than those in S5 and also had a larger transfer length. Changing the
load level (P/P<sub>u</sub>) from 0.75 to 0.94 did not significantly change the transfer length of S4. By comparing the strain distributions for both specimens at P/P<sub>u</sub> of 0.75 and 0.94 it is seen that the strain values and transfer lengths decreases as the sheet width increases. The difference between the strain values in each specimen also decreases at higher load levels.

Figure 4-52 shows the strain distributions in the centre of the CFRP sheets along the bond length of specimens P11, P12, and P13 with the same bond length of 150 mm. At the same load of 10 kN, specimen P12 with bond width, w, of 25 mm had higher strain values than P13 with w of 50 mm. At the same load level of 0.5, specimens P11 and P13 had no significant difference in the strain values. Specimens P11, P12, and P13 also had no significant difference at the same load level of 70% of the ultimate load.

The S-series shows that the average bond stress decreases as the bond width increases. The majority of research results in the literature are based on narrow sheet widths, mostly 25 mm (Chajes et al., 1996 and Bizindavyi and Neale, 1999). This results in the average bond stress in these research results to be higher compared to the sheets used in strengthening actual structures, which tend to be much wider than 25 mm. The previous discussion shows the need to investigate the effect of the bond width on the value of the effective length and the average bond strength. The width effect was investigated in this research through conducting a parametric study using numerical models. The numerical models and the parametric study are detailed and discussed in Chapters 5 and 6, respectively.

## 4.6 Test Method

The test methods considered here are the modified push-apart test method and the pull-apart test method. Both are indirect test methods. The effect of the test method will be discussed by comparing the results of the S-series and the P-series specimens that had similar variables.

Both test methods exhibited similar failure modes except that unlike the P-series, it was not clear which side would fail in the S-series. The amount of concrete that stayed bonded to the CFRP sheet after failure was different in each test method due to the difference in the crack initiator (metal sheet) distance from the concrete face to which the CFRP sheets were bonded. This caused the S-series to behave as an uncracked strengthened member while the P-series behaved as a pre-cracked strengthened member.

The load-displacement relationships for specimens in the two test methods with similar bond widths of 150 mm are shown in Figure 4-53. The load-displacement behaviour in the two methods were essentially the same but with different magnitudes.

The strain distributions along the length of the sheet from both the P-series and the S-series at different load levels are shown in Figure 4-54 through Figure 4-56. Although the strain gauge locations were different, there was no significant difference between the strain distributions in the two methods when the load levels were both above or both below 70% of the ultimate load level. When the load level in one method was above 70% the ultimate load while the other method was below that level, there appeared to be a difference in the strain distribution, as shown in Figure 4-54 at 20 kN and Figure 4-55 at 30 kN.

The test method effect was studied in detail in Chapter 7 using numerical models. Four test methods were investigated and compared. These test methods are the modified push-apart test, the pull-apart test, the direct pull test, and the push-apart test. The numerical models used are described and the results from the various test methods analysed and compared.

# 4.7 L/w Ratio

Specimens S1, S3, and S4 had the same bond area but different values of the ratio of bond length to bond width, L/w, in order to study its effect on the bond strength. Figure 4-57 shows the load-displacement behaviour for these specimens at mid span. From the curves it is apparent that as the L/w ratio decreases, the stiffness of the sheet increases. The stiffness of the sheet is directly related to the sheet width rather than the bond area, where increasing the width by 50% (from 100 mm to 150 mm), and 33.3% (from 150 mm to 200 mm) increases the stiffness by 57.6%, and 26.9%, respectively.

Figure 4-57 also shows that S4 had a lower ultimate load than S1 and S3, which showed no significant difference in their ultimate loads. The low ultimate load in S4 is as result of the lower bond width of 100 mm compared to 150 and 200 mm in S1 and S3, respectively. A bond length of 150 mm, which is higher than the effective length, does not add to the ultimate load.

The strain distributions along the length of the CFRP sheets for the same specimens are shown in Figure 4-58. At a load, P, of 40 kN the strain values and the transfer lengths increase as the ratio L/w increases. The difference between the strain values and transfer length of specimens S3 and S4 are significantly more than those between specimens S1 and S3. The reason is attributed to the difference in load level, P/P<sub>u</sub>, and the difference in strain trend associated with it as discussed in section 4.3.1 and not as a result of the different L/w ratio. It was also observed that there is no significant difference between the transfer lengths at P/P<sub>u</sub> of 0.71.

From the previous discussion it is concluded that the individual effects of the CFRP sheet bond length and bond width are the governing factors in the behaviour rather than the L/w ratio.

### 4.8 Anchor Sheet Configuration

The effect of FRP anchor sheet configurations on the bond behaviour of FRP sheets was studied in an attempt towards developing FRP reinforcement schemes that could avoid premature delamination failures. Thirteen P-series specimens were tested with different anchor sheet configurations, designed to study the effect of locating the anchor sheet above or below the tested sheet on the bond behaviour. The effect of the distance of the anchor from the crack "a" and the effect of the anchor sheet extension beyond the tested sheet "b" were also studied. The distances a and b are illustrated in Figure 3-8. Specimen P20 was discarded as discussed in section 4.2.2 and specimen P26 was

repeated for verification of results and the average values from both tests were considered.

#### **4.8.1** Anchor Sheet Location Relative to the Tested Sheet

The effect of locating unidirectional CFRP anchor sheets above or below the tested CFRP sheet on the bond behaviour was investigated by comparing the results of six pairs of specimens with the same anchor configurations and dimensions except for the location of the anchor sheets relative to the tested sheet. Specimens P14 through P19 had the anchor sheets bonded above the tested sheet while specimens P21 through P26 had them bonded below the tested sheet.

The load-displacement responses of all specimens are shown in Figure 4-59 through Figure 4-64. The load was applied through stroke control. A sudden drop in the loaddisplacement curves indicates a crack opening. The drop in the load value at 0.5 mm displacement in Figure 4-59 was due to the sudden opening of the centre crack in the opposite unopened cracked face of the specimen. The descending branch in the curves indicates that a sheet was debonding, including the portion of the anchor sheet in line with the tested sheet. The sudden drop in the load at failure occurred when the sheet was totally debonded in a sudden zip action.

Comparison of the load vs. displacement responses in Figure 4-59 through Figure 4-64 indicates no significant difference in the load and displacement values in the case of specimens P15 and P22. Specimen P17 with the anchor sheet above the tested sheet achieved 11% higher load capacity but with 12.6% less displacement values relative to specimen P24, which had the anchor sheet below the tested sheet. In the last four pairs of specimens P14/P21, P16/P23, P18/P25, and P19/P26 the anchor below the sheet achieves between 10.9% and 25% higher load values and between 29.9% and 46.4% higher displacement values. The anchor sheet located below the tested sheet was, therefore, observed to show better or equivalent response compared to anchor sheets located above the tested sheet.

# 4.8.2 Anchor Distance from Crack

The load-displacement responses for specimens studying the anchor distance from the crack "a" for anchor sheets above the tested sheet and those below the tested sheet are shown in Figure 4-65 and Figure 4-66, respectively. When the anchor sheet was above the tested sheet, the distance "a" did not appear to affect the load capacity or the ductility of the joint except when the anchor sheet was at the crack face (a = 0). The average load capacity was slightly higher in the case of an anchor sheet below the tested sheet. This is illustrated in the ultimate load-distance "a" relationships shown in Figure 4-67. When the anchor sheet was at the crack face (a = 0) both anchor locations showed an increase in the load capacity and ductility of the joint.

### **4.8.3** Anchor Extension Beyond Tested Sheet

The load-displacement responses for specimens studying the extension of the perpendicular anchor sheets beyond the tested sheet "b" for anchor sheets above the tested sheet and those below the tested sheet are shown in Figure 4-68 and Figure 4-69, respectively. There was no significant difference in the ultimate load when the anchor sheet was above the tested sheet while the displacement did not have a defined trend. In the case of an anchor sheet below the tested sheet, neither the ultimate load nor the displacement had a defined trend. The ultimate load trend is illustrated in Figure 4-70, which shows the ultimate load plotted against the distance "b" for the specimens in Figure 4-68 and Figure 4-69.

From the previous discussion it is concluded that unidirectional CFRP anchor sheet located below the tested CFRP sheet showed better or equivalent overall bond behaviour compared to anchor sheets located above the tested sheet. The anchor distance from the crack "a" did not appear to affect the load capacity or the ductility of the joint, except when the anchor sheet was at the crack face (a = 0) where an increase in the load capacity and ductility of the joint occurred. The extension of the perpendicular anchor sheets beyond the tested sheet "b" does not appear to have a direct relation with the ultimate load or the ductility of the joint.

Test	L	w	L/w	dı	d <sub>2</sub>	с	¢	f' <sub>c</sub>	Pa	$\tau_{b}$
#	(mm)	(mm)		(mm)	(mm)	(mm)		(MPa)	(kN)	(MPa)
S1	75	200	0.38	20	45	61	36.4	51.1	54.81	1.83
S2	50	150	0.33	16	27	40	34.0	50.4	39.85	2.66
S3	100	150	0.67	17	32	36	41.6	51.3	56.40	1.88
S4	150	100	1.50	15	33	40	39.5	50.4	42.71	1.42
S5	150	150	1.00	26	30	38	38.3	51.1	53.52	1.19

Table 4-1 Push-apart bond test specimens' details and results

Table 4-2 Pull-apart bond test specimen's details and results

Test #	L (mm)	w (mm)	a (mm)	b (mm)	f' <sub>c</sub> (MPa)	Pu (kN)	τ <sub>b</sub> (MPa)	Remarks
P1	45	100	~		46.9	20.06	2.22	~~~~ <u>~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~</u>
P2	60	100			46.9	22.55	1.88	
P3	75	100	25	50	46.9	23.27	1.55	
<b>P</b> 4	90	100			46.9	23.04	1.28	
P5	105	100			46.9	24.30	1.16	Effective length
P6	120	100			46.9	33.60	1.40	(No anchor sheet)
P7	135	100			46.9	22.14	0.82	
P8	150	100			46.9	24.90	0.83	
	50	150	25	25	46.9	28.78	1.92	Length effect
P10	100	150	25	25	46.9	40.49	1.35	(Anchor sheet at one end)
P11	150	150	25	25	47.3	38.1	1.69	
P12	150	25	25	75	47.3	9.88	1.32	Width effect, including P11
P13	150	50	25	75	47.3	13.93	0.93	(Anchor sheet at one end)
P14	100	.50	0	75	47.3	18.77	1.88	₩ <u>₩₽₽₽₩₽₽₽₩₩₽₽₩₽₽₩₽₽₩₩₩₽₩₽₽₽₩₩</u> ₩₩₩₩₩₩₩₩₩
P15	100	50	25	75	47.3	15.55	1.56	
P16	100	50	50	75	47.3	15.96	0.94	Anchor at both ends
P17	100	50	75	75	47.3	16.19	1.62	Anchor sheets above
P18	100	50	25	50	47.3	15.49	1.55	
P19	100	50	25	25	47.3	15.19	1.52	
P20	100	50	·		47.3	31.48	3.15	
P21	100	50	0	75	46.4	20.33	2.03	······································
P22	100	50	25	75	46.4	15.08	1.51	
P23	100	50	50	75	46.4	18.26	1.83	Anchor at both ends
P24	100	50	75	75	46.4	15.61	1.56	Anchor sheet below
P25	100	50	25	50	46.4	19.17	1.92	
P26	100	50	25	25	46.4	18.09	1.81	



Figure 4-1 Crack propagation and failure zone formation sketch



Figure 4-2 Schematic sketch of the failure zone plan view in the push-apart tests



Figure 4-3 Push-apart failure surface (a) concrete (b) CFRP sheet



Figure 4-4 Side view of a pull-apart specimen showing crack opening



Figure 4-5 Typical failure mode for specimens P1 through P13, accept for specimen P3 (a) side view (b) CFRP sheet surface (c) concrete surface



Figure 4-6 Specimen P3 failure mode



Figure 4-7 Specimen P15 failure mode



Figure 4-8 Typical failure for specimens with anchor (a) above sheet (b) below sheet



Figure 4-9 Failure mode (a) Specimen P17 (b) Specimen P24



Figure 4-10 Average strain distributions for specimen S1



Figure 4-11 Average strain distributions for Specimen S2

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.



Figure 4-12 Average strain distributions for Specimen S3



Figure 4-13 Average strain distributions for Specimen S4



Figure 4-14 Average strain distributions for Specimen S5



Figure 4-15 Strain distributions at centre of sheet for specimen P9



Figure 4-16 Strain distribution at centre of sheet for specimen P10



Figure 4-17 Strain distributions at centre of sheet for specimen P11







Figure 4-19 Strain distributions at centre of sheet for specimen P13



Figure 4-20 Load vs. transfer length for S-series specimens with w = 150 mm



Figure 4-21 Load vs. transfer length for P-series specimens with w = 150 mm







Figure 4-23 Schematic sketch of deriving incremental average bond stress



Figure 4-24 Average bond stress in sheet centre vs. relative load for specimen P10



Figure 4-25 Strain distributions along the sheet width of S1 at crack location



Figure 4-26 Strain distributions along half the sheet width of S2 at 0, 15, 30, and 45 mm from the crack location



Figure 4-27 Strain distributions along the sheet width of S3 at crack location



Figure 4-28 Strain distributions along the sheet width of S4 at crack location



Figure 4-29 Strain distributions along the sheet width of S5 at crack location



Figure 4-30 Strain distributions across the width of P9 at crack location



Figure 4-31 Strain distribution across the width of P10 at crack location







Figure 4-33 Strain distributions across the width of P12 at crack location



Figure 4-34 Strain distributions across the width of P13 at crack location



Figure 4-35 CFRP sheet width vs. strain difference between the edge and centre of the sheet at the crack location



Figure 4-36 Strain distributions in the sheet centre and edge along its length at various load levels in S4



Figure 4-37 Average bond stresses along the sheet width VS. normalized load level, specimen S1



Figure 4-38 Average bond stresses across the sheet width vs. normalized load level, specimen S4



Figure 4-39 Average bond stresses across the sheet width VS. normalized load level, specimen S2



Figure 4-40 Average bond stresses across the sheet width vs. normalized load levels, specimen P10



Figure 4-41 Average bond stresses across the sheet width vs. normalized load level, specimen P11







Figure 4-43 Response for specimens with w = 150 mm







Figure 4-45 Strain distributions along the sheet centre at 20 kN, w = 150 mm



Figure 4-46 Response of specimens with the same sheet width, w = 100 mm



Figure 4-47 Load vs. bond length for specimens with w = 100 mm





Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.



Figure 4-49 Response of specimens with L = 150 mm



Figure 4-50 Width and average bond strength for P-series specimens, L = 150 mm



Figure 4-51 Strain distributions along centre of the sheet for S-series, L = 150 mm



Figure 4-52 Strain distributions along centre of the sheet for P-series, L = 150 mm



Figure 4-53 Response comparison for specimens with w = 150 mm



Figure 4-54 Strain distributions at centre of sheet for specimens P9 and S2



Figure 4-55 Strain distributions at centre of sheet for specimens P10 and S3



Figure 4-56 Strain distributions at centre of sheet for specimens P11 and S5







Figure 4-58 Strain distributions along the sheet centre, same bond area

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.







Figure 4-60 Response of specimens with a =25 mm and b=75 mm



Figure 4-61 Response of specimens with a = 50 mm and b=75 mm



Figure 4-62 Response of specimens with a =75 mm, b=75 mm










Figure 4-65 Effect of distance from crack "a", anchor above and b = 75 mm



Figure 4-66 Effect of distance from crack "a", anchor below sheet and b=75 mm



Figure 4-67 Anchor distance from crack for specimens with b = 75 mm



Figure 4-68 Effect of anchor extension beyond sheet "b" when the anchor is above the sheet and a = 25 mm



Figure 4-69Effect of anchor extension beyond sheet "b" when the anchor is below the sheet and a = 25 mm



Figure 4-70 Anchor spacing for specimens with a = 25 mm

# 5. NUMERICAL MODELS

## 5.1. Introduction and Overview

Chapters 3 and 4 described in detail the parameters tested experimentally in this research. These parameters covered a limited range of variability. In order to extend the range, a number of finite element models were developed. In this chapter these models are verified against the test results, and are used in the following chapters to produce additional data for the development of design charts and guidelines.

Two types of models were developed: using the commercial finite element program ABAQUS (1998) two-dimensional (2D) and three-dimensional (3D) models. Two 2D non-linear finite element models were developed, one each to simulate the modified push-apart test and the pull-apart test; both were previously tested experimentally and detailed in Chapter 3. As mentioned above, 3D non-linear finite element models were also developed in order to study the strain distribution across the width of the CFRP sheets, which could not be performed using the 2D models. The numerical models were developed with the objective to conduct a parametric study on the main parameters affecting the bond behaviour between FRP sheets and concrete. The effect of the type of test conducted on the test results was also investigated using the numerical models. To confirm the validity of the models, the output was compared to the experimental results from the test series detailed in Chapters 3 and 4. All the models are detailed and discussed in this chapter together with a comparison between the numerical output and experimental results.

The formulation of the problem in all cases was based on small displacements and infinitesimal strains. Material nonlinearities were accounted for as well as debonding in the form of interface surfaces, where possible. The analysis was conducted incrementally with equilibrium established in every increment iteratively.

# 5.2. Geometric Modeling

# 5.2.1. 2D Models

Only one half of each specimen in both the modified push-apart test and the pullapart test was modeled making use of symmetry along the line of action of the loads, which is shown as axis of symmetry 1 in Figure 5-1 and Figure 5-2. There was no symmetry along axis of symmetry 2 in both tests because of the existence of an anchor sheet on one side of the tested sheet, which was placed in order to ensure the failure of the other unanchored side. The effect of the anchor sheet was taken into consideration in the model as will be explained later. The geometry and mesh dimensions of the modified push-apart test model and half the pull-apart test model are illustrated in Figure 5-3 and Figure 5-4.

The concrete mesh was built from 10×10 mm four-node plane stress elements with reduced integration (ABAQUS element CPS4R). The 10×10 mm dimension was the minimum mesh size, considering the maximum aggregate size used in the specimens. Reduced integration was used in order to soften the behaviour of the model because numerical solutions using four node elements are generally stiffer than observed in the experiments (Bathe, 1996). First order quadrilateral elements with reduced-integration were also chosen based on recommendations of ABAQUS/Standard (1998) for their use in problems involving contact or large distortions.

The CFRP sheet mesh was built of 0.38×10 mm CPS4R elements, where the thickness of the sheet was 0.38 mm and the 10 mm side was taken similar to the concrete mesh dimension to facilitate the formation of the initially bonded condition. Four-node plane stress elements were also used here similar to the concrete elements for compatibility at the contact nodes. Node to node correspondence was established with the adjacent elements at the concrete surface. Contact and debonding are modelled at that interface.

The steel rebars in the modified push-apart model were modeled using onedimensional strain compatible threads that can be defined singly or embedded in oriented surfaces. These elements are superposed on the mesh of plain concrete elements and are used with a standard metal J2 plasticity material constitutive relation.

On the other hand, the steel rebars in the pull-apart test were modeled using 10 mm long truss elements. The rebars were 400 mm long with 250 mm of their length embedded in the concrete specimen. Since half of the specimen was modeled along the rebar axis, only half the bar cross-section area was specified.

The 20 mm thick steel plate used to uniformly distribute the load to the inner face of the modified push-apart specimen was modeled with a mesh composed of 10×20 mm CPS4R elements. The elements dimensions were taken to match the concrete elements. Figure 5-3 and Figure 5-4 show the geometry and mesh of the 2D modified push-apart and half the pull apart models, respectively.

#### 5.2.2. 3D Models

Due to symmetry in geometry and loading, one eighth of both the modified pushapart specimen and the pull-apart specimen were modeled along the planes of symmetry 1, 2 and 3 shown in Figure 5-5a and Figure 5-6a. This reduced significantly the size of the models and in return reduced the computational time. The portion that was modeled is shown in Figure 5-5b and Figure 5-6b for a typical modified push-apart specimen and a typical pull-apart specimen, respectively.

Concrete was modelled with a fine mesh at the bonded CFRP sheet locations in the modified push-apart model in order to obtain more accurate results at the stress concentration zones. A coarse mesh was used at locations away from the stress concentrations in order to reduce the mesh size and in turn the computational time. The fine mesh in the modified push-apart model was built from  $10 \times 10 \times 10$  mm eight-node cube elements with reduced integration (C3D8R), where the cube size was limited by the

maximum aggregate size of 10 mm used in the specimens. The pull-apart model had a concrete mesh built from  $10 \times 10 \times 10$  mm C3D8R; the model was relatively small.

The mesh for the CFRP sheet was built from  $10 \times 10$  mm four-node shell elements with reduced integration. The  $10 \times 10$  mm dimensions were taken similar to the corresponding concrete element mesh dimension at the face to facilitate bonding the corresponding nodes together for compatibility at the contact surface. The thickness of the shell elements was 0.38 mm, which is the CFRP sheet thickness in the experimental tests.

The steel rebars in the modified push-apart model were simulated using one dimensional strain compatible threads, similar to those used in the 2D model. The steel rebar in the 3D pull-apart model was modeled using  $10 \times 10 \times 10$  mm C3D8R elements, where the use of line elements caused local failure and distortion of the concrete elements at the loaded embedded rebar to concrete location. Since one eighth of the specimen was modeled along the rebar axis, only one column of elements was specified for the rebar. The rebar elements were continuous inside the concrete mesh and were assumed to have perfect bond with the concrete.

The steel plate used to load the specimen in the modified push-apart model was modeled with a mesh of  $10 \times 20 \times 30$  mm C3D8R elements. The  $10 \times 30$  mm dimensions were taken similar to the opposing concrete element face dimension for compatibility at their interface and the 20 mm side was the thickness of the steel plate.

Figure 5-7a and Figure 5-8 show the geometry and mesh of the 3D modified pushapart and pull apart models, respectively.

## 5.3. Boundary Conditions and Loading

The modified push-apart tests were self-contained with the load applied with a hydraulic jack placed in the core of the concrete specimen. The specimen is placed on a

flat steel surface with the hydraulic jack placed on steel saddles in a horizontal position. On the other hand, the push-apart tests were placed vertically in a universal testing machine with the load applied to the specimen through the rebars. Symmetry was utilized in order to simplify the models and to reduce computational time. The proper boundary conditions at the axis and planes of symmetry for the 2D and 3D models, respectively, were applied and are illustrated and explained in the following subsections.

#### **5.3.1.** Boundary Conditions and Loading of 2D Models

One half of the modified push-apart and pull-apart specimens was modeled making use of the symmetry along the line of action of the load, which is axis of symmetry 1 in Figure 5-1 and Figure 5-2. Rollers were taken along axis of symmetry 1 in the figures, where the movement was allowed in the direction of the load and was restrained in the perpendicular direction. The boundary conditions for the modified push-apart and pull-apart 2D models are shown in Figure 5-3 and Figure 5-4, respectively.

The load was applied quasi-statically to the models in a displacement control mode, which is similar to that employed in the experimental tests. Displacement control was implemented in the model as a roller support that moves with specified increments of displacement. The load location was at the centre of the steel bearing plates in the modified push-apart model, where the hydraulic jack applied the load to both ends. On the other hand, the load location for the pull-apart model was at the end of one of the steel bars while the other end had a hinge preventing movement and rotation in all directions, which simulates the machine grip in the experimental tests. An initial displacement of 0.01 mm was applied with increments of 0.01 mm for a total displacement of 1.0 mm, which was taken higher than the maximum displacement in the experimental tests.

#### 5.3.2. Boundary Conditions and Loading of 3D Models

As mentioned in Section 5.2.2, one eighth of both the modified push-apart specimen and the pull-apart specimen were modeled in the 3D models due to symmetry along the planes of symmetry 1, 2, and 3, shown in Figure 5-5a and Figure 5-6. Rollers were placed along planes of symmetry 1, and 2. The rollers allowed the movement of the

joints in the direction parallel to the planes of symmetry but restrained their movement in the perpendicular direction due to symmetry in geometry and loading. Rollers were also placed at the FRP sheet intersection with the plane of symmetry 3. The boundary conditions for the modified push-apart and pull-apart 3D models are shown in Figure 5-7b and Figure 5-8, respectively.

The end of the steel rebar was fixed in order to simulate the test machine grips on the bar ends. The load was applied quasi-statically to the models in displacement control similar to the 2D model except that the analysis procedure was different as will be discussed in Section 5.6.

## **5.4.** Material Properties

The material properties in the models were taken similar to those obtained from the experimental test program, while properties that were unavailable from the tests were reasonably assumed. The materials used in the experimental tests were concrete, CFRP, adhesive, and steel. The modeling of each material property in the numerical model is discussed in detail in the following subsections.

## 5.4.1. Concrete

The constitutive theory implemented in the ABAQUS concrete material model was used and is fairly simple. When the principal stress components are predominantly compressive, the response of the concrete is modeled with an elastic-plastic constitutive theory, using a simple form of yield surface. In that case, associated flow is assumed and isotropic hardening values are used. In tension, however, cracking in the concrete is assumed to occur when the stresses reach a failure surface, called the "crack detection surface" shown in Figure 5-9, which shows the concrete failure surface in plane stress.

The shape of the concrete failure surface is defined through several parameters, which depend on the concrete material properties. These parameters are the ratio of the ultimate biaxial compressive stress to the uniaxial compressive ultimate stress, which was taken as 1.16, and the absolute value of the ratio of uniaxial tensile stress at cracking to the ultimate uniaxial compressive strength, assumed to be 0.053. The ratios of the

magnitude of the principal component of plastic strain at ultimate stress in biaxial compression and at cracking to the tensile cracking stress under uniaxial were taken to be 1.28 and 0.333, respectively. These are all default values generally accepted by researchers and recommended by ABAQUS (1998) except for the uniaxial tensile to compressive stress ratio, which was taken to match the experimental tests.

A linear elastic portion and a nonlinear plastic portion defined the concrete behaviour in compression. The linear elastic portion of the concrete behaviour was defined by Young's modulus ( $E_c$ ) and Poisson's ratio ( $v_c$ ). The value of  $E_c$  was calculated in accordance with the CSA Standard A23.3-94, (1995) recommended 45% secant modulus:

$$E_c = 4500 \sqrt{f_c}$$
(5-1)

in which  $f_c$  is the uniaxial concrete compressive strength. The value of  $v_c$  was taken as 0.15, which is a typical concrete value. The concrete was assumed to have isotropic behaviour in the elastic range.

The properties of the concrete in compression outside the elastic range were defined through the input of the absolute value of the compressive yield stress ( $f_{cy}$ ), the compressive strength ( $f_c$ '), and the absolute value of the plastic strain ( $\varepsilon_{cp}$ ), which corresponds to elastic unloading from  $f_c$ '. The value of  $f_c$ ' was taken similar to the experimental tests, which were 51 MPa and 47 MPa in the modified push-apart and the pull-apart test, respectively, and the plastic strain was taken as 0.0013 in both models. The concrete behaviour beyond  $f_{cy}$  produces some unrecoverable (inelastic) straining and the response of the material softens. An ultimate stress ( $f_c$ ') is reached after which the material softens until it can no longer carry any stress. The failure stress is defined by the failure surface described previously.

The concrete model in ABAQUS is a smeared crack model in the sense that it does not track individual micro cracks, rather state variables are calculated independently at each integration point of the model, and are assumed to represent the behaviour of the material in the region in a gross manner. The presence of cracks then enters into these calculations by the manner in which the cracks affect the stress and material stiffness associated with an integration point.

The modeling of the concrete-rebar interaction and the energy release during cracking are critical to the response in the post cracking range in tension. This effect is introduced in the model using an option in ABAQUS that models the associated concrete-rebar interaction as a gradual loss of strength in reinforced concrete beyond the cracking strain, known in the literature as tension stiffening, or alternatively the energy release as cracking propagates across the volume represented by a material point, known in the literature as tension softening (Chen, 1982).

Introducing the tension stiffening effect is intended to model the fact that concrete cracks discretely, so that between the cracks the rebar strains are much lower than at the cracks. This behaviour is dependent on the ratio of reinforcement volume to concrete volume. In ABAQUS this effect is not well defined. In any case, proper tension stiffening behaviour representation is an issue in micro behaviour. In our case, this is not an issue, because cracking at the interface of concrete and FRP sheet is away from the rebar, and tension softening is more important. Tension softening in the post-cracking regime was defined by defining the displacement, u<sub>o</sub>, at which a linear loss of stiffness after cracking, caused by the full development of a crack process zone, gives zero stress. Properly the value of u<sub>o</sub> should yield energy of fracture per unit area typical of the particular concrete under consideration. For our concrete and mesh size, the value of u<sub>o</sub> was equivalent to ten times the value of displacement at cracking, u<sub>ct</sub>, and was calculated as shown in Figure 5-10. This is a limiting value for standard concrete behaviour and gives a reasonable match to experimentally measured responses. The plain concrete uniaxial behaviour in both the compression and tension regions is shown in Figure 5-11.

#### 5.4.2. CFRP

The CFRP in both models was in the form of sheets bonded to the concrete surface. The CFRP sheets used have a linear-elastic stress-strain relationship ending with brittle failure. The CFRP sheets material was defined as orthotropic with a strong direction in

the direction of the fibres. The general stress-strain behaviour for the CFRP material is shown in Figure 5-12a. The cross direction strength is governed by the matrix strength. In our tests, which were unidirectional, this was not an issue.

The CFRP sheet orthotropic elasticity in plane stress is defined by the values of the modulus of elasticity in the main and secondary direction in the plane of the sheet ( $E_1$ ,  $E_2$ ), the shear modulus ( $G_{12}$ ,  $G_{13}$ , and  $G_{23}$ ), and Poisson's ratio ( $v_{12}$ ). The model used is valid for small elastic strains and must satisfy the conditions of Drucker Stability (ABAQUS/Standard, 1998). The values of  $E_1$  and  $v_{12}$ , were taken as 50000 MPa and 0.26, respectively, which match the manufactures specified properties. The values of  $E_2$ ,  $G_{12}$ ,  $G_{13}$ , and  $G_{23}$  were taken as 2000, 4000, 4000, and 4000 MPa, respectively, which are taken from the literature (ABAQUS/Standard, 1998) to represent matrix properties.

Plane stress orthotropic failure measures were used in the model. The measures are indications of material failure, normally used for FRP materials (ABAQUS/Standard, 1998). The input data for the stress-based failure theories used in ABAQUS are the tensile and compressive stress limits in both the main and secondary direction of the CFRP sheet and the maximum shear stress in its plane. Only the tensile strength in the main direction is important and it was taken to be 3400 MPa similar to the manufacturer specified properties.

#### 5.4.3. Adhesive

In an earlier stage of the work (not shown here) a parametric study was carried out to determine the effect of the elastic properties of the adhesive layer between the CFRP sheet and concrete on the behaviour, and whether the thickness of this layer could be ignored in the model. No significant change in the strain distributions in the CFRP sheet or in the load-displacement response was observed when the adhesive layer thickness was disregarded. The debonding behaviour in the model was unchanged with and without the adhesive layer. This finding was in agreement with previous conclusions by Fashole-Luke (1999). Consequently, it was decided not to model the thickness of the adhesive layer in the analysis. Details of the CFRP-concrete interface modelling are discussed in Section 5.5.

## 5.4.4. Steel

The steel used in the specimens was in the form of bars and plates. The specific modelling in each case was discussed above. The material model used was an elastoplastic model for the rebar and linear elastic for the bearing plate. The elastic behaviour for both materials was governed by the modulus of elasticity ( $E_s$ ) and Poisson's ratio ( $v_s$ ). The values of  $E_s$  and  $v_s$  were taken as 220000 MPa and 0.3, respectively. Beyond the elastic range the rebar behaviour was modeled by defining the yield strength ( $f_{sy}$ ), ultimate strength ( $f_{su}$ ), and the plastic strain at ultimate strength ( $\varepsilon_{sp}$ ). The values of  $f_{sy}$  and  $f_{su}$  for the rebars were taken as the manufacturers specifications as 425 MPa and 645 MPa, respectively, while those for the steel plate were taken as 350 MPa, 500 MPa, respectively. A typical value for  $\varepsilon_{sp}$  was taken as 0.096 for both the steel rebar and plate. The stress-strain behaviour for the steel taken in the models is shown in Figure 5-12b.

# 5.5. Concrete-CFRP Sheet Interface

The CFRP sheet was modeled as initially bonded to the concrete surface in the 2D models then debonded at the interface joints when the interface stresses reached a defined critical limit. The crack location was predefined at the specimen centre line of symmetry. Introducing the debonding of the CFRP sheet from the concrete surface was essential in modeling the failure mechanism. The CFRP sheet and concrete interface modeling in the 2D models is explained below.

The nodes of the shell elements that model the CFRP sheets defined a surface, while the corresponding nodes on the concrete face defined another surface. In order to identify the initially bonded condition of the CFRP sheet to the concrete, the nodes on the contact surfaces of both the CFRP sheet and the concrete elements were constrained such that, initially, the displacements of each corresponding pair of nodes were identical. As the width of the CFRP sheet was less than that of the concrete specimen, the widths of the contact surfaces were taken equal to the sheet bonded width.

The ABAQUS contact surface model assumes one surface is a master surface (in this case the concrete surface) and the other is a slave surface (CFRP sheet surface). Critical interface shear ( $\tau$ ) and normal stresses ( $\sigma$ ) are at the contact interface nodes. Debonding of the slave surface from the master surface occur at the initially bonded interface nodes when the value of a critical interface stress criterion (f) reaches 1.0 within a certain tolerance " $\eta$ ", default value of 0.1, such that

$$1 - \eta \le f \le 1 + \eta \tag{5-2}$$

in which "f" is defined by:

$$f = \sqrt{\left(\frac{\sigma_n}{\sigma_f}\right)^2 + \left(\frac{\tau_n}{\tau_f}\right)^2}$$
(5-3)

Here  $\sigma_n$  and  $\tau_n$  denote observed normal and shear stresses carried across the interface, and  $\sigma_f$  and  $\tau_f$  are limiting failure values. The adhesive normal and shear strengths,  $\sigma_f$  and  $\tau_f$ , were taken from reported values in the literature as 6.0 and 3.5 MPa, respectively.

In order to activate the debonding between the FRP sheet and concrete surfaces, ABAQUS requires that the first node in the bonded surfaces be initially debonded to initiate the debonding mechanism. Therefore, in test specimen locations where an anchor sheet was placed, this node was not debonded to force failure in the unanchored side. Figure 5-13b shows magnification of the CFRP sheet debonding from the concrete in the 2D modified push-apart model. It was observed in the experimental tests that a chunk of concrete stayed bonded to the FRP sheet after the debonding failure of the specimen, as detailed in Section 4.2.1. This was not simulated in the tested part of the sheet due to the unbonding of the first node in the bonded surfaces. The strain in the load direction contour of the modified push-apart model at failure shown in Figure 5-14, shows that large strain values occur at the concrete loaded corner, which indicate the eventual failure of that portion of the concrete. The debonding option in ABAQUS is also limited to 2D models. Since the 3D simulation was intended to investigate the strain distribution FRP sheet width and the sheet width to concrete width ratio effect, it was felt that debonding is not an issue, and the surface nodes in the 3D model were fully constrained.

#### 5.6. Analysis Procedures

A nonlinear finite element analysis procedure was used to determine the response of the model. The solution is found in ABAQUS by specifying the loading as a function of time and incrementing time to obtain the nonlinear response. Therefore, by breaking the simulation into a number of time increments, an approximate equilibrium configuration is found at the end of each time increment.

The first increment size was assumed to be 0.1% of a limiting step of 1.0, which was chosen after several initial trials with higher increments that caused the program to stall. Subsequently, automatic control of the increment size was such that the increment size changes as the program develops the response in the step. This approach is more efficient than using a user defined increment size, where the user cannot predict the response ahead of time. The program determines if convergence is likely in a reasonable number of iterations by comparing consecutive values of the maximum achieved force residuals following every iteration. If subsequent iterations show the solution to be closer to equilibrium and convergence is likely, the program continues with the iteration process with the current increment size. If convergence is not likely, the program chooses a smaller increment size but up to a minimum specified value in order to prevent excessive computation in cases where a limit load approaches or some modeling error grows causing the solution to stall. At the end of each increment the structure is in approximate equilibrium and the results are recorded.

To avoid premature cutbacks of the time increment and stalling of the program because of the high nonlinearity in the system, the analysis was defined as discontinuous. This had the effect of increasing the default number of equilibrium iterations after which a logarithmic rate of convergence check is made. The iteration number at which check of

the logarithmic rate of convergence begins also increased as a result of defining the analysis as discontinuous.

Reinforced concrete solutions involve regimes where the load-displacement response is unstable because of excessive cracking. Since considerable nonlinearity was expected in the response, the modified RIKS solution strategy was used in the 3D model with automatic incrementation. The RIKS procedure uses the load magnitude as an additional unknown together with the displacements and uses 1% extrapolation of the strain increment. The 2D model did not use RIKS method, where the method does not work with problems involving debonding (loss of contact), as was the case in the 2D model. Modified Newton's iterations were used to solve the nonlinear equilibrium equations in all models.

## 5.7. Verification

The numerical models were verified by comparing their output with the experimental results that were discussed in detail in Chapter 4. The comparisons between the experimental results and the model output were performed on the overall load-deflection behaviour of the specimens and the strain distribution in the CFRP sheets at different load levels along the sheet length.

In order to find the overall load-displacement behaviour of the specimen, the reaction and displacement at the applied moving roller at all time increments were recorded throughout the analysis. The values were then plotted and compared to the corresponding experimental plots. As the program ABAQUS plots the strain distribution at a specified load increment, the strain distribution at the required load level was obtained by first determining the displacement at the required load from the load-displacement response. The strain distribution was then plotted along the specified nodes at the load increment corresponding to that specific displacement level.

128

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

# 5.7.1. 2D Model Verification

As mentioned before, Figure 5-13a shows the modified push-apart model mesh in a deformed configuration with a magnification factor of 30 and Figure 5-13b shows the CFRP sheet in the model debonding from the concrete with a magnification factor of 100. The figures shows that the model behaves similarly to the experimental tests, where the CFRP sheet separates from the concrete surface as the critical interface stresses are reached. Similar results were obtained from the pull-apart model, as shown in Figure 5-15. The similarity in behaviour between the model and the experimental tests shows that the debonding type failure can be simulated using numerical models.

The overall behaviour comparison between the 2D modified push-apart numerical model output and the experimental results shows good agreement in the overall load-displacement behaviour as shown in Figure 5-16 through Figure 5-20 for specimens S1 through S5. These figures describe the total load plotted against the displacement between the loading points. The figures also show consistently higher strength predicted by the numerical model varying from 3.5% to 17.2%. This higher strength could be attributed to the variability in the adhesive-concrete interface strength.

The pull-apart model had the same trend as the experimental results, as shown in the load-deflection curves in Figure 5-21 through Figure 5-25 for specimens P9 through P13, respectively. These figures describe the total load plotted against the displacement at the loading point. The difference in values is attributed to the fact that the numerical model represents the perfect case with no eccentricity and the assumption that the load is transferred equally between the two bonded faces of the specimen.

A comparison is also made between the average strain distribution along the CFRP sheet length observed from the modified push-apart experimental results and that from the model at different load values. The comparison shows good agreement as shown in Figure 5-26 through Figure 5-30 for specimens S1 through S5. The introduction of the debonding effect allows modelling of the increase in the transfer length of the strain

distribution. The increasing transfer length in the strain distribution indicates the debonding of the sheet from the concrete as the load increases.

The strain distribution from the experimental tests and the numerical model along the FRP sheet length for specimens P9 through P13 are shown in Figure 5-31 through Figure 5-35. The experimental strain distribution for the pull-apart tests was measured at the centre of the sheet. The 2D model, however, represents the average strain across the width, which causes the model to predict higher strain values. With these taken into consideration, the strain distribution in the 2D pull-apart model is adequate for the overall trends.

#### 5.7.2. 3D Model Verification

The 3D models were developed to study the strain distribution across the CFRP sheet width, which the 2D models could not investigate. The models were verified by comparing the strain distributions across the CFRP sheet width with the corresponding strain distributions from the experimental tests. The strain distributions from the experimental tests were taken as the average values from both bonded FRP sheets. As the debonding effect was not introduced in the 3D models, the load-deflection behaviour was affected and the model had a stiffer response at higher loads compared to the experimental response. This is shown in Figure 5-36 through Figure 5-39 for the modified push-apart model and in Figure 5-40 through Figure 5-44 for the pull-apart model. Therefore, the loads at which the strain distribution comparison is made are limited to lower load levels.

The model and experimental strain distributions across the CFRP sheets at different load levels at the crack location for specimens S1 through S5 are shown in Figure 5-45 through Figure 5-49, respectively, while those for P9 through P13 are shown in Figure 5-50 through Figure 5-54, respectively. Only one side of the experimental tests in the Pseries specimens had strain gages across the sheet width. This could explain the big difference between the model and experimental strain of P12, which failed on the side with no instrumentation across the sheet width. The numerical model represented the

perfect case with no material irregularity or eccentricity in loading. The differences in the strain values between the models and the experimental tests were also attributed to the smaller CFRP sheet width used in the models due to the chosen mesh size. The specimens effected were S2, S3, and S5 in the S-series and all P-series, P9 through P13, where the 25, 50, and 150 mm widths were taken narrower in the models by 5, 10, and 10 mm, respectively. The model prediction of the strain distribution across the width and the relative strain distribution at different loads is seen as adequate to perform an investigation on the factors affecting the strain distribution across the FRP sheet width.

From the previous discussions, the good agreement of the numerical simulation with the experimental results in the 2D model and the trend agreement of the 3D model prove that it is possible to predict the load-displacement behaviour, the debonding type failure, and the strain distribution along the length as well as across the width of FRP sheets bonded to concrete using numerical models. This lends credibility to performing a study on various parameters and their effect on the bond behaviour of FRP sheets bonded externally to concrete surfaces using the proposed numerical models. It was decided that the modified push-apart model be used for the parametric study, where it showed better simulation with the corresponding experimental tests that had more strain readings than the pull-apart tests. The parametric study is detailed in the following chapter.



Figure 5-1 Modified push-apart specimen plan view and symmetry lines



Figure 5-2 Pull-apart specimen elevation and symmetry lines

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.



Figure 5-3 2D modified push-apart model mesh, loading, and boundary conditions



Figure 5-4 Half the 2D pull-apart model mesh, loading, and boundary conditions



Figure 5-5 Modified push-apart specimen (a) planes of symmetry (b) modeled part







(b) Loading and boundary conditions

# Figure 5-7 Modified push-apart 3D model



Figure 5-8 Pull-apart test mesh, loading, and boundary conditions



Figure 5-9 Concrete yield and failure surfaces in plain stress (ABAQUS, 1998)

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.



Figure 5-10 Calculation of the displacement at which a linear loss of strength after cracking gives zero stress



Figure 5-11 Uniaxial behaviour of plain concrete (ABAQUS, 1998)







Figure 5-14 Strain contour in load direction for modified push-apart model



Figure 5-15 2D pull-apart mesh, boundary conditions, and deformed shape



Figure 5-16 Load-displacement curves experimental and 2D model of S1



Figure 5-17 Load-displacement experimental and 2D model of S2







Figure 5-19 Load-displacement experimental and 2D model of S4



Figure 5-20 Load-displacement experimental and 2D model of S5



Figure 5-21 Load-displacement for experimental and 2D model of P9



Figure 5-22 Load-displacement for experimental and 2D model of P10



Figure 5-23 Load-displacement for experimental and 2D model of P11



Figure 5-24 Load-displacement for experimental and 2D model of P12



Figure 5-25 Load-displacement for experimental and 2D model of P13



Figure 5-26 Strain distributions for experimental and 2D model of S1



Figure 5-27 Strain distributions for experimental and 2D model of S2



Figure 5-28 Strain distributions for experimental and 2D model of S3



Figure 5-29 Strain distributions for experimental and 2D model of S4



Figure 5-30 Strain distributions for experimental and 2D model of S5



Figure 5-31 Strain distributions for experimental and 2D model of P9


Figure 5-32 Strain distribution for experimental and 2D model of P10



Figure 5-33 Strain distributions for experimental and 2D model of P11



Figure 5-34 Strain distributions for experimental and 2D model of P12











Figure 5-37 Load-displacement for experimental and 3D model of S3



Figure 5-38 Load-displacement for experimental and 3D model of S4



Figure 5-39 Load-displacement for experimental and 3D model of S5



Figure 5-40 Load-displacement for experimental and 3D model of P9



Figure 5-41 Load-displacement for experimental and 3D model of P10



Figure 5-42 Load-displacement for experimental and 3D model of P11



Figure 5-43 Load-displacement for experimental and 3D model of P12



Figure 5-44 Load-displacement for experimental and 3D model of P13



Figure 5-45 Strain distributions across sheet width of S1 at the crack location



Figure 5-46 Strain distributions across half sheet width of S2 at the crack location



Figure 5-47 Strain distributions across sheet width of S3 at the crack location



Figure 5-48 Strain distributions across sheet width of S4 at the crack location



Figure 5-49 Strain distributions across sheet width of S5 at the crack location



Figure 5-50 Strain distributions across the width of P9 at crack location







Figure 5-52 Strain distributions across the width of P11 at crack location







Figure 5-54 Strain distributions across the width of P13 at crack location

159

# 6. PARAMETRIC STUDY

#### 6.1 Introduction

Design guidelines are still developing for specific retrofitting requirements using FRP sheets bonded to concrete structures. Such guidelines are to govern how fibre systems should be tailored in order to provide adequate improvements in strength, ductility, or both. Evaluating the effect of externally bonded FRP sheets based on their tensile strength and using guidelines similar to those for conventional steel reinforcement in concrete members is insufficient. The failure modes associated with FRP sheets externally bonded to concrete (see Section 2.5) need to be accounted for in the design. It is difficult, expensive, and time consuming to optimize FRP performance completely experimentally. Alternatively numerical models that consider both concrete and FRP properties through mechanical interaction provide a possible means of extending the available database.

The following subsections will investigate the debonding type failure, which could occur in members strengthened in flexure or shear. The numerical models used are able to investigate this type of failure by allowing the FRP sheet to debond. A discussion and verification of the 2D and 3D numerical models used are detailed in Chapter 5. The good agreement of the model trends with the experimental results were seen as sufficient for the models to be used to conduct a parametric study on the behaviour of FRP sheets bonded to concrete. Debonding is assumed to take place when the normal and shear stresses at the interface reach an interaction function defined by critical values of 6.0 and 3.5 MPa, respectively. These levels are based on values reported in the literature, (Section 2.6.3), and the values used in the numerical models in Chapter 5.

The aim of the parametric study presented in this chapter was to provide further insight into the behaviour of FRP sheets bonded to concrete. The literature review presented in Chapter 2 and the experimental program carried out in this study, presented in Chapters 3 and 4, have both shown that the main parameters affecting the

bond behaviour between concrete and FRP sheets are the FRP sheet bond length, bond width, stiffness, and the concrete strength. These parameters are investigated in the parametric study presented in this chapter. The FRP sheet bond length and width were also investigated in the experimental tests, detailed in Chapters 3 and 4, but with considerably lower values and smaller range than the numerical parametric study. In addition to the previous parameters, the strain distribution across the FRP sheet width and the relative FRP sheet width to the concrete specimen width are investigated.

Table 6-1 shows the concrete and FRP sheet default material properties used in the parametric study unless otherwise stated. A summary of the variables investigated is given in Table 6-2 through Table 6-7. The tables indicate that a total number of 124 specimens were analyzed. For each parameter investigated, the numerical model generated several relationships including load-deflection responses and strain distributions along the length and across the width of the FRP sheet. The relationships generated were further used to plot relationships that provide insight into the bond behaviour. Each parameter investigated is discussed in detail in the following subsections.

#### 6.2 FRP Sheet Bond Length

The effect of the bond length of the FRP sheet on the bond behaviour was studied using the 2D modified push-apart model. The bond length was studied in relation to the bond width, stiffness, and concrete strength; each was varied individually while maintaining the other parameters constant. Ultimate load and average bond strength were plotted against the FRP sheet bond length at different bond width, stiffness, and concrete strength values in order to investigate the bond length effect on the bond behaviour and the effect of each variable on the effective bond length.

The variables in the investigation of the bond length effect with varying bond widths were the FRP sheet bond length and width, while the FRP sheet stiffness and

concrete strength were maintained at 25 GPa mm and 50 MPa, respectively. The bond lengths (Table 6-2) varied from 50 mm to 200 mm. This range of bond lengths was chosen in order to cover a wide range of bond lengths and was based on the literature value range of the effective bond length. The bond width was varied from 25 to 200 mm, which was 12.5% and 100% of the concrete width, respectively. The smallest width of 25 mm was used in several research investigations, specially the direct pull test, while widths over 100 mm were closer to the widths used in practice.

The ultimate load values for models with various bond lengths were obtained from the load-displacement responses, similar to that shown in Figure 6-1. The figure shows the load-displacement response for models with the same FRP sheet bond lengths of 150 mm. The ultimate load plotted against the bond length at bond width values ranging from 25 mm to 200 mm is shown in Figure 6-2. It is observed from the figure that the ultimate load increases as the bond length increases but beyond a certain bond length, which was different for each bond width, there is no increase in the ultimate load. This bond length is defined as the effective bond length. The reason why the ultimate load does not change can be attributed to the fact that debonding of the FRP sheet propagates without an increase in the applied force in the sheet for the longer bond lengths.

The effective bond lengths are determined for each bond width from Figure 6-2 by estimating the bond length beyond which no increase in the ultimate load is gained. The locations of the estimated effective bond lengths are marked on Figure 6-2 by vertical dotted lines. There is no significant change in the ultimate load as the bond length increases for models with bond widths of 25 mm. The effective bond length, although not clear, is estimated to be 110 mm for bond widths of 25 mm. The effective bond length plotted against the bond width is shown in Figure 6-3. The figure shows that the effective bond length increases as the width increases. The reason for the increase in the average bond stress across the FRP sheet as the bond width increases, which requires a larger effective bond length in order to develop the

full bond capacity. This will be discussed in the following subsection. The literature reports effective bond length values ranging from 60 to 275 mm as discussed in Section 2.7.1. To our best knowledge, there has been no research relating the effective bond length to the bond width.

The computed average bond strengths for models investigating the bond length effect with varying bond width values are listed in Table 6-2. The average bond strength was calculated by:

$$\tau_{\rm b} = \frac{P_{\rm u}}{L \cdot {\rm w}} \tag{6-1}$$

in which,  $P_u$  is the ultimate load, and L and w are the FRP sheet bond length and width, respectively. The average bond strength plotted against the FRP sheet bond length at bond width values ranging from 25 to 200 mm is shown in Figure 6-4. The figure shows that the average bond strength decreases as the bond length increases. There is also no significant difference in behaviour for bond widths higher than 100 mm. Figure 6-5 shows the average bond strength evaluated on the basis of the effective bond lengths estimated from Figure 6-3. The figure shows that the average effective bond strength values are constant beyond the effective bond lengths and that they decrease as the FRP sheet bond width increases.

Models investigating the bond length effect at various FRP sheet stiffness values are listed in Table 6-3. The table shows that the only variables are the FRP sheet bond length and stiffness, while the bond width and concrete strength are maintained at 100 mm and 50 MPa, respectively. The FRP sheet stiffness, Et, values range from 25 to 125 GPa·mm, which represent the common values used in the literature and that represent a range of one to five plies of FRP sheets having a ply thickness of 0.38 mm and a modulus of elasticity of 65789 MPa. Figure 6-6 shows the FRP sheet bond length plotted against the ultimate load at the various Et values listed in Table 6-3. The figure shows that the ultimate load increases as the sheet stiffness increases. Estimating the effective bond lengths for the different Et values from Figure 6-6 and plotting them against the sheet stiffness in Figure 6-7 shows that the effective bond 163 length increases as the sheet stiffness increases. This could be attributed to the decrease in the ductility of the FRP sheet with the increase in its stiffness, which delays the occurrence of debonding. This delay also increases the capacity of the joint, where a larger effective bond area resists the applied load. The expression "joint" used here refers to the assembly of concrete, adhesive, and FRP sheet. The figure also shows a constant effective bond length beyond the sheet stiffness of 100 GPa·mm. Maeda et al. (1997) who studied the effect of the number of plies on the bond behaviour using pull-apart specimens, stated that the effective bond length decreases as the sheet stiffness increases, which is contrary to findings of this investigation.

Table 6-4 lists the models investigating the bond length effect at various concrete strength values. The table shows that the only variables were the FRP sheet bond length and the concrete strength, while the FRP sheet bond width and stiffness are maintained at 100 mm of 25 GPa·mm, respectively. The value of the concrete modulus of elasticity,  $E_c$ , was calculated in accordance with the CSA Standard A23.3-94, (1995):

$$E_c = 4500 \sqrt{f_c}$$
 (6-2)

in which  $f_c$  is the uniaxial concrete compressive strength. The concrete strength values range from 20 to 60 MPa, chosen from practical value ranges for existing and new reinforced concrete structures. The concrete tensile strength was taken equal to  $0.53\sqrt{f_c}$ . The overall load-displacement responses for models with L=200 mm, listed in Table 6-4, are plotted in Figure 6-8. Figure 6-9 shows the FRP sheet bond length plotted against the ultimate load diagrams at the various concrete strength values listed in Table 6-4. The figure shows that the ultimate load increases as the bond length increases up to 200 mm, beyond which no increase in the ultimate load is observed. The figure also shows that as  $f_c$  increases, the ultimate load increases and that the curves for different  $f_c$  values are almost parallel, which results in similar effective bond lengths of 200 mm for all  $f_c$  values. Further investigation into the

previous observation is detailed in Section 6.5, which studies  $f'_c$  effect on the bond behaviour.

A plot of the average bond strength against the FRP sheet bond length for  $f'_c$  values ranging from 20 to 60 MPa for the same models listed in Table 6-4 is shown in Figure 6-10. The figure shows the relationship converging and the average bond strength stabilizing at longer bond lengths. This indicates that the concrete strength does not have a significant effect on the bond behaviour of longer bond lengths. This could be attributed to the debonding type failure simulated in the models, which is dependent on the adhesive strength represented in the interface shear and normal stresses between the FRP sheet and the concrete rather than the concrete strength. This explanation was also stated by De Lorenzis et al. (2000) who studied  $f'_c$  effect on bond behaviour using beam specimens. Assuming an effective bond length of 200mm in calculating the average bond strength values beyond the effective bond lengths are constant and range between 1.1 and 1.3 MPa for  $f'_c$  values of 20 and 60 MPa, respectively.

From the previous discussion it is concluded that increasing the bond length increases the capacity and bond strength of the joint. There also exists an effective bond length beyond wish no increase in the capacity is gained, which is similar to the experimental conclusion. The effective bond length increases as the sheet bond width increases. Increasing the sheet stiffness also increases the effective bond length up to a certain stiffness value of 100 GPa·mm beyond which no increase in the effective bond length is gained. The concrete strength does not appear to have a significant effect on the effective bond length.

## 6.3 FRP Sheet Bond Width

# 6.3.1 General

The following presents the effect of the FRP sheet bond width on the bond behaviour with concrete studied using the 2D modified push-apart model as detailed in Chapter 5. As the models developed in Chapter 5 were able to capture the width effect in an average sense, they were seen as sufficient for this investigation. Table 6-2 lists the models and their variables used in investigating the bond width effect. The variables are the FRP sheet bond width and length. The bond width values ranged from 25 to 200 mm, which was 12.5% and 100% of the concrete width, respectively. The bond lengths ranged from 50 to 200 mm, which covered a wide range of development lengths practically used.

Figure 6-12 shows the FRP sheet bond width plotted against the ultimate load relationships for bond length values ranging from 50 to 200 mm while maintaining a constant sheet stiffness and concrete strength of 25 GPa mm and 50 MPa, respectively. The figure shows that the ultimate load increases as the bond width increases. At bond width values less than 100 mm the percentage increase in ultimate load is almost similar to the percentage increase in the bond width while for bond width values larger than 100 mm the percentage increase in load is significantly lower than the percentage increase in bond width. This behaviour is apparent from the change in slopes at the bond width of 100 mm in Figure 6-12. Increasing the bond width by 100% from 50 to 100 mm increases the ultimate load by 94% while increasing the bond width from 100 to 200 mm increases the ultimate load by 51% on average for bond lengths ranging from 50 to 200 mm. The ultimate load per unit width plotted against the bond length is shown in Figure 6-13. The figure shows that there is no significant difference in  $P_u/w$  for bond widths below 100 mm, but beyond this value P<sub>u</sub>/w decreases as the bond width increases. This indicates a variation in the stress distribution across the FRP sheet width as the width increases. Figure 6-13 also shows that the P<sub>u</sub>/w value increase as the bond length increases, but stabilizes as it reaches lengths over 150 mm.

The previous discussion indicates a reduction in the average bond strength for wider FRP sheets, which is directly proportional to the ultimate load and inversely proportional to the sheet bond width. The FRP sheet bond width plotted against the average bond strength relationship for bond length values ranging from 50 to 200 mm and using constant FRP sheet stiffness and bond width values of 25 GPa·mm and 100 mm, respectively, is shown in Figure 6-14. The average bond strength in the figure was calculated by equation (6-1). The figure shows that the average bond strength decreases as the bond width exceeds 100 mm while there is no significant change in its value at bond widths less than 100 mm.

The first portion of the previous conclusion is consistent with the literature (Ueda et al., 1999) and with the S-series experimental results detailed in Section 4.5, which shows the average bond strength decreasing as the bond length increases from 100 to 150 mm. The second portion agrees with literature research done on narrow bond widths, which indicates that the bond width has no effect on the bond behaviour (Brosen and Van Germert, 1999, and De Lorenzis et al., 2000). On the other hand, the conclusion does not match the results of the P-series, also detailed in Section 4.5, where the average bond strength is seen to decrease as the width increased even for width values lower than 100 mm. The following chapter will investigate if the type of bond test used is the cause of this variation.

Figure 6-15 shows the average bond strength based on an estimated effective bond length of 110 mm. The figure show that using the average effective bond length stabilizes the average bond strength beyond the average effective bond length. This indicates that the bond width is important only when the bond length is less than the effective bond length.

The previous discussion concludes that the average bond strength decreases as the bond width exceeds 100 mm while no significant change occurs at bond width values below 100 mm. This explains the contradictory results in the literature

regarding the influence of the bond width on the bond behaviour, presented in Section 2.7.2. The same conclusion implies that narrow sheets over-estimate the average bond strength. Practical strengthening schemes, however, tend to use wider FRP sheets. This indicates the importance of studying the behaviour of wider sheets on the bond behaviour.

## 6.3.2 The Effect of FRP Sheet Width to Concrete Width Ratio

Both the 2D and 3D modified push-apart models discussed in Chapter 5 were used to investigate the ratio of the sheet bond width to the concrete width on the bond behaviour. The 3D model was only used to investigate the strain distribution across the FRP sheet width, where it did not have the debonding capability but was seen as sufficient to investigate the strain distribution across the FRP sheet width. On the other hand, the 2D model was shown in Chapter 5 to be able to capture the bond behaviour in an average sense as the bond width varied.

The distance from the edge of the bonded FRP sheet to the specimen edge is analogous to half the spacing between FRP strips in shear strengthening of RC girders or flexural strengthening of RC slabs in practical applications. The ratio of the width of the FRP sheet to the width of the concrete specimen was taken into consideration in the literature in some proposed models predicting the bond strength of bonded FRP sheets to concrete as detailed in Section 2.8. The strain distribution across the FRP sheet width was investigated in the experimental tests detailed in Chapters 3 and 4. The experimental results discussed in Section 4.3.2 showed the strain distribution to vary across the sheet width. This experimental observation pointed out the need to study the strain distribution across the FRP sheet width in more detail using numerical models.

Table 6-5 lists the variables used in this investigation. The variables were the concrete width and FRP sheet bond length, while all the other parameters were maintained constant. The concrete width values considered were 200, 300, and 400

mm, while FRP sheet bond widths were maintained at 200 mm in order to investigate whether the ratio of the FRP sheet to concrete width or the actual value of the concrete width was the effective factor influencing the bond behaviour. FRP sheet lengths of 50 and 150 mm were used in order to study whether the FRP sheet length had any influence on the behaviour.

The load-displacement responses of 2D models with concrete width values of 200, 300, and 400 mm are presented in Figure 6-16 and Figure 6-17 for FRP sheet bond lengths of 50 and 150 mm, respectively. The figures show that as the concrete width increases, both the ultimate load and stiffness of the joint increases, regardless of the bond length value. This shows that the concrete width itself affects the bond behaviour rather than the FRP sheet to concrete width ratio.

Figure 6-18 shows the contour lines of the strain in the main direction of the FRP sheet, E11, in 3D models with sheet width to concrete width ratios,  $w/w_c$ , of 0.7 and 1.0. It is clear from the figure that the strain values are higher at the FRP sheet edge than at its centre for the  $w/w_c$  of 0.7 case, while the strain is almost uniform across the FRP sheet width in the case of the FRP sheet covering the whole concrete width ( $w/w_c$  of 1.0). Figure 6-18b also shows that the strain is distributed in the concrete beyond the bonded width of the FRP sheet, which could explain the reason behind the increase in ultimate load with wider concrete widths.

The strain distribution, E11, across half the concrete width at the crack location in 3D models with FRP sheet bond widths of 100 and 200 mm are plotted in Figure 6-19 at 15 kN and ultimate load. The figure shows that for the model with FRP sheet width of 100 mm, the strain distribution in the concrete is higher underneath the FRP sheets but continues around 30 mm beyond the sheet width at ultimate load. The strain gradient is also high in this case, while in the case of the FRP sheet covering the whole concrete width, the strain distribution in the concrete is almost constant at 15 kN and significantly more uniform at ultimate load. The distribution of the stress component S11 in the direction of the load across half the concrete width at the crack location for 3D models with FRP sheet widths of 100 and 200 mm are shown in Figure 6-20. The figure shows that the stress distribution in the concrete underneath the 100 mm wide FRP sheet is not uniform and the stresses are compressive, while the stresses in the concrete beyond the FRP sheet are in tension. The compression in the concrete balances the applied tensile stresses in the FRP sheets to satisfy equilibrium, while the tension stresses in the concrete beyond the FRP sheet. The values of the stresses in Figure 6-20 indicate that the concrete is uncracked. The model with the FRP sheet covering the whole concrete width has a more uniform stress distribution in the concrete.

Figure 6-21 shows the strain distributions at 15 kN across the FRP sheet width at the crack location for 3D models with sheet widths ranging from 60 to 200 mm. The figure shows that the strain at the edge of the sheet is higher than that at its centre. This is similar to the experimental observations in Section 4.3.2. At the same load of 15 kN, the strain difference between the edge and centre of the FRP sheet increases as the bond width decreases. The strain values are also shown to increase as the width decreases. The same behaviour is observed for the strain difference and strain values at loads 50, 70, and 100% of the ultimate load, as shown in Figure 6-22, Figure 6-23, and Figure 6-24, respectively. The figures also show the strain values increasing as the applied load increases.

Figure 6-25 shows the relationship between the FRP sheet width and the difference in strain between the edge and centre of the sheet,  $\Delta \varepsilon$ , at load levels, P/P<sub>u</sub>, of 0.5, 0.7, and 1.0 and also at a load of 15 kN. The figure shows that  $\Delta \varepsilon$  increases as P/P<sub>u</sub> increases and the 15 kN curve lies between the P/P<sub>u</sub> of 0.5 and 1.0 curves, where the 15 kN load had a different load level value for each sheet width. Figure 6-25 also confirms that  $\Delta \varepsilon$  increases as the bond width decreases. There are no significant differences in  $\Delta \varepsilon$  observed when the sheet covers the concrete width (i.e. w = 200

mm), except for small  $\Delta \varepsilon$  at ultimate load. This was also observed in the experimental tests detailed in Section 4.3.2.

Further investigation was performed into whether the previous behaviour is related to the actual sheet width or, instead, to the sheet-to-concrete width ratio. The strain in the FRP sheets plotted against the ratio of the distance from the specimen centre to half the concrete width for FRP sheet widths of 100, 200, and 400 mm, concrete widths of 200 and 400 mm, and all having bond lengths of 100 mm are shown in Figure 6-26. The figure shows that the strain values and distributions across the width for models with the same sheet widths are more related to each other than the corresponding models with the same w/w<sub>c</sub> values. This indicates that the strain values and distributions across the width are related to the actual sheet width rather than the relative sheet to concrete width. This conclusion is important when specifying design guidelines for Reinforced Concrete (RC) beam shear strengthening and RC slab flexure strengthening. It indicates that actual width values should be used rather than relating the distance between the FRP strips to the strip widths. On the other hand, Figure 6-26 shows that the strain difference between the edge and centre of the FRP sheet is higher with wider concrete models as shown for the models with FRP sheet width of 100 mm and  $w/w_c$  values of 0.25 and 0.5.

The previous discussions conclude that the strains at the edge of the FRP sheets are higher than those at its centre and the difference between them increases as the FRP sheet bond width decreases and as the concrete width increases. This indicates that the investigations carried out with strain measurements only at the centre of the FRP sheet underestimate the strain values. The strain values are also shown to increase as the width decreases. The actual values of the concrete width and FRP sheet bond width are found to affect the bond behaviour rather than the relative FRP sheet to concrete width. This further indicates that the concrete specimen width needs to be defined and implemented.

171

# 6.4 FRP Sheet Stiffness

The literature shows that the FRP sheet stiffness is one of the important variables in determining the bond strength. The product of the FRP sheet modulus of elasticity, E, and its thickness, t, defines its stiffness, Et. A study was performed on models with different E and t values but with the same  $E \times t$  value of 100 and 50 GPa·mm as shown in Table 6-7. The results showed no change in behaviour between models with similar  $E \times t$  value. In this investigation, the FRP sheet stiffness was changed by changing the sheet thickness, t, to values of 0.38, 0.76, 1.14, 1.52, and 1.9 mm, while maintaining E constant at 65,789 MPa. These values of E and t were taken in order to produce Et values of 25, 50, 75, 100, 125 GPa·mm, respectively.

The effect of the FRP sheet stiffness, Et, on the bond behaviour was studied using the 2D modified push-apart model. The variables used in this investigation are listed in Table 6-3. The FRP sheet stiffness investigated ranged from 25 to 125 GPa·mm while the bond width was sustained at 100 mm. The ultimate load plotted against the FRP sheet stiffness at different bond length values is shown in Figure 6-27. The figure shows that the ultimate load increases as the stiffness increases but the effect is more significant at longer bond lengths. As previously stated this could be attributed to reduction in the elongation of the FRP sheet as the stiffness increases, which in turn causes less strain at the interface. This conclusion agrees with previous findings by other researchers who stated that increasing the sheet stiffness increases the capacity of the joint, as presented in Section 2.7.3. It is also observed from Figure 6-27 that the rate at which the ultimate load increases falls as the stiffness increases.

The average bond strength plotted against the FRP sheet stiffness at different bond length values ranging from 50 to 200 mm is shown in Figure 6-28. The figure shows that there is a stiffness value for every bond length beyond which no change in the average bond strength was observed. This limiting stiffness value will be called the effective stiffness. The effective FRP sheet stiffness estimated from Figure 6-28 and indicated by dotted vertical lines is plotted against the bond length in Figure 6-29. The figure shows that the effective FRP sheet stiffness increases as the bond length

increases. This can be attributed to the limited average bond strength each bond area can resist. The average bond stress will increase as the sheet stiffness increases until the sheet stiffness reaches a value that applies the maximum average bond stress to the bond area. Any increase in the sheet stiffness beyond that value will not increase the average bond strength and this limiting value could be considered the effective sheet stiffness value.

From the previous discussion it is concluded that the ultimate load increases as the FRP sheet stiffness increases. The average bond strength also increases as the FRP sheet stiffness increases but up to a certain limit. This limit is defined as the effective sheet stiffness beyond which no increase in the average bond strength is gained. The effective sheet stiffness changed with the bond lengths. This conclusion has a practical implication of an optimum FRP sheet stiffness or number of plies for every bond length.

# 6.5 Concrete Strength

The concrete strength,  $f'_c$ , effect on the bond behaviour was studied using  $f'_c$  values ranging from 20 to 60 MPa and bond lengths ranging from 50 to 200 mm, while the bond width was sustained at 100 mm. The bond length range was chosen in order to capture the influence of the effective bond length on the  $f'_c$  effect. The models used to study this effect are listed in Table 6-4. The overall load-displacement responses, measured at the point of load application, for all models listed in Table 6-4 were plotted, similar to that shown in Figure 6-8, which were used in further analysis. Figure 6-30 shows the concrete compressive strength plotted against the ultimate load at different bond length values. The figure shows that the ultimate load increases as  $f'_c$  increases, with the effect decreasing at longer bond lengths. The strain distributions along the sheet length of 100 mm at ultimate load plotted in Figure 6-31 also show that the strains increase as  $f'_c$  increases. There is no significant increase in the ultimate load or the strain values for  $f'_c$  values higher than 40 MPa.

The average bond strength plotted against the concrete compressive strength at various bond length values ranging from 50 to 200 mm and a constant bond width of 100 mm is shown in Figure 6-32. The figure shows that the average bond strength increases as the concrete strength increases. The increase in the average bond strength is more noticeable at shorter bond lengths, where an increase in the concrete strength by 200% from 20 to 60 MPa increases the average bond strength for bond lengths of 50 and 200 mm by 41.2 and 18%, respectively.

Figure 6-8 shows the load-deflection response for models with concrete strengths ranging from 20 to 60 MPa and a constant bond length of 100 mm. The figure shows that the overall load-deflection response is stiffer for higher concrete strengths. To further investigate whether the test method had an effect on the results, the 2D pull-apart models were applied for the same variables and the same material properties as the modified push-apart models listed in Table 6-4. The overall loaddeflection responses for pull-apart models with  $f_c'$  values ranging from 20 to 60 MPa, similar to those shown in Figure 6-33, were used to plot the ultimate load against the  $f_c'$  values. Figure 6-34 shows the ultimate load from the pull-apart models plotted against  $f'_c$  for bond lengths of 70, 100, and 200 mm. The figure shows that there is no significant difference in the ultimate load between the different bond lengths. The differences in ultimate load for the various  $f'_c$  values were also smaller than those from the modified push-apart model. The strain distributions along a sheet length of 100 mm at ultimate load for the pull-apart models are shown in Figure 6-35. The figure shows that for the pull-apart model the strains also increase as  $f'_c$  increases but at  $f_c'$  values of 40, 50, and 60 MPa the maximum strain is almost uniformly distributed along the length.

The previous discussion indicates that increasing  $f'_c$  stiffens the overall behaviour of the specimen and in return affects the capacity of the joint. The amount of influence will depend on the geometry and loading of the specimen. This shows

the importance of studying the effect of the test method on the bond behaviour. A detailed investigation on the conventional bond test methods and the relationship between their results is presented in the following chapter.

	Width	w <sub>c</sub> (mm)	200
Concrete	Modulus of Elasticity,	E (MPa)	31820
	Poisson's ratio,	υ	0.15
	Compressive yield strength,	fy (MPa)	28
	Ultimate comp. Strength,	f <sub>c</sub> ' (MPa)	50
	Thickness	t (mm)	0.38
CFRP Sheet	Modulus of Elasticity,	E (MPa)	65789
	Poisson's ratio,	υ	0.26
	Ultimate tensile Strength,	fu (MPa)	2500

Model	L	W	eff. τ <sub>b</sub>	fc	Et	Pu	τ <sub>b</sub>
No.	(mm)	(mm)	(MPa)	(MPa)	(GPa.mm)	(kN)	(MPa)
1	50	25	4.36	50	25	5.45	4.36
2	50	50	4.32	50	25	10.80	4.32
3	50	70	4.29	50	25	15.00	4.29
4	50	100	4.08	50	25	20.40	4.08
5	50	150	3.32	50	25	24.90	3.32
6	50	200	2.88	50	25	28.80	2.88
7	70	25	3.31	50	25	5.80	3.31
8	70	50	3.29	50	25	11.50	3.29
9	70	70	3.27	50	25	16.00	3.27
10	70	100	3.24	50	25	22.70	3.24
11	70	150	2.67	50	25	28.00	2.67
12	70	200	2.36	50	25	33.00	2.36
13	100	25	3.43	50	25	6.00	2.40
14	100	50	3.49	50	25	12.20	2.44
15	100	70	3.47	50	25	17.00	2.43
16	100	100	3.43	50	25	24.00	2.40
17	100	150	2.93	50	25	30.80	2.05
18	100	200	2.57	50	25	36.00	1.80
19	130	25	3.49	50	25	6.10	1.88
20	130	50	3.66	50	25	12.80	1.97
21	130	70	3.59	50	25	17.60	1.93
22	130	100	3.53	50	25	24.70	1.90
23	130	150	3.10	50	25	32.50	1.67
24	130	200	2.76	50	25	38.70	1.49
25	150	25	3.60	50	25	6.30	1.68
26	150	50	3.71	50	25	13.00	1.73
27	150	70	3.65	50	25	17.90	1.70
28	150	100	3.57	50	25	25.00	1.67
29	150	150	3.14	50	25	33.00	1.47
30	150	200	2.85	50	25	39.95	1.33
31	200	25	3.60	50	25	6.30	1.26
.32	200	50	3.71	50	25	13.00	1.30
33	200	70	3.65	50	25	17.90	1.28
34	200	100	3.57	50	25	25.00	1.25
35	200	150	3.14	50	25	33.00	1.10
36	200	200	2.85	50	25	39.95	1.00

Table 6-2 Models with FRP sheet bond lengths varying with bond width

Model	L	W	ťc	Et	Pu	τ <sub>b</sub>
No.	(mm)	(mm)	(MPa)	(GPa.mm)	(kN)	(MPa)
37	50	100	50	25	20.40	4.08
38	50	100	50	50	24.00	4.80
39	50	100	50	75	25.50	5.10
40	50	100	50	100	26.30	5.26
41	50	100	50	125	26.50	5.30
42	70	100	50	25	22.70	3.24
43	70	100	50	50	26.50	3.79
44	70	100	50	75	28.00	4.00
45	70	100	50	100	28.90	4.13
46	70	100	50	125	29.60	4.23
47	100	100	50	25	24.00	2.40
48	100	100	50	50	29.20	2.92
49	100	100	50	75	30.40	3.04
50	100	100	50	100	31.40	3.14
51	100	100	50	125	32.20	3.22
52	150	100	50	25	25.00	1.67
53	150	100	50	50	31.67	2.11
54	150	100	50	75	34.00	2.27
55	150	100	50	100	35.50	2.37
56	150	100	50	125	37.00	2.47
57	200	100	50	25	25.00	1.25
58	200	100	50	50	32.50	1.63
59	200	100	50	75	36.57	1.83
60	200	100	50	100	38.95	1.95
61	200	100	50	125	40.52	2.03
62	250	100	50	25	24.95	1.00
63	250	100	50	50	32.80	1.31
64	250	100	50	75	37.80	1.51
65	250	100	50	100	40.70	1.63
66	250	100	50	125	43.08	1.72
67	300	100	50	25	24.95	0.83
68	300	100	50	50	32.90	1.10
69	300	100	50	75	37.82	1.26
70	300	100	50	100	41.27	1.38
71	300	100	50	125	43.60	1.45

Table 6-3 Models with bond lengths varying with sheet stiffness, Et

Model	. Lue	W	ťc	E	Ēt	Pu	τ <sub>b</sub>
No.	(mm)	(mm)	(MPa)	(MPa)	(GPa.mm)	(kN)	(MPa)
72	50	100	20	20125	25	15.30	3.06
73	50	100	30	24648	25	16.62	3.32
74	50	100	40	28460	25	18.30	3.66
75	50	100	50	31820	25	20.40	4.08
76	50	100	60	34857	25	21.59	4.32
77	70	100	20	20125	25	18.00	2.57
78	70	100	30	24648	25	19.60	2.80
79	70	100	40	28460	25	21.20	3.03
80	70	100	50	31820	25	22.70	3.24
81	70	100	60	34857	25	23.20	3.31
82	100	100	20	20125	25	20.10	2.01
83	100	100	30	24648	25	21.80	2.18
84	100	100	40	28460	25	22.90	2.29
85	100	100	50	31820	25	24.00	2.40
86	100	100	60	34857	25	24.50	2.45
87	150	100	20	20125	25	21.75	1.45
88	150	100	30	24648	25	23.00	1.53
89	150	100	40	28460	25	23.80	1.59
90	150	100	50	31820	25	25.00	1.67
91	150	100	60	34857	25	25.50	1.70
92	200	100	20	20125	25	22.10	1.11
93	200	100	30	24648	25	23.50	1.18
94	200	100	40	28460	25	24.30	1.22
95	200	100	50	31820	25	25.20	1.26
96	200	100	60	34857	25	25.90	1.30
97	250	100	20	20125	25	22.05	0.88
98	250	100	30	24648	25	23.60	0.94
99	250	100	40	28460	25	24.50	0.98
100	250	100	50	31820	25	25.30	1.01
101	250	100	60	34857	25	25.98	1.04
102	300	100	20	20125	25	22.06	0.74
103	300	100	30	24648	25	23.60	0.79
104	300	100	40	28460	25	24.50	0.82
105	300	100	50	31820	25	25.30	0.84
106	300	100	60	34857	25	26.00	0.87

 Table 6-4 Models with bond lengths varying with concrete strength

model #	W	L	W <sub>c</sub>	w/w <sub>c</sub>	Pu	f <sub>c</sub> '	Et
	(mm)	(mm)	(mm)		(kN)	(MPa)	(GPa.mm)
107	200	50	400	0.50	35.2	50	25
108	200	50	300	0.75	34.5	50	25
109	200	50	200	1.00	28.6	50	25
110	200	150	400	0.50	50.0	50	25
111	200	150	300	0.75	46.9	50	25
112	200	150	200	1.00	40.0	50	25

Table 6-5 Models with variable concrete width and FRP bond length

Table 6-6 3D models with variable concrete width and FRP bond widths

model #	W	L	wc	w/w <sub>c</sub>	f <sub>c</sub> '	Et
	(mm)	(mm)	(mm)		(MPa)	(GPa.mm)
113	60	100	200	0.30	50	25
114	100	100	200	0.50	50	25
115	140	100	200	0.70	50	25
116	180	100	200	0.90	50	25
117	200	100	200	1.00	50	25
118	100	100	400	0.25	50	25
119	200	100	400	0.50	50	25
120	400	100	400	1.00	50	25

Table 6-7 2D models with variable E and t values but with the same Et product

model #	W	L.,	f <sub>c</sub> '		t	Et
	(mm)	(mm)	(MPa)	(MPa)	(mm)	(GPa.mm)
121	100	150	50	263158	0.38	100
122	100	150	50	65789	1.52	100
123	100	150	50	131579	0.38	50
124	100	150	50	65789	0.76	50



**Displacement (mm)** 

Figure 6-1 Load-displacement response for models with L = 150 mm



Figure 6-2 Bond length-ultimate load at different w values



Figure 6-3 Effective length and Bond width relationship



Figure 6-4 Average bond strength and bond length for different w values



Figure 6-5 Average effective bond strength for different w values



Figure 6-6 Bond length and ultimate load for different Et values






Figure 6-8 Load-displacement response for modified push-apart (L = 200 mm)







Figure 6-10 Bond length and Average bond strength for different f'c values



Figure 6-11 Bond length and Average effective bond strength for different f'<sub>c</sub> values



Figure 6-12 Bond width and ultimate load for different L values



Figure 6-13 Ultimate load per width and bond length for various bond widths



Figure 6-14 Bond width and bond strength for different L values



Figure 6-15 Bond width and bond strength for different L values ( $L_e = 110 \text{ mm}$ )



Figure 6-16 Load-Displacement response for different  $w_c$  values (w = 200 and L = 50 mm)



Figure 6-17 Load-Displacement response for different  $w_c$  values (w = 200 and L = 150 mm)









Figure 6-19 Strain distribution across the concrete width







Figure 6-21 Strain distribution across different w values at crack location, 15 kN



Figure 6-22 Strain distribution across width at crack location,  $P/P_u = 0.5$ 

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.



Figure 6-23 Strain distribution across width at crack location,  $P/P_u = 0.7$ 



Figure 6-24 Strain distribution across width at crack location,  $P/P_u = 1.0$ 







Figure 6-26 Strain distribution across FRP sheet width at 15 kN

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.



Figure 6-27 FRP stiffness vs. ultimate load for various bond lengths



Figure 6-28 FRP stiffness vs. average bond strength for various bond lengths



Figure 6-29 Effective FRP stiffness vs. bond length relationship



Figure 6-30 Concrete strength vs. ultimate load for various bond lengths



Figure 6-31 Strain distribution for various  $\,f_c^{\,\prime}\,$  values



Figure 6-32 Concrete strength vs. ultimate load for various bond lengths



Figure 6-33 Load-displacement response for pull-apart models (L = 200 mm)



Figure 6-34 Concrete strength vs. ultimate load for push-apart model





Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

## 7 COMPARISON BETWEEN BOND TEST METHODS

#### 7.1 Introduction and Overview

Bond stress curves proposed in the literature describe well the data from which they were defined. However, there are large discrepancy and scattering between these curves. One of the reasons for this discrepancy is the different bond test methods used to conduct the investigations. There are several different test arrangements conventionally used for FRP bond investigations. A brief description of the conventional bond test methods were presented and grouped into categories based on the method of load application in Section 2.6. These include the "Direct pull" method, the "Pull-apart" method, and the "Push-apart" method. The "Modified push-apart" method was also presented in Chapter 3.

There is a need to assess the performance of these methods comparatively in order to develop conversion standards. There is also a need to develop a rational and simplified standard test method for FRP sheets bonded to concrete.

A comparison between bond test methods has been conducted in the form of a parametric study of the effects of the support conditions and loading configurations on the behaviour. In this chapter the different bond test methods, grouped into categories depending on the loading arrangement, are analyzed using 2D nonlinear finite element models. The analysis includes the stress and strain distributions in the concrete and FRP sheet, respectively. The overall load-displacement behaviour and failure mode of each specimen type were also analysed.

In order to initiate the debonding mechanism between the FRP sheet and the concrete surfaces in ABAQUS, the first node in the bonded surfaces is required to be initially debonded. Therefore, the FRP sheet was unbonded from the concrete at the loaded edge node. This prevented the capture of the concrete edge failure reported in the modified push-apart experimental tests in Chapter 4 and stated by Bizindavyi and Neale (1999).

The bond tests investigated in this chapter are those that include the direct pull, indirect pull-apart, push-apart, and the modified push-apart tests. The material properties used in the models are listed in Table 7-1. A relationship between the test methods is proposed based on the numerical study. The relationship is correlated to experimental results from the current study and studies from the literature. Two bond tests are proposed as standard test methods based on the loading apparatus available and comprehensibility, drawn from the analysis in this chapter.

### 7.2 Direct Pull Bond Test

The direct pull bond test specimen description and the load setup are shown in Figure 7-1a. A brief description of the test was presented in Section 2.6.3.1. The direct pull test is one of the simple bond tests, where an easy to cast concrete block can be used for several tests. The method is conducted using a testing machine, which means it could not be performed on site. The method also restricts the FRP sheet width to a maximum of 50 mm due to gripping problems that could cause slipping of the FRP sheet during testing and premature failure of the FRP sheet. Bizindavyi and Neale (1999) stated that to reduce the risk of sliding and premature failure of the FRP sheets within the grips, the FRP sheets were limited to two plies with bond width of 25.4 mm. Bonding the FRP sheet to the concrete block and impregnating the free unbonded portion of the sheet while maintaining the sheet straight is also difficult. The concrete block is in compression during the test while the FRP sheet is in tension, which is not the stress state in flexure and shear. Special precautions must be taken during testing to ensure that no eccentricity exists in the FRP sheet.

Due to the practical reasons discussed, the direct pull test is limited to one-layer FRP sheet width of 50 mm or two layers 25 mm wide. However, for the sake of comparison, FRP sheet widths ranging from 25 to 150 mm will be investigated. As the direct pull test specimen has only one FRP sheet bonded to a block of concrete, the whole specimen was modeled. The concrete and FRP sheet element meshes were taken as  $10 \times 10$  mm four-node plain stress solid elements with reduced integration. The 2D

direct pull model mesh, deformed shape, loading, and boundary conditions are shown in Figure 7-1b. The direct pull model represents the perfect case with no eccentricity in loading the FRP sheet.

The length of the unbonded portion of the FRP sheet, which is the sheet length between the concrete face and the machine grip, was investigated. Figure 7-2 shows the load-displacement response for direct pull models with unbonded sheet lengths of 10, 50, 100 mm and constant bond lengths and widths of 150 mm each. The figure shows that the load-displacement response was stiffer as the unbonded length decreased. This is attributed to the fact that the concrete block was fixed, causing the displacement to be only from the extension of the unbonded FRP sheet including the debonded portions of the sheet. This caused higher displacement to occur as the unbonded sheet lengths increased. Figure 7-2 also showed no significant difference in ultimate load values for the different unbonded lengths. In practice the unbonded FRP sheet length must allow for the bearing steel plate and clearance for aligning the sheet into the machine grip. This length is estimated to be at least 50 mm. Based on practical test setup implementation, the direct pull model used in the comparison between the test methods will have an unbonded FRP sheet length of 50 mm. This behaviour cannot be confirmed from the literature, as there are no studies on the effect of the unbonded sheet length.

Figure 7-3 shows the contour lines for the principal stresses at various load increments in the direct pull model. The figure shows the development of the principal stresses, where a compression strut is formed between the bearing steel plate and the loaded (bonded) portion of the FRP sheet. The figure also shows that as the load increases, the compression strut shifts along the sheet. This is attributed to the debonding propagation of the FRP sheet from the concrete, which causes the bonded area to shift away from the loaded end.

The effect of the compression field in the concrete adjacent to the bearing steel plate on the FRP sheet bond behaviour was investigated. The investigation focused on

the distance between the end of the bearing steel plate and the FRP sheet. That distance ranged from 10 to 50 mm. It is found from the numerical investigation that there is no significant effect on the load-displacement behaviour or the strain distribution in the FRP sheet due to the variation of that distance. Therefore, the distance between the bearing steel plate and the FRP sheet used in the comparison between the test methods was arbitrarily chosen to be 30 mm.

Figure 7-4 shows the load-displacement behaviour for direct pull models with FRP sheet widths of 25, 50 100, and 150 mm. The figure shows that increasing the bond width increases the ultimate load, but the ultimate load per unit width is constant at 0.2 kN/mm for all bond widths. This indicates that the maximum average bond strength across the FRP sheet width is similar for all bond widths. Figure 7-5 shows the load level,  $P/P_u$ , plotted against the displacement for models with FRP sheet widths of 25, 50, 100, and 150 mm. The figure shows that there is no significant difference between the responses with the change in bond width.

Figure 7-6 and Figure 7-7 show the strain distribution along the FRP sheet length for bond widths of 25 and 150 mm, respectively. The figures show that the strain increases as the load increases, but at the ultimate load there is a limiting strain value that corresponds to the model ultimate interface normal and shear stresses. The figures also show that for the same load levels, the strain in the model with 150 mm bond width is higher than that in the model with 25 mm bond width. The transfer length, the distance from the crack location to the point where the strain is negligibly small, is also longer in the 150 mm bond width model. At ultimate load both the 25 and 150 mm bond width models have almost the same ultimate strain plateau value, but with the 25 mm model covering more of the bond length and having higher strain values beyond the plateau portion.

## 7.3 Pull-Apart Test

The pull-apart bond test specimen is illustrated in Figure 7-8a. The specimen description and loading setup were also presented in Sections 2.6.3.2 and 3.3. The pull-

apart test is an indirect bond test that requires a testing machine or a hydraulic jack with a special loading frame. The specimen is an easy to cast concrete prism but special care must be taken to ensure the alignment of the steel rebars and to secure the location of the crack initiator during casting. The number of FRP sheet plies and dimensions are limited by the bond strength of the steel rebar. Aligning the specimen in the testing machine and preventing any inclination of the specimen during testing is very important, where eccentricity has a significant effect on the results.

Although some pull-apart tests have been performed with specimens having a gap between the concrete prisms, the pull-apart model used in this investigation assumes no gap between the prisms. As no anchor sheets are applied to the tested sheets when the specimens are used to determine the bond strength or the development length, one quarter of the pull-apart specimen was modeled, making use of symmetry. The 2D pullapart numerical model is as described in Chapter 5 with the exception that one quarter of the specimen is modeled rather than one half. The mesh, deformed shape, loading, and boundary conditions of the pull-apart model are shown in Figure 7-8b.

The pull-apart model represents the perfect case with no load eccentricity, where the steel rebars are assumed perfectly aligned and have perfect bond with the concrete. The failure mechanism of the pull-apart specimens in the experimental investigation, detailed in Section 4.2.2, showed that failure occurred on one side of the specimen, causing eccentricity in loading during testing. This behaviour was not represented in the numerical model.

A numerical investigation of the rebar free length effect, which is the length between the machine grip and the concrete specimen, was conducted using lengths of 10, 50, and 100 mm. The investigation showed that the free length of the rebar did not affect the model output. This is because the applied stresses were significantly less than the rebar yield stress and also because perfect alignment was assumed. The free length of the steel rebar in the model used in the comparison between the test methods was taken to be 50 mm. A numerical investigation studying the specimen width separating

the bonding concrete faces also showed no significant effect on the numerical results. The concrete prism dimension between the bonding surfaces was taken as 100 mm in the model used for the comparison between the test methods.

Figure 7-9 shows the applied load plotted against the displacement for pull-apart models with FRP sheet widths of 25, 50, 100, and 150 mm. The figure shows that increasing the bond width increases the ultimate load, but the ultimate load per unit width is constant at 0.21 kN/mm for all bond widths. Figure 7-10 shows the load level,  $P/P_u$ , plotted against the displacement for models with FRP sheet widths of 25, 50, 100, and 150 mm. The figure shows that at the same displacement, the load level is higher for smaller bond widths but there is no significant difference between the responses at ultimate load.

Figure 7-11 and Figure 7-12 show the strain distribution along the FRP sheet length at various load levels for models with bond widths of 25 and 150 mm, respectively. The figures show the strain increasing as the load increases, but with no limiting strain plateau. As the value of the ultimate strain is related to the combination of both the interface shear and normal stresses, an inclination between the FRP sheet and the concrete will result in changing the ultimate shear value. The FRP sheet in the pull-apart test debonds and inclines away from the concrete as load progresses, which could explain their descending strain distribution. The figures also show that there is no significant difference between the strain values or the transfer lengths at the same load levels for the bond widths of 25 and 150 mm.

### 7.4 Push-Apart Test

The push-apart bond test specimen and loading are shown in Figure 7-13a. The specimen description and loading setup are also described and presented in Section 2.6.3.3. Only one quarter of the specimen was modeled making use of symmetry. The concrete, FRP sheet, and bearing steel plate meshes were taken as four-node plain stress solid elements with reduced integration, with dimensions similar to those of the

modified push-apart model detailed in Chapter 5. The 2D indirect push-apart model mesh, deformed shape, loading, and boundary conditions are shown in Figure 7-13b.

The push-apart test is an indirect bond test that only requires the use of a hydraulic jack to apply the load, which makes it suitable for field verifications in retrofit situations. The specimen consists of two concrete blocks that are easy to cast. On the other hand, bonding the FRP sheets to the concrete and impregnating the unbonded portion of the sheet while maintaining the alignment must be done carefully. Special precautions must also be taken to ensure no eccentricity in loading. The concrete is in compression while the FRP is in tension, similar to that in the direct pull test.

The effect of the unbonded FRP sheet length, which is the distance between the two concrete blocks, on the bond behaviour was investigated. Figure 7-14 shows the load-displacement response for models with unbonded FRP sheet lengths of 40, 100, 180, and 200 mm. The figure shows that the behaviour becomes stiffer as the unbonded FRP sheet length decreases. The unbonded length in practice needs to account for a hydraulic jack and two bearing steel plates. The unbonded length was taken as 180 mm in the model used in the comparison between the test methods.

An investigation of the effect of the specimen width separating the two bonding faces on the results was conducted. Numerical investigation of prism dimensions of 120 and 200 mm show that there is no change in behaviour due to the change in the width separating the bonded sheets. The prism dimension between the bonding faces was taken as 200 mm in the comparison between the test methods. The effect of the distance between the bearing steel plate and the bonding faces was also investigated numerically. It was found from the numerical investigation that changing the distance from 20 to 50 mm did not have any significant effect on the results. Therefore, the steel plate distance from the bonding faces was taken as 20 mm in the comparison between the test methods.

Figure 7-15 shows the applied load plotted against the displacement for pushapart models with FRP sheet widths of 25, 50, 100, and 150 mm. The figure shows that increasing the bond width increases the ultimate load, but the ultimate load per unit width is almost constant at 0.175 kN/mm for all bond widths. Figure 7-16 shows the load level,  $P/P_u$ , plotted against the displacement for models with FRP sheet widths of 25, 50, 100, and 150 mm. The figure shows that there is no significant difference between the responses for the different bond widths.

Figure 7-17 and Figure 7-18 show the strain distribution along the FRP sheet length for bond widths of 25 and 150 mm, respectively. The figures show that at the ultimate load there is a limiting strain value that corresponded to the ultimate interface normal and shear stresses. The figures also show that for the same load levels, the strain in the model with 150 mm bond width is higher than that in the model with 25 mm bond width but that there is no significant difference in the transfer length between the models. At ultimate load both the 25 and 150 mm bond width models show almost the same ultimate strain plateau value, but with the 25 mm model covering more of the bond length and having slightly higher strain values beyond the plateau portion.

## 7.5 Modified Push-Apart Test

The modified push-apart bond test specimen is shown in Figure 7-19a. The specimen description and loading setup are also described and presented in Section 3.2. As no anchor sheets are used when the specimens are used to determine the bond strength or the development length, only one quarter of the specimen was modeled making use of symmetry. The 2D modified push-apart numerical model was described in detail in Chapter 5, except that one half of the specimen was modeled rather than one quarter. The mesh, deformed shape, loading, and boundary conditions of the pull-apart model are shown in Figure 7-19b.

Similar to the push-apart test, the modified push-apart test only requires a hydraulic jack to apply the load. Bonding the FRP sheets to the arms of the specimen away from the load application point eliminates the compression in the concrete on

which the sheets are bonded, which is a better simulation of the practical situation. The modified push-apart test is more stable with no free unbonded FRP sheets. It is also easy to bond the FRP sheets to the straight flat concrete surface. On the other hand, the specimen is more difficult to prepare and is relatively large in size.

Figure 7-20 shows the applied load plotted against the displacement for four modified push-apart models with FRP sheet widths of 25, 50, 100, and 150 mm. The figure shows that increasing the bond width increases the ultimate load, but the ultimate load per unit width is almost constant at 0.205 kN/mm for all bond widths. Figure 7-21 shows the load level,  $P/P_u$ , plotted against the displacement for models with FRP sheet widths of 25, 50, 100, and 150 mm. The figure shows that at the same displacement, the load level is higher for smaller bond widths but the responses converges at ultimate load.

Figure 7-22 and Figure 7-23 show the strain distribution along the FRP sheet length at various load levels for models with bond widths of 25 and 150 mm, respectively. The figures show the strain increasing as the load increases, but with no limiting strain plateau, where the increase in the angle between the FRP sheet and the concrete as debonding progresses reduces interface shear stresses and in return the ultimate strain along the sheet. The figures also show that there is no significant difference between the strain values or the transfer lengths at the same load levels for the bond widths of 25 and 150 mm.

The effect of the steel rebar in the concrete arms was investigated using models with different arm widths. Figure 7-24 shows the load-displacement response for four specimen arm widths of 60, 100, 150, and 200 mm with and without steel rebar. The figure shows that the existence of the steel rebar in the concrete arms significantly increases the ultimate load for the narrow arm widths, which prevents premature failure of the concrete arm. The effect of the steel rebar decreases as the arm width increases, up to an arm width of 200 mm, at which no significant difference between models with and without rebar occurs.

The effect of the concrete specimen arm width on the bond behaviour was investigated. The concrete specimen arm widths investigated ranged from 40 to 300 mm. The applied load plotted against the deflection at the point of loading for arm widths ranging from 40 to 100 mm is shown in Figure 7-25. The figure shows that the ultimate load increases as the specimen arm width increases. On the other hand, the behaviour for arm widths ranging from 100 to 300 mm shown in Figure 7-26 shows that the ultimate load decreases as the specimen arm width increases above 100 mm. The reason for the lower ultimate load in the case of narrower arm widths is the excessive deformation of the specimen arms, which causes concrete cracking beyond the FRP sheet. The strain concentrations in the concrete arm beyond the FRP sheet of the model with a 40 mm arm width are shown in Figure 7-27. The tensile stress values in the stress distribution in the concrete arm shown in Figure 7-28 also indicate that the concrete is cracked. On the other hand, the reason for the lower ultimate load in the case of wider arm widths is the concrete failure at the outer concrete face. The strain concentrations at the outer concrete face of the model with an arm width of 300 mm are shown in the strain contour in Figure 7-29. The tensile stress values in the stress distribution shown in Figure 7-30 also indicate that the concrete is cracked.

The previous discussion suggests an optimum arm width for the modified pushapart specimen. According to the numerical analysis, this optimum width was 100 mm for the range of material properties used. The optimum arm width is also expected to depend on the dimensions and stiffness of the bonded FRP sheet.

#### 7.6 Comparison Between the Bond Tests

A comparison is performed between the test methods based on the previous discussions and the data generated in order to recommend a test method to be used as a standard test method and to relate the various test methods to each other.

As discussed separately in the previous subsections, each test method has advantages and limitations. The preparation of the specimen, serviceability, performing

the test, and the stability of the test are all factors in deciding on the preferred bond test method for FRP sheets bonded to concrete. On the other hand, relating the different types of tests will enable the use of their experimental results in developing design guidelines and equations.

#### 7.6.1 Specimen Preparation

The concrete preparation for both the direct pull and push-apart tests are the easiest, where the specimens consist of simple concrete square prisms. But on the other hand, bonding the FRP sheets is difficult. The pull-apart test needs special arrangements to ensure that the rebar are concentric and aligned. If the pull-apart concrete specimen is initially intact then bonding the sheets will be on a flat continuous surface. The modified push-apart concrete specimen is the most difficult to prepare but a flat continuous surface makes bonding the sheet a simple task. All the test methods require special precautions when handling in order to preserve the FRP sheet and its connection with the concrete.

### **7.6.2** Test Serviceability and Execution

The FRP sheet in the direct pull test is limited to 25 mm bond width while the other tests do not have this restriction. Both the direct pull and the pull-apart tests need a testing machine to perform the test, while the push-apart and the direct push-apart tests require a hydraulic jack, which enables the test to be conducted on any flat surface in any location. The modified push-apart specimen size is relatively large, which could cause handling problems and is considered a drawback. The direct pull and push-apart tests require special precautions to ensure that the unbonded FRP sheets are perfectly straight.

#### 7.6.3 Test Stability

The definition of test stability in the context of this discussion is the sensitivity of the test to change in the concrete specimen dimensions, FRP sheet unbonded length, and load eccentricity. The modified push-apart method showed change in the results

due to change in the concrete arm width dimension, which affected the output of the model. The existence of rebar in the modified push-apart specimen arm also affected the results. The unbonded FRP sheet length affected both the direct pull and push-apart results. Eccentricity affects all test methods and special precautions must be taken to ensure that no eccentricity exists.

#### 7.6.4 Relationship Between Test Methods

All the models had the same concrete and FRP material properties as listed in Table 7-1. The FRP sheet bond length and width were also similar in all the models. The only differences between the models were the geometry, loading, and boundary conditions.

The push-apart model had the same load-displacement behaviour as the modified push-apart model when the unbonded FRP sheet length was taken as 40 mm, which is shown in Figure 7-14 and was investigated for the sake of discussion but is not practically possible. For longer unbonded sheet lengths, the push-apart model had a more ductile response and a lower ultimate load than the modified push-apart model. The strain distribution along the FRP sheet in the push-apart model had a limiting plateau regardless of the unbonded length value.

The load-deflection behaviour for all investigated models with bond widths of 25, 50, 100, and 150 mm are shown in Figure 7-31 through Figure 7-34. The figures show that the pull-apart and modified push-apart models have the same trend while the direct pull and push-apart models have a different more ductile trend. The load-deflection relationship between the pull-apart and modified push-apart models is closer at bond widths 25 and 150 mm than at bond widths 100 and 150 mm. On the other hand, for the direct pull and push-apart models is an almost unchanged load-deflection relationship regardless of the bond width.

The ultimate load plotted against the bond width for the results of the models illustrated in Figure 7-31 through Figure 7-34 is shown in Figure 7-35. The figure shows that the relationship in all the models is almost linear with the pull-apart model

having the highest ultimate loads and the push-apart model having the lowest ultimate loads, while there was no significant difference between the modified push-apart and direct pull models. The ultimate load in the pull-apart model was higher than that in the push-apart model by 16.7% at 25 mm and 19.2% at 150 mm.

The ultimate load per unit width is plotted against the bond width in Figure 7-36. The figure shows that the ultimate load per unit width was almost constant in all the models. The average ultimate load per unit width was 0.21 for the pull-apart model, 0.2 for both the modified push-apart and direct pull models, and 0.17 for the push-apart model. This could provide a conversion relationship between the results of the test methods, specially the ultimate load that is used to determine the average bond strength. If one takes the direct pull test as the basis for comparison then the ultimate load for the pull-apart, modified push-apart, and push-apart tests are equal to 0.95, 1.0, and 0.8 of the direct pull test ultimate load, respectively.

The experimental results of modified push-apart and pull-apart tests with bond length of 150 mm listed in Tables 4-1 and 4-2 are also plotted in Figure 7-36. The figure shows that the modified push-apart experimental result is 7.3% higher than the model at bond width 100 mm and 10.9% lower at 150 mm, while both results from the pull-apart test were 16.3 and 20.2% higher. The push-apart experimental result from Alexander and Cheng (1997) shows a 5.8% higher value than the corresponding numerical value. On the other hand the experimental result from the direct pull test done by Bizindavyi and Neale (1999) show a much higher ultimate load per unit width of 70.3%, which could be attributed to the lower bond width of 25 mm used in the tests.

The pull-apart test was the only test with sufficient results including width variations to plot the ultimate load per unit width against the bond width, as shown in Figure 7-37. The figure shows that the ultimate load per unit width is higher when the bond width is less than 50 mm with an average of 0.25, while that for bond widths of 50mm and above is 0.17.

The ultimate load per unit width plotted against the bond length for experimental results from the current study and the literature for the direct pull, pull-apart, push-

apart, and modified push-apart tests are shown in Figure 7-38 through Figure 7-41. The tests chosen had no significant difference in FRP sheet stiffness. The figures show that the experimental results do not have a defined trend relating the ultimate load per unit width to the bond length.

The strain distribution in all test methods at fractions of the ultimate load,  $0.6P_u$ ,  $0.8P_u$ ,  $0.9P_u$ , and  $P_u$ , are shown in Figure 7-42 through Figure 7-45. The figures show that there is no significant difference in the strain distributions at  $0.6P_u$ , but the differences increase as the load increases. The strain distribution at ultimate load in Figure 7-45 shows two trends, one for the methods with unbonded FRP sheets, namely the direct pull and push-apart tests, and another for the methods without unbonded sheets. The first trend has a maximum limiting strain plateau, which corresponds to the ultimate interface normal and shear stresses. The second trend shows a varying strain along the bond length with a maximum value at the crack location. The same behaviour is recorded for bond widths of 150 mm as shown in Figure 7-46.

The load-deflection responses for models with bond widths of 25 mm at 4 kN are shown in Figure 7-47. The figure shows that the responses are relatively close except for the pull-apart test, which is observed at a relatively higher load level, 0.92  $P_u$ . Figure 7-48 shows the load-deflection responses for models with bond widths of 150 mm at 24 kN. The figure shows that the previously discussed trends have already been developed although the load levels were almost the same as for the models with 25 mm bond width.

The previous discussion of the numerical analysis and the practical aspects of the different test methods shows that each test method has advantages and limitations. Two bond tests are proposed as standard test methods based on the loading mechanism available and the advantages shown in the analysis in the current investigation. The pull-apart test is recommended over the direct pull test if a testing machine is available for its consistency in results and ease of performing the test. On the other hand, the modified push-apart test is recommended over the push-apart test for its self-confinement and less sensitivity to eccentricity.

Concrete	Width	w <sub>c</sub> (mm)	200
	Modulus of Elasticity	E <sub>c</sub> (MPa)	31820
	Poisson's ratio	υ <sub>c</sub>	0.15
	Compressive yield strength	f <sub>c</sub> (MPa)	28
	Ultimate comp. Strength	f <sub>c</sub> (MPa)	50
CFRP Sheet	Thickness	t <sub>p</sub> (mm)	0.38
	Bond length	L (mm)	150
	Modulus of Elasticity	E <sub>p</sub> (MPa)	65789
	Poisson's ratio	υ <sub>ρ</sub>	0.26

# Table 7-1 Material properties used in all models

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.



(a) Specimen details and load setup





Figure 7-1 Direct pull bond test



Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.



Figure 7-2 Load-displacement for direct pull tests with various unbonded lengths



Figure 7-3 Development of principal stresses in direct pull concrete specimen



Figure 7-4 Load-displacement for direct pull models



Figure 7-5 Load level-displacement for direct pull models



Figure 7-6 Strain distribution for direct pull models with FRP width of 25 mm



Figure 7-7 Strain distribution for direct pull models with FRP width of 150 mm




(b) Mesh, deformed shape, loading, and boundary conditions

Figure 7-8 Pull-apart bond test



Figure 7-9 Load-displacement for pull-apart models



Figure 7-10 Normalized load-displacement response for pull-apart models



Figure 7-11 Strain distribution for pull-apart models with FRP width of 25 mm



Figure 7-12 Strain distribution for pull-apart models with FRP width of 150 mm





Figure 7-13 Push-apart bond test



Figure 7-14 Load-displacement push-apart tests with different unbonded lengths



Figure 7-15 Load-displacement for push-apart models



Figure 7-16 Normalized load-displacement response for push-apart models



Figure 7-17 Strain distribution for push-apart models with FRP width of 25 mm



Figure 7-18 Strain distribution for push-apart models with FRP width of 150 mm



(a) Specimen details and load setup



(b) Mesh, deformed shape, loading, and boundary conditions

Figure 7-19 Modified push-apart test



Figure 7-20 Load-displacement response for models with L = 150 mm



Figure 7-21 Normalized load-displacement response for models with L = 150 mm



Figure 7-22 Strain distribution for modified push-apart models, w = 25 mm



Figure 7-23 Strain distribution for modified push-apart models, w = 150 mm



Figure 7-24 Load-displacement, modified push-apart tests, various arm widths



Figure 7-25 Load-displacement behaviour for modified push-apart with arm widths ranging from 40 to 100 mm



Figure 7-26 Load-displacement behaviour for modified push-apart with arm widths ranging from 100 to 300 mm



Figure 7-27 Strain distribution in load direction for modified push-part model with 40 mm arm width



Figure 7-28 Stress distribution along the concrete arm for modified push-part model with 40 mm arm width



Figure 7-29 Strain distribution in perpendicular direction to loading for modified push-apart model with 300 mm arm width



Figure 7-30 Stress distribution across outer concrete for modified push-part model with 300 mm arm width







0.1 0.2 0.3 0.4 0.5 0.6 0.7 0.8 0.9 1.0 Displacement (mm)

![](_page_265_Figure_2.jpeg)

0

0.0

![](_page_266_Figure_0.jpeg)

Figure 7-34 Load-displacement for all bond tests, w = 150 mm

![](_page_266_Figure_2.jpeg)

![](_page_266_Figure_3.jpeg)

![](_page_267_Figure_0.jpeg)

Figure 7-36 Ultimate load per width vs. bond width for L = 150 mm

![](_page_267_Figure_2.jpeg)

Figure 7-37 Ultimate load per width vs. bond width, pull-apart tests

![](_page_268_Figure_0.jpeg)

Figure 7-38 Ultimate load per width vs. bond length, direct pull tests

![](_page_268_Figure_2.jpeg)

![](_page_268_Figure_3.jpeg)

![](_page_269_Figure_0.jpeg)

Figure 7-40 Ultimate load per width vs. bond length, push-apart tests

![](_page_269_Figure_2.jpeg)

Figure 7-41 Ultimate load per width vs. bond length, modified push-apart test

![](_page_270_Figure_0.jpeg)

Figure 7-42 Strain distribution for models with FRP width of 25 mm at 0.6  $P_u$ 

![](_page_270_Figure_2.jpeg)

![](_page_270_Figure_3.jpeg)

![](_page_271_Figure_0.jpeg)

Figure 7-44 Strain distribution for models with FRP width of 25 mm at 0.9 Pu

![](_page_271_Figure_2.jpeg)

Figure 7-45 Strain distribution for models with w = 25 mm at ultimate load

![](_page_272_Figure_0.jpeg)

Figure 7-46 Strain distribution for models with w = 150 mm at ultimate load

![](_page_272_Figure_2.jpeg)

Figure 7-47 Strain distribution for models with FRP width of 25 mm at 4 kN

![](_page_273_Figure_0.jpeg)

Figure 7-48 Strain distribution for models with FRP width of 150 mm at 24 kN

### 8. FRP SHEET BOND MODEL

## 8.1. Introduction and Overview

Since FRP materials are non-corrosive, non-magnetic, and generally resistant to chemicals, they are an excellent option for external reinforcement. Strengthening with externally bonded FRP sheets has been shown to be applicable to many types of reinforced concrete (RC) structural members such as columns, beams, slabs, and walls. This method has also been used to strengthen RC chimneys, tunnels, and silos. External FRP reinforcement is employed for flexural and shear strengthening as well as to improve confinement and ductility of compression members. In all cases, however, both the fibres as well as the concrete substrate in the direction of the fibres are all subjected to tensile strain.

Despite the diversity of reinforcing materials used, the different concrete strengths investigated, and the different specimen geometries and loading employed, there is a rather general agreement on the main factors affecting the capacity of the bond between the FRP sheets and concrete. These factors are the FRP sheet bond length, width, and stiffness, and the concrete tensile strength and width.

One of the difficulties with verifying the bond strength of FRP sheets is the various bond test methods used, which generally produce different results. These test methods were presented in Section 2.6 and analyzed in Chapter 7. The analysis of the test methods proposed a simple linear relationship between the results of these methods. Analytical studies such as this and their experimental verification are necessary for the development of sound design guidelines. These linear relationships will not be used in the regression analysis in this study.

#### 8.2. Design Implementations of the Analytical Model

Limit state design demands that the axial force capacity, flexural capacity, and shear capacity of members in a structure are not exceeded. Evaluating the contribution

of externally bonded FRP sheets based on their tensile strength and using guidelines similar to those for conventional steel reinforcement in concrete members is insufficient. The failure modes associated with externally bonded FRP sheets to concrete, detailed in Section 2.5, need to be accounted for in the design. The following subsections will deal with implementation of the debonding type failure mode in both flexure and shear. Proposed equations to predict the average bond strength and the effective bond length may be used to quantify the capacity available of the FRP sheets at delamination.

Once shear forces develop inclined cracks in the concrete, high tensile stresses are developed in the portions of FRP sheet that bridge these cracks. The tensile stresses in vertically oriented FRP sheets are a result of the separation of rigid bodies of concrete on either side of the crack. These tensile stresses must be transferred to the concrete on each side of the crack by interfacial bond stresses. If this interfacial bond is compromised before rupture of the FRP sheet, a debonding failure occurs. Similarly, the use of FRP in direct flexural application is obvious.

In order to address the debonding failure mode in shear design, the bond characteristics of FRP sheets with concrete is considered including concrete strength effect and bonded surface conditions. The use of shear bond tests to characterize the bond mechanism for shear strengthening is reasonable considering the mechanism of force transfer in a shear strengthening configuration. The proposed equations used to predict the average bond strength and effective bond length may also be used to quantify the ultimate capacity of FRP sheets at debonding.

A design approach based on the tensile strength of the FRP sheet assumes perfect bond with the concrete surface. In order to assure the previous conditions, it is necessary to provide a development length,  $L_d$ , calculated from:

$$L_{d} = \frac{\sigma_{f} \cdot n \cdot t_{f}}{\tau_{f}}$$
(8-1)

where  $\sigma_f$  is the maximum stress in the FRP sheet, n is the number of plies,  $t_f$  is the thickness of one FRP sheet, and  $\tau_f$  is the average bond strength. The derivation of equation (8-1) is illustrated in Figure 8-1. This evaluation cannot be applied in the case where there is not enough development length, available physically. A case in point is shear reinforcement, where the depth of the web and the end distances are typically small.

#### 8.3. Design Model

The parametric study in Chapter 6 illustrated the relationship between the ultimate load and the main parameters affecting the bond behaviour. The main parameters directly proportionate to the ultimate load are the bond length, width, and stiffness of the FRP sheet, and the width available and strength of concrete. There are limiting values for all parameters beyond which they no longer contribute to the bond capacity. In the following two equations are proposed. First the effective bond length, then the ultimate bond capacity is defined.

#### 8.3.1. Effective Bond Length Model

The expression predicting the effective bond length proposed by Maeda et al. (1997), which is used in the CSA S806-02 (2002) takes the form:

$$L_{e} = \frac{25350}{\left(t_{p} \cdot E_{p}\right)^{0.58}}$$
(8-2)

in which,  $E_p$  and  $t_p$  are respectively tensile modulus of elasticity in MPa and thickness in mm of the FRP sheet. Equation (8-2) shows that  $E_p t_p$ , which represents the FRP sheet stiffness, is inversely proportional to the effective bond length. This is contradictory to the conclusion derived from the parametric study presented in Section 6.2, which showed that the effective bond length increases with an increase in the FRP sheet stiffness. Chen and Teng (2001) compared five tests with predicted effective bond lengths, and also found that the expression proposed by Maeda et al. (1997) predicted the wrong trend for the effect of  $E_p t_p$  on the effective bond length. Neubauer and Rostásy (1999) and Chen and Teng (2001) proposed expressions based on fracture mechanics similar to that developed by Holzenkämpfer (1994):

$$L_{e} = \sqrt{\frac{E_{p}t_{p}}{4f_{ct}}}$$
(Holzenkämpfer, 1994) (8-3)  
(Holzenkämpfer, 1994)

in which,  $f_{ct}$  is the concrete tensile strength. Equation (8-3) lets the effective bond length be proportional to the FRP sheet stiffness and inversely proportional to the concrete strength. This also does not agree with the conclusion from the parametric study that showed the effective bond length increasing as the concrete strength increases. The expression proposed by Chen and Teng (2001) takes the form:

$$L_{e} = \sqrt{\frac{E_{p}t_{p}}{\sqrt{f_{c}^{\prime}}}}$$
(8-4)

Figure 8-2 shows the effective length plotted against the FRP sheet stiffness,  $E_p t_p$ , for results from the parametric study detailed in Section 6.2. The regression line for the data in the figure shows a relationship between the effective bond length and the stiffness to the power of 0.39. Plotting the effective bond length against the square root of the FRP sheet stiffness for simplicity resulted in an almost linear relationship as compared to the linear regression line of the data points, as shown in Figure 8-3.

Figure 8-4 shows the effective bond length plotted against the FRP sheet bond width for results from the parametric study detailed in Section 6.2. The power regression line for the data in the figure shows a relationship between the effective bond length and the bond width to the power of 0.14. Plotting the effective bond length against the bond width to the power of 0.14 shows an almost linear relationship, as shown in Figure 8-5.

Based on the previous discussion and the trends derived from the parametric study, an expression predicting the effective bond length is proposed:

$$L_{e} = 0.25 w^{0.14} \sqrt{E_{p} t_{p}}$$
 (mm) (8-5)

245

in which  $E_p$ ,  $t_p$ , and w are the FRP sheet modulus of elasticity in MPa, thickness in mm, and width in mm, respectively.

The proposed expression (8-5) predicting the effective bond length is compared with the experimentally evaluated effective bond lengths by Bizindavyi and Neale (1999), Brosens and VanGermert (1997), and the current study. The expressions (8-2) and (8-4) proposed by Maeda et al. (1997) and Chen and Teng (2001), respectively, are also used to evaluate the accuracy of the expression proposed in this study relative to other expressions from the literature. Figure 8-6 shows the effective bond length from test results plotted against the predicted values using Equation (8-2), (8-4), and (8-5). The figure shows that the expression proposed in this study, Equation (8-5), is more accurate and reliable in predicting the effective bond length, where the test to predicted values are gathered around the  $45^{\circ}$  line. The figure also clearly shows that the proposed expression by Maeda et al. (1997) predicts the wrong trend.

Table 8-1 shows the experimental test values used in the comparison. The table also shows the test to predicted ratio of the effective bond length and the values predicted by the proposed expressions. Table 8-2 shows the mean, coefficient of variation (COV), and minimum and maximum values of the test to predicted effective bond length ratios. The table shows that the expression proposed in the current study has the closest mean value to unity and the smallest COV and range between the minimum and maximum values, which is an indication of the better accuracy of the expression proposed in this study.

### 8.3.2. Bond Capacity Model

The bond force is used as a measure of strength rather than the bond stress because the bond stress is usually expressed as an average value at failure, when in fact bond stress varies significantly over the length of the FRP sheet as discussed in Section 4.3.2. Since the bond stress is a structural, rather than material property, the bond force provides a better measure of member response than bond stress.

To improve the accuracy of the analysis, test results from various studies are used representing the different bond test methods. The results used for the modified pushapart method are those from this study, while the results from Udea et al. (1999), Alexander and Cheng (1997), and Drimoussis and Cheng (1994) are used for the pushapart method. The results used for the direct pull method are those by Bizindavy and Neale (1999), Udea et al. (1999), Taljsten (1997), and Chajes et al. (1996), while the results from Adhikary and Mutsuyoshi, (2001), Kamiharako et al. (1999), Udea et al. (1999), Brosens and VanGemert (1997), Horiguchi and Saeki (1997), Maeda et al. (1997), and the current study are used for the pull-apart method.

Figure 8-7 shows the concrete compressive strength,  $f_c$ , plotted against the ultimate bond capacity. The figure shows that in the numerical model,  $f_c$  had no significant effect on the ultimate bond capacity in the debonding type failure. The literature indicates that  $f_c$  is directly proportional to the bond capacity if the failure was shear in the concrete below the FRP sheet.

The concrete width,  $w_c$ , represents the distance between FRP strips in shear strengthening or beam width in flexure strengthening. The numerical analysis in Section 6.3.2 showed that the concrete specimen width,  $w_c$ , is directly proportional to the bond capacity of the bonded FRP sheet. In order to simplify the equation predicting the ultimate bond capacity,  $w_c$  effect was disregarded.

Using the results of the parametric study in Section 6.3.1, Figure 8-8 shows the ultimate load plotted against the FRP sheet bond width. The figure shows that there is an almost bilinear relationship between the ultimate load and the bond width with a change in trend at 100 mm. Linear regression lines are shown in the figure and the relationship was approximated as linear.

Figure 8-9 shows the ultimate load plotted against the FRP sheet bond length for the results from Section 6.2. The figure shows that the ultimate bond capacity increases

as the bond length increases up to an effective bond length. Linear regression lines are shown in the figure and the relationship was also approximated as linear.

To help reduce the effect of bias in the data and to isolate the effects of bond length, width, stiffness, and concrete width, the first approximation of bond capacity in this study uses the following expression:

$$P_{u} = r \cdot w L_{e}$$
(8-6)

in which  $L_e$  is the smaller of the FRP sheet bond length and the effective bond length calculated from Equation (8-5), and r is a factor analogous to the average bond stress. Figure 8-10 shows the ultimate load plotted against the effective bond area, with the effective bond area being equal to the bond width times the effective bond length calculated from Equation (8-5). The linear trend line of the relationship was y = 0.0013x+7.537, which resulted in the expression:

$$P_u = 0.0013 \text{ w } L_e + 7.5$$
 (8-7)

Figure 8-11 shows the ultimate load test to predicted values using Equation (8-7), where the experimental data used were those from this study and from the literature as detailed previously. The figure shows the results scattered around the 45° line.

Figure 8-12 shows the load plotted against the square root of the FRP sheet stiffness. The figure shows that there is an almost linear relationship between the ultimate load and the square root of the FRP sheet stiffness as shown from the proximity of the linear regression lines to the curves.

Using Equation (8-7) as the predicted bond capacity, the next step is to determine the effect of the FRP sheet stiffness. To do this the ratio of the test to predicted ultimate load ratio is plotted against the sheet stiffness per unit effective area,  $\sqrt{E_p t_p}$  /w L<sub>e</sub>, as shown in Figure 8-13. The best fit expression for the ratio of the test to predicted values versus  $\sqrt{E_p t_p}$  /wL<sub>e</sub>, excluding the exceptionally high values, was as follows:

$$\frac{(P_u)_{test}}{0.0013 w L_e} = 1.36 + 636 \frac{\sqrt{E_p t_p}}{w L_e}$$
(8-8)

Rewriting and rounding equation (8-8) to include the higher ratio values gives an expression for the bond capacity of FRP sheets bonded to concrete, as follows:

$$P_{\rm u} = 0.0016 \,{\rm w} \,{\rm L}_{\rm e} + 0.75 \sqrt{{\rm E}_{\rm p} {\rm t}_{\rm p}} \tag{8-9}$$

where the first term represents the effect of the bond area and the second term represents the effect of the FRP sheet stiffness.

Figure 8-14 shows the ultimate load from test results plotted against those predicted by Equation (8-9). The figure shows the 45° line and a linear regression line that shows the proximity of the predicted values to the test results. Table 8-3 shows the ratio of the test ultimate bond load results to those predicted using Equations (8-7) and (8-9). The table includes the ultimate bond capacity test over predicted ratio mean, coefficient of variation (COV), and the minimum and maximum values for each individual research and also for all the tests combined. The predicted ultimate bond capacity values are more accurate if the mean value of the test to predicted ratio is closer to unity, the lower the COV, and the smaller the difference between the maximum and minimum values. Table 8-3 shows the improvement of Equation (8-9) over (8-7) in 12 out of 15 researches that make up 143 out of 155 test results. The table also shows the summary for 126 test results, excluding results by Bizindavyi and Neale (1999) and Chajes et al. (1996), which showed unexpected higher test results. The overall test results using Equation (8-9) showed improvement over that by Equation (8-7) for the 126 test results.

In order to assess the performance of the proposed equation, a comparison with other proposed equations in the literature was performed. The models chosen for comparison are those from Chen and Teng (2001) and Maeda et al. (1997).

The ultimate bond capacity model proposed by Chen and Teng (2001) was based on existing fracture mechanics analysis and experimental observations, which took the form:

$$P_{u} = 0.427 \times 10^{-3} k_{p} k_{L} \sqrt{f_{c}'} w L_{e} \quad (kN)$$
(8-10a)

$$k_{L} = \begin{cases} 1 & \text{if } L \ge L_{e} \\ \sin \frac{\pi L}{2L_{e}} & \text{if } L < L_{e} \end{cases}$$
(8-10b)

$$k_{p} = \sqrt{\frac{2 - w/w_{c}}{1 + w/w_{c}}}$$
 (8-10c)

in which, w is the FRP sheet bond width in mm,  $L_e$  is the effective bond length in mm calculated by (8-4),  $f_c$  and  $w_c$  are the concrete width in mm and compressive strength in MPa, respectively, and k and  $k_p$  are the length factor and width ratio factor, respectively.

The proposed model by Maeda et al. (1997) is an empirical model based on the experimental results obtained from pull-apart test specimens using CFRP sheets.

$$P_u = 110.2 \times 10^{-9} E_p t_p L_e w \ (kN) \tag{8-11}$$

in which,  $t_p$ ,  $L_e$ , w and  $E_p$  are the FRP sheet thickness in mm, effective bond length in mm, bond width in mm, and modulus of elasticity in MPa, respectively.

Figure 8-15 and Figure 8-16 show the bond capacity test values plotted against the predicted values using Equation (8-10) and (8-11), respectively. Comparing the figures together with Figure 8-14 shows that Equations (8-9) and (8-10) are more accurate than (8-11), where the data points in Figure 8-16 are more scattered.

Table 8-4 contains the summary of the ratio of the ultimate load predicted by equations (8-9), (8-10), and (8-11) to that from the test results. The table includes the mean, COV, and the minimum and maximum values for each individual research and also for all the tests combined. Equation (8-11) showed improved prediction over Equation (8-9) and (8-10) in 3 researches and gives equivalent results in 2 others. Due to the limited prediction accuracy by Equation (8-11), the comparison will be between Equation (8-9) and (8-10).

Equation (8-10) gives better predictions than (8-9) in the current study modified push-apart tests, Ueda et al. (1999) pull-apart tests, Kamiharako et al. (1999), Udea et al. (1999), and Taljsten (1997), which amount to 24 out of 155 test results. Equation (8-9) gives better predictions than (8-10) in the seven research series by Ueda et al. (1999) pull-apart tests, Adhikary and Mutsuyoshi (2001), Brosens and VanGermert (1997), Maeda et al. (1997), Drimoussis and Cheng (1994), Bizindavyi and Neale (1999), and Chajes et al. (1996), which amount to 96 out of 155 test results. The two equations have no significant difference in prediction accuracy and reliability in the remaining three research series, which amount to 35 out of 155 test results. Table 8-4 also shows that the model proposed in this study, Equation (8-9), gives an overall better match with the test data and has more reliable results than Equations (8-10) and (8-11) proposed by Chen and Teng (2001) and Maeda et al. (1997), respectively. Excluding the results of Bizindavyi and Neale (1999) and Chajes et al. (1996), shown in brackets in Table 8-4, also shows that Equation (8-9) gives more accurate predictions and is more reliable than Equations (8-10) and (8-11).

In summary, the proposed equations predicting the effective bond length and ultimate bond capacity developed in the current study give more accurate predictions and are more reliable than the proposed equations in the literature. This is clear from the mean value that is closer to unity, the COV values closer to zero, and with smaller range between the minimum and maximum values. For practical applications, the proposed equations must be modified to include knockdown factors to account for longterm loading and cyclic loading as well as durability issues such as freeze and thaw.

	w f <sub>c</sub> ' Et L <sub>e</sub>		Le (Test) Current Study		Maeda et al., 1997	Chen and Teng, 2001	
	mm	MPa	GPa.mm	mm	Test/Pred	Test/Pred	Test/Pred
zindavyi and Jeale, 1999	25.4	42.5	29.2	75.0	1.12	1.15	1.12
	25.4	42.5	58.4	100.0	1.05	2.29	1.06
	25.4	42.5	25.0	55.0	0.88	0.77	0.89
	25.4	42.5	50.0	70.0	0.80	1.47	0.80
al.,	50	57.6	87.19	133.5	1.05	3.86	1.25
a et	50	57.6	70.14	120.3	1.05	3.07	1.25
20 20	50	57.6	43.6	95.7	1.06	1.85	1.26
Ž	50	57.6	21.8	63.5	0.99	0.82	1.18
sens and VanGemert, 1997	120	47.9	38.4	96.9	1.01	1.74	1.30
	120	47.9	76.8	136.2	1.01	3.66	1.29
	120	47.9	115.2	165.9	1.00	5.64	1.29
	80	47.9	38.4	96.1	1.06	1.73	1.29
	80	47.9	76.8	135.4	1.06	3.64	1.28
Bro	80	47.9	115.2	165.2	1.05	5.62	1.28
Current Study	100	47	25.0	75.0	1.00	1.05	1.24
L <u> </u>	<u>, L. e.a.onn, on</u> gogenostational	uuruuntutugettelekkohtaideesta yoo etoosoo	1999 - January Martin, 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 199 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999	Mean COV Min. Max	1.01 0.08 0.80 1.12	2.56 0.63 0.77 5.64	1.19 0.13 0.80 1.30

# Table 8-1 Effective bond length comparison details

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

		Current Study	Maeda et al., 1997	Chen and Teng, 2001	
_ ف <u>ج</u> ز	Mean	0.96	1.42	0.97	
dav Veal 9 (4	cov	0.15	0.46	0.15	
lizin nd N 199	Min.	0.80	0.77	0.80	
<u></u> м м	Max.	1.12	2.29	1.12	
al.,	Mean	1.04	2.40	1.24	
a et 1 (4	COV	0.03	0.56	0.03	
kab 200	Min.	0.99	0.82	1.18	
Na	Max.	1.06	3.86	1.26	
ort,	Mean	1.03	3.67	1.29	
ns a eme 7 (6	COV	0.03	0.47	0.01	
ose anG 199	Min.	1.00	1.73	1.28	
۳ چ ۳	Max.	1.06	5.64	0.46 $0.15$ $0.77$ $0.80$ $0.29$ $1.12$ $0.40$ $1.24$ $0.56$ $0.03$ $0.82$ $1.18$ $0.86$ $1.26$ $0.67$ $1.29$ $0.47$ $0.01$ $0.73$ $1.28$ $0.64$ $1.30$ $0.05$ $1.24$ $0.61$ $0.14$ $0.77$ $0.80$ $0.64$ $1.30$	
Current Study (1)	Mean	1.00	1.05	1.24	
	Mean	1.01	2.67	1.18	
ests	cov	0.08	0.61	0.14	
All t	Min.	0.80	0.77	0.80	
	Max.	1.12	5.64	1.30	

Table 8-2 Test vs. predicted effective bond length summary

	Eq.	8-7	8-9	a constant a	Eq.	8-7	8-9
Current Study	Mean	1.16	1.19	Alexander and Cheng,	Mean	0.84	0.93
Modified push-apart (5)	COV	0.08	0.11	1997	COV	0.17	0.17
	Min.	1.02	0.97	Push-apart test (10)	Min.	0.67	0.73
	Max.	1.23	1.34		Max.	1.09	1.18
Ueda et al., 1999	Mean	0.88	1.02	Drimoussis and Cheng,	Mean	0.88	0.96
Pull-apart test (11)	COV	0.23	0.18	1994	COV	0.13	0.13
	Min.	0.52	0.67	Push-apart test (3)	Min.	0.76	0.82
	Max.	1.18	1.28		Max.	0.98	1.06
Adhikary and Mutsuyoshi,	Mean	1.14	1.11	Udea et al., 1999	Mean	1.21	1.23
2001	COV	0.13	0.09	Push-apart tests (2)	COV	0.04	0.13
Pull-apart test (7)	Min.	0.96	1.01		Min.	1.18	1.12
	Max.	1.38	1.31		Max.	1.25	1.35
Brosens and VanGemert,	Mean	0.88	0.84	Bizindavyi and Neale,	Mean	1.26	1.67
1997	cov	0.12	0.10	1999	COV	0.34	0.23
Pull-apart test (28)	Min.	0.61	0.70	Direct pull tests (24)	Min.	0.68	1.03
	Max.	1.11	1.05		Max.	2.05	2.39
Kamiharako et al., 1999	Mean	0.65	0.85	Chajes et al., 1996	Mean	1.16	1.76
Pull-apart test (10)	COV	0.35	0.23	Direct pull tests (5)	cov	0.15	0.13
	Min.	0.35	0.57		Min.	0.87	1.37
	Max.	1.02	1.15		Max.	1.33	2.01
Maeda, et al., 1997	Mean	0.92	0.97	Taljsten, 1997	Mean	1.42	1.07
Pull-apart test (18)	cov	0.25	0.18	Direct pull tests (4)	COV	0.16	0.16
	Min.	0.49	0.62		Min.	1.24	0.92
	Max.	1.31	1.28		Max.	1.74	1.32
Nakaba et al., 2001	Mean	1.29	1.36	Udea et al., 1999	Mean	1.38	1.29
Pull-apart test (12)	cov	0.20	0.11	Direct pull tests (3)	COV	0.13	0.05
	Min.	0.99	1.04		Min.	1.19	1.22
ι,	Max.	1.72	1.57		Max.	1.55	1.36
Current Study	Mean	0.73	0.82	Summary of all tests	Mean	1.00	1.12
Pull-apart test (13)	COV	0.17	0.12	155 Specimens		(0.95)	(0.99)
	Min.	0.52	0.69	(126 Specimens)	cov	0.31 (0.28)	0.33 (0.22)
	Max.	0.97	1.05		Min.	0.35 (0.35)	0.57 (0.57)
					Max.	2.05 (1.74)	2.39 (1.57)

# Table 8-3 Evolution of proposed model

	Eq.	8-9	8-10	8-11		Eq.	8-9	8-10	8-11
Current Study	Mean	1.19	1.14	0.92	Alexander and Cheng,	Mean	0.93	0.93	0.77
Modified push-apart (5)	COV	0.11	0.07	0.15	1997	COV	0.17	0.17	0.18
	Min.	0.97	1.04	0.70	Push-apart test (10)	Min.	0.73	0.74	0.59
	Max.	1.34	1.23	1.09		Max.	1.18	1.21	0.96
Ueda et al., 1999	Mean	1.02	1.28	1.11	Drimoussis and Cheng,	Mean	0.96	1.08	0.78
Pull-apart test (11)	COV	0.18	0.20	0.26	1994	cov	0.13	0.13	0.13
	Min.	0.67	0.73	0.64	Push-apart test (3)	Min.	0.82	0.92	0.67
	Max.	1.28	1.53	1.51		Max.	1.06	1.19	0.86
Adhikary and Mutsuyoshi,	Mean	1.11	1.22	0.90	Udea et al., 1999	Mean	1.23	1.18	0.99
2001	COV	0.09	0.08	0.09	Push-apart tests (2)	COV	0.13	0.01	0.16
Pull-apart test (7)	Min.	1.01	1.12	0.82		Min.	1.12	1.18	0.87
	Max.	1.31	1.37	1.07		Max.	1.35	1.19	1.10
Brosens and VanGemert,	Mean	0.84	0.83	0.73	Bizindavyi and Neale,	Mean	1.67	1.92	2.17
1997	COV	0.10	0.12	0.11	1999	COV	0.23	0.23	0.25
Pull-apart test (28)	Min.	0.70	0.65	0.59	Direct pull tests (24)	Min.	1.03	1.18	1.30
	Max.	1.05	1.10	0.92		Max.	2.39	2.75	3.20
Kamiharako et al., 1999	Mean	0.85	1.16	1.11	Chajes et al., 1996	Mean	1.76	2.08	2.33
Pull-apart test (10)	COV	0.23	0.10	0.18	Direct pull tests (5)	COV	0.13	0.08	0.05
	Min.	0.57	0.99	0.77		Min.	1.37	1.85	2.25
	Max.	1.15	1.33	1.48		Max.	2.01	2.31	2.54
Maeda, et al., 1997	Mean	0.97	1.06	0.94	Taljsten, 1997	Mean	1.07	1.00	1.13
Pull-apart test (18)	COV	0.18	0.18	0.19	Direct pull tests (4)	COV	0.16	0.12	0.27
	Min.	0.62	0.68	0.59		Min.	0.92	0.88	0.73
	Max.	1.28	1.38	1.24		Max.	1.32	1.15	1.48
Nakaba et al., 2001	Mean	1.36	1.36	1.36	Udea et al., 1999	Mean	1.29	0.97	1.12
Pull-apart test (12)	COV	0.11	0.11	0.12	Direct pull tests (3)	COV	0.05	0.06	0.08
	Min.	1.04	1.04	1.07		Min.	1.22	0.91	1.05
	Max.	1.57	1.57	1.64		Max.	1.36	1.01	1.21
Current Study	Mean	0.82	0.84	0.70	Summary of all tests	Mean	1.12	1.21	1.17
Pull-apart test (13)	COV	0.12	0.16	0.18	155 Specimens (126 Specimens)	0.014	(0.99)	(1.02)	(1.04)
	Min.	0.69	0.70	0.56		COV	0.33 (0.219)	0.370 (0.220)	0.504 (0.223)
	Max.	1.05	1.08	1.01		Min.	0.57	0.65	0.56
					Reserves and a second se		(0.57)	(0.67)	(0.65)
						Max.	2.39 (1.57)	2.75 (1.63)	3.20 (1.57)

# Table 8-4 Summary of comparison with all test results


Figure 8-1 Schematic sketch and derivation of development length



Figure 8-2 Effective length-stiffness for w = 100 mm (Section 6.2)











Figure 8-5 Effective length vs. square root of bond width relationship



Figure 8-6 Test vs. predicted effective length



Figure 8-7 Concrete compressive strength vs. ultimate load (Section 6-5)



Figure 8-8 Ultimate load vs. FRP sheet bond width (Section 6.3.1)



Figure 8-10 Ultimate load vs. bond area



Figure 8-11 Ultimate load test to predicted values, Equation 8-7



Figure 8-12 Ultimate load vs. square root of stiffness (Section 6-4)



Figure 8-13 Ultimate load test to predicted ratio vs  $\sqrt{E_p t_p}$  /wL<sub>e</sub>, all tests



Figure 8-14 Ultimate load test to predicted values, Equation 8-9



Figure 8-15 Ultimate load test to predicted values, Equation 8-10



Figure 8-16 Ultimate load test to predicted values, Equation 8-11

# 9 SUMMARY, CONCLUSIONS, AND RECOMMENDATIONS

#### 9.1 Summary

The FRP strengthening schemes for flexure and shear and the possible failure mechanisms show the existence of a premature type failure caused by debonding of the FRP sheets. Incomplete understanding of the parameters affecting the bond behaviour was seen from the literature review. The literature review also showed the need for reliable experimental test results to investigate the effect of the different parameters on the bond behaviour. In an attempt to provide insight into the behaviour of the FRP to concrete bond, an experimental and numerical investigation was undertaken.

Two experimental test series were conducted in this research using two different types of bond test methods. The first series used a proposed modified push-apart test and the other used a pull-apart test. The experimental tests were also used to verify a number of numerical models, which were then used to conduct further numerical analysis. A parametric study was conducted using numerical models in order to investigate the effect of the different parameters on the bond behaviour.

Reviewing the different conventional bond test methods used in different bond studies pointed out the need to develop a rational standard test method for FRP sheets bonded to concrete. A numerical investigation was conducted in order to investigate the bond test methods and recommend a method to be used as a standard test method. A conversion standard between the test methods was also proposed.

Several researchers have proposed different models to predict the effective length and the ultimate bond capacity. A model was proposed based on evaluation of the proposed models in the literature, the extensive parametric study mentioned above, and test results from the current study and the literature.

# 9.2 Conclusions

Observations were recorded and conclusions drawn from the experimental study that were further investigated using numerical analysis. The conclusions for both the experimental and numerical analysis are listed separately for more clarity in the following subsections. These conclusions are of course subject to the scope stated in Section 1.2.

## 9.2.1 Conclusions from the Experimental Analysis

In all tests conducted debonding propagates from the crack location towards the free end of the CFRP sheet and the resisting bonded area of the sheet increases as the applied load increases. For a constant bond width, bond lengths above 100 mm caused no increase in the ultimate capacity of joints tested in this study. This agrees with previous findings by other researchers of the existence of an effective bond length, above which no contribution to the capacity of the joint will be obtained.

The tests also showed that the strain distribution across the width of the CFRP sheet had different behaviour depending on whether the specimen was precracked or not. In general, however, the strain values at the edge of the CFRP sheet are higher than those in the centre throughout the sheet length. The difference in strain increases as the load increases and as the distance from the crack decreases. At the ultimate load at the crack location, the difference between the strain values at the edge and those at the centre of the sheet was 35% on average. The difference also increases as the sheet width increases. This is likely due to a shear lag effect in the concrete substrate. There was also no significant difference in the strain distributions between the pull-apart test method and the modified push-apart test method at the same load value. But the ultimate loads were different in the two methods, which could be attributed to the different loading arrangements and boundary conditions.

The tests also showed that the width of a perpendicular unidirectional CFRP anchor sheet that is bonded outside the tested sheet is ineffective. It was also observed

that placing the anchor sheet above or below the test sheet makes no difference to the failure mechanism. However, a perpendicular unidirectional CFRP anchor sheet located below the tested CFRP sheet showed better or equivalent overall bond behaviour compared to anchor sheets located above the tested sheet. The anchor distance from the crack did not appear to affect the load capacity or the ductility of the joint, except when the anchor sheet was at the crack face in which case there was an increase in the load capacity and ductility of the joint.

The individual values of the CFRP sheet bond length, L, and bond width, w, are the governing factors in the bond behaviour rather than the L/w ratio. Increasing the bonded width of the FRP sheet decreases the average bond strength and the ductility of the joint. The effective bond length governs the ultimate load while the effective bond width governs the average bond strength.

### 9.2.2 Conclusions from the Numerical Analysis

A comparison between the output of the numerical models and the experimental results proved that it is possible to predict the load-displacement behaviour, the debonding type failure, and the strain distribution in FRP sheets bonded to concrete members using simple numerical models. This allowed a number of generalized conclusions

The results of the numerical models confirmed the existence of an effective bond length beyond which no increase in the capacity is gained for a wide range of parameters. The FRP sheet effective bond length increased as the bond width increased. Increasing the sheet stiffness also increased the effective bond length up to a certain stiffness value of 100 GPa·mm beyond which no increase in the effective bond length was gained. The concrete strength did not appear to have a significant effect on the effective bond length.

The numerical analysis showed also that the average bond strength decreases as the bond width exceeds 100 mm while no significant change occurs at bond width

values below 100 mm. This explains the contradictory conclusions in the literature regarding the effect of the bond width on the bond behaviour. The same conclusion implies that results from narrow sheets over-estimate the average bond strength.

The strains at the edge of the FRP sheets are higher than those at its centre and the difference increases as the FRP sheet width increases and as the concrete width increases. This indicates that the investigations carried out with strain measurements only at the centre of the FRP sheet underestimate the strain values. The strain values are also shown to increase as the width decreases.

The actual values of the concrete specimen width and FRP sheet width affect the bond behaviour rather than the ratio of the FRP sheet to concrete width. This further indicates that the concrete width effect on the bond behaviour needs to be considered or a unified specific concrete specimen width defined and implemented in the standards.

The average bond strength increases as the FRP sheet stiffness increases but up to a certain limit. This limit is defined as the effective sheet stiffness and marks the point beyond which no increase in the average bond strength is gained. The effective sheet stiffness changed with the bond lengths. This conclusion has a practical implication of an optimum number of FRP sheet plies for every bond length.

The expression used to predict the effective bond length in the CSA S806-02 (2002) was shown to be inaccurate at higher FRP sheet stiffness values. New expressions based on regression are proposed to predict the effective bond length and the ultimate bond capacity of the FRP sheets bonded to concrete:

$$L_{e} = 0.25 w^{0.14} \sqrt{E_{p} t_{p}} \quad (mm)$$
(9-1)

$$P_{u} = 0.0016 \text{ w } L_{e} + 0.75 \sqrt{E_{p} t_{p}} \quad (\text{kN})$$
(9-2)

These expressions are shown to be more accurate and reliable than those proposed in the literature to date.

It is necessary that the debonding effect be introduced in the 3D numerical models to numerically study the anchor sheet effect on the behaviour of FRP sheets bonded to concrete.

#### 9.3 **Recommendations for code changes**

Based on the experimental, numerical, and regression analysis, some recommendations are proposed. For example, it is necessary to introduce refinements to the procedure in Annex P in CSA S806-02 (2002) for determining the value of the FRP sheet effective bond length. These are explained in detail below

Due to the relatively high variability in bond related test results, taking the average of two tests for each bond length together with the expected trend generated by the data points will safeguard for any inconsistent results. The length factor is proposed to range from 0.6 to 1.6 with increments of 0.2 in order to optimize the number of specimens tested. In this proposed change an overall number of twelve specimens would be tested with two specimens for each variable.

The proposed expression (9-1) that predicts the effective bond length for FRP sheets bonded to concrete developed in this study is recommended to replace the existing expression in the CSA S806-02 (2002) and the ISIS Design Manual (2001), where the expression used in these guides was shown to predict the wrong trend.

Test procedures should employ a constant concrete specimen length rather than changing the specimen length with the change in the FRP sheet length. This is to use a constant mould sizes. The proposed concrete specimen length is 500 mm for up to three plies of FRP sheets, where the effective length increases as the number of plies. The specimen length shall be increased by 100 mm for each extra ply.

## 9.4 Recommendations for Future Research

Based on the experimental and numerical analysis, some recommendations are proposed for future research. For example, studying different anchorage systems and their effect on the bond behaviour in order to propose a rational standard anchor system is recommended.

Experimental test results used to compare and relate bond test methods are from different research studies, which vary in material properties and test conditions. Investigating the effect of the test method experimentally is therefore recommended. This would help develop a conversion standard for the test methods in order to relate the test results from the different bond tests.

The numerical analysis showed that the actual values of the FRP sheet width and the concrete specimen width affect the bond behaviour rather than the FRP sheet width to the concrete specimen width ratio value. Investigating experimentally the effect of the FRP sheet width relative to the concrete specimen width is recommended. The concrete width is analogous to beam width in beam flexure strengthening or the distance between FRP strips in shear strengthening.

The bond test methods investigated in this study involved applying the load in the direction of the fibre. Studying a different state of stress, where the load is applied at an angle to the FRP sheets or the crack is inclined relative to the applied load and it's effect on the bond behaviour is also recommended.

### REFERENCES

- ABAQUS Standard (1998), Version 5.8. Habbitt, Karlsson, and Sorenson Inc., Pawtucket, RI.
- Adhikary, B. B. and Mutsuyoshi, H. (2001) "Study on the Bond between Concrete and Externally bonded CFRP Sheet," FRPPCS-5, Thomas Telford, London, 371-378.
- Alexander J. S. and Cheng, J. J. R. (1997) "Shear Rehabilitation of G-Girder Bridges Using CFRP Sheets." Structural Engineering Report No. 218, Department of Civil and Environmental Engineering, University of Alberta, October, 181 p.
- Ali, M, Oehlers, D.J. (1997) "Enhancing the Shear Capacity of Reinforced Concrete Beams by Bonding Steel Plates to their Sides" 15th ACMSM '97, Melbourne.
- Arduini M. and Nanni A. (1997) "Behavior of Pre-cracked RC Beams Strengthened with Carbon FRP Sheets." ASCE, Journal of Composites in Construction, Vol. 1, No. 2, May, pp. 63-70.
- Arduini M. and Nanni A. (1997) "Parametric Study of Beams with Externally Bonded FRP Reinforcement." ACI Structural Journal, 94 (5): 493-501.
- Arduini M., Di Tommaso A., Manforoni O., and Nanni A. (1996) "Failure Mechanisms of Concrete Beams Reinforced with FRP Flexible Sheets." Advanced composite materials in bridges and structures, M. El-Badry, Editor, Canadian Society for Civil Engineering, Montreal, Quebec.
- Arduini, M., Di Tommaso, A., and Manfroni, O. (1995) "Fracture mechanisms of concrete beams bonded with composite plates. Non-metallic (FRP) reinforcement for concrete structures. *Edited by* L. Taerwe. RILEM, London, U.K., pp. 483–491.
- ASTM D-3039/D 3039M-95a (1995) "Standard Test Method for Tensile Properties of Polymer Matrix Composite Materials." American Society for Testing and Materials, West Conshohocken, Pennsylvania, Vol. 15.03.

Bathe, K. J. (1996) "Finite Element Procedures" Prentice Hall.

- Bizindavyi, L. and Neale, K. W. (1999) "Transfer Lengths and Bond Strengths for Composites Bonded to Concrete," Journal of Composites for Construction, 153-160.
- Bonacci, J. F. (1996) "Strength, Failure Mode and Deformability of Concrete Beams Strengthened Externally with Advanced Composites." Advanced composite materials in bridges and structures, M. El-Badry, Editor, Canadian Society for Civil Engineering, Montreal, Quebec.

- Brozens, K. and Van Gemert, D. (1997) "Anchoring Stresses Between Concrete and Carbon Fibre Reinforced Laminates," Non-Metallic (FRP) Reinforcement for Concrete Structures, proceedings 3<sup>rd</sup> International Symposium.
- Chaallal, O., Nollet, M. J., and Perraton, D. (1998) "Strengthening of Reinforced Concrete Beams with Externally Bonded Fiber-Reinforced-Plastic Plates: Design Guidelines for Shear and Flexure." Canadian Journal of Civil Engineering, 25, 692-704.
- Chajes M.J., Finch W.W., Januszka T.F., Thomson T.A., and Mertz D.R. (1996) "Bond and Force Transfer of Composite Material Plates Bonded to Concrete." ACI Structural Journal, 92, 295-303.
- Chajes M.J., Januszka T.F., Mertz D.R., Thomson T.A. and Finch W.W. (1995) "Shear Strengthening of Reinforced Concrete Beams using Externally Applied Composite Fabrics." ACI Structural Journal, 92, 295-303.
- Chen, J. F. and Teng J. G. (2001) "Anchorage Strength Models for FRP and Steel Plates Bonded to Concrete." Journal of Structural Engineering, Vol. 127, No. 7, 784-791.
- Chen, W. F. (1982) "Plasticity in Reinforced Concrete" McGraw Hill.
- Cheng R.J.J. (1998) "Strengthening of Concrete Beams with FRP Materials." ISIS Alberta workshop, Edmonton, Alberta.
- Collins, M. P. and Mitchell, D. (1987) "Prestressed Concrete Basics". Canadian Prestressed Concrete Institute, CPCI, 1<sup>st</sup> edition.
- CSA A23.3-94 (1994) "Design of Concrete Structures." Canadian Standards Association, Rexdale, Ontario.
- CSA S6-88 (1988) "Design of Highway Bridges." Canadian Standards Association, Rexdale, Ontario.
- CSA S806-02 (2002), "Design and Construction of Building Components with FRP." Canadian Standards Association, Rexdale, Ontario.
- Deniaud C. and Cheng J. J. R. (2000) "Behaviour of Reinforced Concrete Beams Strengthened in Shear with FRP Sheets." University of Alberta, Structural Engineering Report No. 234.
- De Lorenzis L., Miller B., and Nanni A. (2000) "Bond of FRP Laminates to Concrete." ACI Structures Journal, July.
- Dolan B. E., Hamilton H. R., Dolan C. W. (1998) "Strengthening with Bonded FRP Laminate." Concrete International, Vol. 20, No. 6, 51-55.

Donnet, J-P, and R. C. Bansal, (1990) "Carbon Fibers," Marcel Dekker, Inc., New York.

- Drimoussis E.H. and Cheng J.J.R. (1994) "Shear Strengthening of Concrete Girders using Carbon Fiber Reinforced Plastic Sheets." Structural Engineering Report No 205, University of Alberta, Department of Civil Engineering, Edmonton, Alberta, 177 p.
- Erki, M.A., and Heffernan, P.J. (1995) "Reinforced concrete slabs externally strengthened with fiber-reinforced plastic materials," Non-metallic (FRP) reinforcement for concrete structures. *Edited by* L. Taerwe. RILEM, London, U.K., pp. 509–516.
- Fashole-Luke P. S. (1999) "Strengthening of Reinforced Concrete Structures; using externally-bonded FRP composites in structural and civil engineering" Edited by L. C. Holloway and M. B. Leeming, Woodhead publishing limited, 222-241.
- Hiroyuki, Y., and Wu, Z. (1997). "Analysis of debonding fracture properties of CFS strengthened member subjected to tension." Non-Metallic (FRP) Reinforcement for Concrete Structures, 3<sup>rd</sup> International Symposium, Vol. 1., 287-294.
- Hollaway, L. C. and Leeming, M. B. (1999) "Strengthening of Reinforced Concrete Structures; using Externally Bonded FRP Composites in Structural Engineering." Woodhead publishing Ltd, Cambridge, England.
- Holzenkämpfer, O. (1994) "Ingenieurmodelle des verbundes geklebter bewehrung für betonbauteile." Dissertation, TU Braunschweig (in German).
- Horiguchi, T. and Saeki, N. (1997) "Effect of Test Methods and Quality of Concrete on Bond Strength of CFRP Sheet," Non-Metallic (FRP) Reinforcement for Concrete Structures, 3<sup>rd</sup> International Symposium, Vol. 1., 265-270.
- ISIS Canada Design Manual (2001) "strengthening Reinforced Concrete Structures with Externally Bonded Fibre Reinforced Polymers" The Canadian Network of Centres of Excellence on intelligent Sensing for innovative Structures, Part 4.
- Kamiharako, A., Shimomura, T., Maruyama, K., and Nishida, H. (1999) "Analysis of Bond and Debonding Behavior of Continuous Fiber Sheet Bonded on Concrete," Journal of Materials, concrete Structures and pavements, Japan Society of Civil Engineers, V.634, 197-208. (in Japanese)
- Khalifa, W. J. Gold, A. Nanni, and M. I. Abdel Aziz (1998) "Contribution of Externally Bonded FRP to Shear Capacity of RC Flexural Members." Journal of composites in construction, Vol. 2. No 4., 195-202.
- Kobayashi K., Fujii, S. Yabe, Y. Tsukagoshi, H. and Sugiyama, T. (2001) "Advanced wrapping system with CF-anchor – stress transfer mechanism of CF-anchor" FRPRCS-5 Conference Cambridge, 379-388.

Maeda T., Asano Y., Sato Y., Ueda T., and Kakuta Y. (1997) "A Study on Bond Mechanism of Carbon Fiber Sheet." Non-Metallic (FRP) Reinforcement for Concrete Structures, 3<sup>rd</sup> International Symposium, Vol. 1., 279-286.

MBrace (1998) "Composite Strengthening System Design Guide." Master Builders Inc.

Microsoft® Excel 2000 (1999), Microsoft corporation.

- Mitsubishi Chemical (1994), "REPLARK: Carbon Fiber Prepreg for Retrofitting and Repair Method," Manufacturer Publication, Tokyo, Japan, 18 pp.
- Nakaba K., Kanakubo T., Furuta T., and Yoshizawa H. (2001) "Bond Behavior Between Fiber-Reinforced Polymer Laminates and Concrete." *American Concrete Institute Structural Journal*, 98, No. 3, 359-367.

Nanni A. (1997) "CFRP Strengthening." Concrete International, Vol. 19, No 6.

- Neubauer, u., and Rostásy, F. S. (1999) "Bond Failure of Concrete Fiber Reinforced Polymer Plates at Inclined Cracks-Experiments and fracture Mechanics Model," 4<sup>th</sup> International Symposium, FRP Reinforcement for RC structures, Editors: Dolan C.W., Rizkalla S.H., and Nanni A, Baltimore.
- Ritchie P.A., Thomas D.A., Lu L.W. and Connelly G.M. (1991) "External Reinforcement of Concrete Beams using Fiber Reinforced Plastics." *ACI Structural Journal*, 88, 490-500.
- Roberts, T.M. and Haji-Kazemi, H. (1989), "Theoretical Study of the Behavior of Reinforced Concrete Beams Strengthened by Externally Bonded Steel Plates," Proc. Inst. Civ. Engrs, Part 2, 39-55.
- Saadatmanesh H., and Ehsani M.R. (1991a) "RC Strengthened with GFRP Plates: I: Experimental Study." Journal of Structural Engineering Division, ASCE, 117, 3417--3433.
- Saadatmanesh H., and Ehsani M.R. (1991b) "RC Strengthened with GFRP Plates: II: Analysis and Parametric Study." *Journal of Structural Engineering Division*, ASCE, 117, 3434-3455.
- Sato Y., Kakuta Y., and Ono S. (1997a) "Shear Strengthening of Existing Reinforced Concrete Beams by CFRP Sheets." Proceedings of the Non-Metallic FRP Reinforcement for Concrete Structures (3<sup>rd</sup> International Symposium), Sapporo, Japan, pp. 507-514.
- Sato Y., Ueda T., Kakuta Y. and Tanaka T. (1996) "Shear Reinforcing Effect of Carbon Fiber Sheet Attached to Side of reinforced Concrete Beams." 2nd International Conference on Advanced Composite Materials in Bridges and Structures, M. El-Badry, ed., CSCE, Montréal, Québec, 621-628.

- Sharif, A. Al-Sulaimani, G.J., Basunbul, I.A., Baluch, M.H., and Ghaleb, B.N. (1994). " Strengthening of Initially loaded Reinforced Concrete Beams using FRP Plates. ACI Structural Journal, V. 91, No. 2, pp. 160-168.
- Spadea G,Bencardino F, and Swamy R. N. (1998) "Structural Behavior of Composite RC Beams with Externally Bonded CFRP." Journal of Composites in Construction, ASCE, 132-137.
- Swamy R.N., Jones R. and Charif, A. (1986) "Shear Adhesion Properties of Epoxy Resin Adhesives." Proceedings of an International Symposium on Adhesion between Polymers and Concrete, pp. 741-755.
- Tadros G., Abdelrahman A., and Rizkalla S. (1998) "Design of Concrete Members with FRP Materials." ISIS Alberta workshop, Edmonton, Alberta.
- Täljesten, B., (1994) "Plate Bonding. Strengthening of existing concrete structures with epoxy bonded plates of steel or fibre reinforced plastics," Doctoral Thesis, Lulea university of technology, Sweden.
- Täljesten, B., (1997) "Defining Anchor Lengths of Steel and CFRP Plates Bonded to Concrete," International Journal of Adhesives, Vol. 17, No. 4.
- Tanaka, T. (1996) "Shear Resisting Mechanism of Reinforced Concrete Beams with CFS as Shear Reinforcement." Graduation thesis, Hokkaido University, Japan (in Japanese).
- Triantafillou, T.C. (1998) "Shear Strengthening of Reinforced Concrete Beams Using Epoxy-Bonded FRP Composites." ACI Structural Journal, 95 (2), 107-115.
- Tumialan, G., Serra, P., Nanni, A., and Belarbi, A. (1999) "Concrete Cover Delamination in RC Beams Strengthened with FRP Sheets," SP-188, ACI Proceedings, 4<sup>th</sup> International Symposium on FRP for Reinforcement of Concrete Structures, Editors: Dolan C.W., Rizkalla S.H., and Nanni A, Baltimore, 725-735.
- Ueda T., Sato Y., and Asono Y. (1999) "Experimental Study on Bond Strength of Continuous Carbon Fiber Sheet." 4<sup>th</sup> International Symposium, FRP Reinforcement for RC Structures. Editors: Dolan C.W., Rizkalla S.H., and Nanni A, Baltimore.
- Van Gemert, D. (1980) "Force Transfer in Epoxy Bonded Steel-Concrete Joints", International Journal of Adhesion and Adhesive, 2, pp. 67-72.
- Varastehpour, H., and Hamelin, P. (1996) "Analysis and Study of Failure Mechanism of RC Beams Strengthened with FRP Plate." 2nd International Conference on Advanced Composite Materials in Bridges and Structures, M. El-Badry ed., CSCE, Montréal, Québec, 527-537.

 Yoshizawa H., Myojo T., Okoshi M. Mizukoshi M., and Kliger H.S. (1996) "Effect of Sheet Bonding Condition on Concrete Members Having Externally Bonded Carbon Fiber Sheet." 4<sup>th</sup> Materials Engineering Conference, ASCE annual convention, Washington D.C.